### PAPER • OPEN ACCESS

## Horizon-scale tests of gravity theories and fundamental physics from the Event Horizon Telescope image of Sagittarius A\*

To cite this article: Sunny Vagnozzi et al 2023 Class. Quantum Grav. 40 165007

View the article online for updates and enhancements.

## You may also like

- Angular distributions of multiply charged fragments from dissociation of Nitrogen molecules by Xe<sup>9+</sup> impact Ajit Kumar, T Sairam, Jyoti Rajput et al.
- Use of ICT for sustainable transportation P Agarwal and M A Alam
- Breaking Boundaries in LED Technology: Exploring the Revolutionary Diode Characteristics of Screen Printed (TiO2) 1x (CuO) x Thick Films Rayees Zargar, Santosh Chackrabarti, Tuiba Mearaj et al.

Class. Quantum Grav. 40 (2023) 165007 (122pp)

# Horizon-scale tests of gravity theories and fundamental physics from the Event Horizon Telescope image of Sagittarius A\*

Sunny Vagnozzi<sup>1,2,3,\*</sup>, Rittick Roy<sup>4,5</sup>, Yu-Dai Tsai<sup>6</sup>, Luca Visinelli<sup>7,8</sup>, Misba Afrin<sup>9</sup>, Alireza Allahyari<sup>10,11</sup>, Parth Bambhaniya<sup>12</sup>, Dipanjan Dey<sup>12,13</sup> Sushant G Ghosh<sup>9,14</sup>, Pankaj S Joshi<sup>12,15</sup>, Kimet Jusufi<sup>16</sup>, Mohsen Khodadi<sup>11,17,18,19</sup>, Rahul Kumar Walia<sup>14,20</sup>, Ali Övgün<sup>21</sup> and Cosimo Bambi<sup>4,5</sup>

<sup>1</sup> Department of Physics, University of Trento, 38123 Povo (TN), Italy

<sup>2</sup> Trento Institute for Fundamental Physics and Applications-INFN, 38123 Povo (TN), Italy

<sup>3</sup> Kavli Institute for Cosmology, University of Cambridge, Cambridge CB3 0HA, United Kingdom

<sup>4</sup> Center for Field Theory and Particle Physics, Fudan University, 200438 Shanghai, People's Republic of China

<sup>5</sup> Department of Physics, Fudan University, 200438 Shanghai, People's Republic of China

<sup>6</sup> Department of Physics and Astronomy, University of California, Irvine, CA 92697-4575, United States of America

<sup>7</sup> Tsung-Dao Lee Institute, Shanghai Jiao Tong University, 201210 Shanghai, People's Republic of China

<sup>8</sup> School of Physics and Astronomy, Shanghai Jiao Tong University, 200240 Shanghai, People's Republic of China

<sup>9</sup> Centre for Theoretical Physics, Jamia Millia Islamia, New Delhi 110025, India

<sup>10</sup> Department of Astronomy and High Energy Physics, Kharazmi University, 15719-14911 Tehran, Iran

<sup>11</sup> School of Astronomy, Institute for Research in Fundamental Sciences, PO Box 19395-5531, Tehran, Iran

<sup>12</sup> International Center for Cosmology, Charusat University, Anand, Gujarat 388421. India

<sup>13</sup> Department of Mathematics and Statistics, Dalhousie University, Halifax, Nova Scotia B3H 3J5, Canada

<sup>14</sup> School of Mathematics, Statistics, and Computer Science, University of KwaZulu-Natal, Durban 4000, South Africa

Author to whom any correspondence should be addressed.

(cc)

Original Content from this work may be used under the terms of the Creative Commons Attribution 4.0 licence. Any further distribution of this work must maintain attribution to the author(s) and the title of the work, journal citation and DOI.

<sup>15</sup> International Centre for Space and Cosmology, Ahmedabad University, Ahmedabad, Gujarat 380009, India

- <sup>16</sup> Physics Department, State University of Tetovo, 1200 Tetovo, Macedonia
- <sup>17</sup> Department of Physics, College of Sciences, Shiraz University, Shiraz 71454, Iran
- <sup>18</sup> Biruni Observatory, College of Sciences, Shiraz University, Shiraz 71454, Iran
- <sup>19</sup> School of Physics, Institute for Research in Fundamental Sciences, PO Box 19395-5531, Tehran, Iran

<sup>20</sup> Department of Physics, University of Arizona, Tucson, AZ 85721, United States of America

<sup>21</sup> Physics Department, Eastern Mediterranean University, Famagusta 99628 North Cyprus via Mersin 10, Turkey

E-mail: sunny.vagnozzi@unitn.it

Received 20 October 2022; revised 26 April 2023 Accepted for publication 26 May 2023 Published 19 July 2023



#### Abstract

Horizon-scale images of black holes (BHs) and their shadows have opened an unprecedented window onto tests of gravity and fundamental physics in the strong-field regime. We consider a wide range of well-motivated deviations from classical general relativity (GR) BH solutions, and constrain them using the Event Horizon Telescope (EHT) observations of Sagittarius A\* (Sgr A\*), connecting the size of the bright ring of emission to that of the underlying BH shadow and exploiting high-precision measurements of Sgr A\*'s mass-todistance ratio. The scenarios we consider, and whose fundamental parameters we constrain, include various regular BHs, string-inspired space-times, violations of the no-hair theorem driven by additional fields, alternative theories of gravity, novel fundamental physics frameworks, and BH mimickers including well-motivated wormhole and naked singularity space-times. We demonstrate that the EHT image of Sgr A\* places particularly stringent constraints on models predicting a shadow size larger than that of a Schwarzschild BH of a given mass, with the resulting limits in some cases surpassing cosmological ones. Our results are among the first tests of fundamental physics from the shadow of Sgr A\* and, while the latter appears to be in excellent agreement with the predictions of GR, we have shown that a number of well-motivated alternative scenarios, including BH mimickers, are far from being ruled out at present.

Keywords: black holes, black hole shadows, tests of gravity, wormholes, naked singularities

(Some figures may appear in colour only in the online journal)

#### 1. Introduction

Black holes (BHs) are among the most extreme regions of space-time [1–3], and are widely believed to hold the key towards unraveling various key aspects of fundamental physics, including the behavior of gravity in the strong-field regime, the possible existence of new fundamental degrees of freedom, the unification of quantum mechanics (QM) and gravity, and the nature of space-time itself [4, 5]. We have now been ushered into an era where BHs and their observational effects are witnessed on a regular basis and on a wide range of scales. Perhaps the most impressive example in this sense are horizon-scale images of supermassive BHs (SMBHs), delivered through very long baseline interferometry (VLBI), and containing information about the space-time around SMBHs. The first groundbreaking horizon-scale SMBH images were delivered by the Event Horizon Telescope (EHT), a millimeter VLBI array with Earth-scale baseline coverage [6], which in 2019 resolved the near-horizon region of the SMBH M87\* [7–12], before later revealing its magnetic field structure [13, 14]. This was recently followed by the EHT's first images of Sagittarius A\* (Sgr A\*), the SMBH located at the Milky Way center [15–24].

The main features observed in VLBI horizon-scale images of BHs are a bright emission ring surrounding a central brightness depression, with the latter related to the BH shadow (see e.g. [25-34] for some of the earliest calculations in this sense [35-39], for more recent studies on the shapes of BH shadows and related observables, and higher-order photon rings in [40]). On the plane of a distant observer, the boundary of the BH shadow marks the apparent image of the photon region (the boundary of the region of space-time which supports closed spherical photon orbits)<sup>22</sup>, and separates capture orbits from scattering orbits: for detailed reviews on BH shadows, see e.g. [41–45]. Under certain conditions and after appropriate calibration, the radius of the bright ring can serve as a proxy for the BH shadow radius, with very little dependence on the details of the surrounding accretion flow  $[11, 12, 28, 46-52]^{23}$  This is possible if (i) a bright source of photons is present and strongly lensed near the horizon, and especially (ii) the surrounding emission regions is geometrically thick and optically thin at the wavelength at which the VLBI network operates. Most SMBHs we know of, including Sgr A\* and M87\* [46], operate at sub-Eddington accretion rates and are powered by radiatively inefficient advection-dominated accretion flows, as a result satisfying both the previous conditions.

The possibility of connecting the ring and BH shadow angular radii opens up the fascinating prospect of using BH shadows to test fundamental physics, once the BH mass-to-distance ratio is known [62, 63]. For a Schwarzschild BH of mass *M* located at distance *D*, the shadow is predicted to be a perfect circle of radius  $r_{\rm sh} = 3\sqrt{3}G_NM/c^2$ , therefore subtending (within the small angle approximation) an angular diameter  $\theta_{\rm sh} = 6\sqrt{3}G_NM/(c^2D) = 6\sqrt{3}\theta_{\rm g}$ , where

<sup>&</sup>lt;sup>22</sup> In the case of spherically symmetric space-times, such as with Schwarzschild BHs, the photon region is simply referred to as photon sphere.

 $<sup>^{23}</sup>$  Some concerns on whether VLBI horizon-scale images are really seeing the image of the BH shadow surrounded by the photon ring were raised in [53–57], but have been addressed in most of the above works. See also theoretical concerns on the possibility of testing gravity using BH images raised in [58]. Finally, see [59–61] for further relevant work.

 $\theta_{\rm g} = G_N M/(c^2 D)$  is the angular gravitational radius of the BH<sup>24</sup>. For Kerr BHs, the shadow is slightly asymmetric along the spin axis, with  $r_{\rm sh}$  being marginally smaller compared to the Schwarzschild case, but still depending predominantly on the BH mass M, and only marginally on the (dimensionless) spin  $a_{\star}$  and observer's inclination angle *i*. Crucially, the values of  $\theta_{\rm sh}$  and  $r_{\rm sh}$  can change considerably for other metrics, including those describing BHs in alternative theories of gravity, the effects of corrections from new physics possibly leading to violations of the no-hair theorem (NHT) [64–68], or 'BH mimickers', i.e. (possibly horizonless) compact objects other than BHs [69–80]. This paves the way towards tests of fundamental physics from the angular sizes of BH shadows of known mass-to-distance ratio.

With the EHT image of M87<sup>\*</sup>, the prospects of testing fundamental physics with BH shadows have become reality, as demonstrated by a large and growing body of literature devoted to such tests, mostly focusing on the size of M87<sup>\*</sup>'s shadow, but in some cases also considering additional observables such as the shadow circularity and axis ratio, as well as future prospects (see e.g. [81–182]). The new horizon-scale image of Sgr A<sup>\*</sup> delivered by the EHT offers yet another opportunity for performing tests of gravity and fundamental physics in the strong-field regime, which we shall here exploit. Even though such tests have already been performed with M87<sup>\*</sup>, there are very good motivations, or even advantages, for independently carrying them out on the shadow of Sgr A<sup>\*</sup> (see also [183–187]):

- Sgr A\*'s proximity to us makes it significantly easier to calibrate its mass and distance, and therefore its mass-to-distance ratio, whose value is crucial to connect the observed angular size of its ring to theoretical predictions for the size of its shadow within different fundamental physics scenarios. This is a distinct advantage with respect to M87\*, whose mass is the source of significant uncertainties, with measurements based on stellar dynamics [188] or gas dynamics [189] differing by up to a factor of 2 (see also section IIIA of [190]).
- 2. Sgr A\*'s mass in the  $\mathcal{O}(10^6)M_{\odot}$  range is several orders of magnitude below M87\*'s mass in the  $\mathcal{O}(10^9)M_{\odot}$  range, allowing us to probe fundamental physics in a completely different and complementary strong curvature regime (see more below and in figure 1). For the same reason, Sgr A\*'s shadow can potentially set much tighter constraints than M87\*'s shadow on dimensionful fundamental parameters which scale as a positive power of mass in the units we adopt (as is the case for several BH charges or 'hair parameters').
- 3. Finally, independent constraints from independent sources are always extremely valuable in tests of fundamental physics, and there is, therefore, significant value in performing such tests on Sgr A\* independently of the results obtained from M87\*. As we shall explicitly show, unlike M87\*, the EHT image of Sgr A\* sets particularly stringent constraints on theories and frameworks which predict a shadow radius *larger* than  $3\sqrt{3}M$ . We will consider many such examples in this work.

To elaborate more on point 2. Above, we consider the Baker–Psaltis–Skordis 'curvaturepotential' diagram introduced in [191] to place, link, and compare on a common set of axes the regimes explored by a wide range of tests of gravity. Specifically, on the x axis we plot  $\log_{10} \epsilon$ , with  $\epsilon$  the magnitude of the Newtonian gravitational potential, quantifying deviations

<sup>&</sup>lt;sup>24</sup> In what follows, we shall work in units where  $4\pi\epsilon_0 = \hbar = c = G_N = 1$ . At times we will find it convenient to work in units of BH mass, and therefore will set M = 1, while rescaling all other parameters accordingly.

from the Minkowski metric. For the relevant case of a test particle orbiting a central object of mass M (such as a BH) at a distance r,  $\epsilon$  is given by (re-introducing SI units):

$$\epsilon \equiv \frac{G_N M}{c^2 r} \,. \tag{1}$$

The larger  $\epsilon$ , the stronger the gravitational field probed by an observer or a test particle, with the limit  $\epsilon \rightarrow 0.5$  corresponding to a test particle approaching the event horizon—loosely speaking,  $\epsilon$  measures the 'strength' of the metric, and in the following we will refer to it as 'potential' (note that  $\epsilon$  is simply the inverse of the radial coordinate in units of gravitational radius, and is sometimes referred to as 'compactness parameter'). On the *y* axis we instead plot  $\log_{10} \xi$ , with  $\xi$  the (square root of the) Kretschmann scalar which, for a Schwarzschild BH of mass *M*, reads:

$$\xi = (R_{\mu\nu\rho\sigma}R^{\mu\nu\rho\sigma})^{\frac{1}{2}} = \frac{4\sqrt{3}G_NM}{c^2r^3}.$$
(2)

The larger  $\xi$ , the larger the total curvature due to both local and distant masses—loosely speaking,  $\xi$  measures the 'strength' of the Riemann tensor, and in the following we will refer to it as 'curvature'. For illustrative purposes, given that the EHT probes the space-time around the BH photon ring, we compute equations (1) and (2) at  $r = 3GM/c^2$ , which is the location of the photon ring for a Schwarzschild BH, finding:

$$\epsilon = \frac{1}{3}, \quad \xi \approx 1.2 \times 10^{-11} \left(\frac{M}{M_{\odot}}\right)^{-2} \text{ cm}^{-2}.$$
 (3)

While the shadows of Sgr A\* and M87\* probe (by construction) the same potential regime  $(\log_{10} \epsilon \approx -0.48)$ , they probe two completely different curvature regimes, the former in the  $\xi \sim \mathcal{O}(10^{-25}) \,\mathrm{cm}^{-2}$  region and the latter in the much weaker  $\xi \sim \mathcal{O}(10^{-31}) \,\mathrm{cm}^{-2}$  region. We illustrate this visually in figure 1, where we show the region of potential-curvature parameter space probed by both Sgr A\* and M87\*, as indicated by the respective icons. Following [191, 192], the region bounded by the green dashed curve encloses other potential more distant EHT sources, for which the expected resolution in units of gravitational radii is of course much lower compared to Sgr A<sup>\*</sup> and M87<sup>\*</sup> (hence the lower values of  $\epsilon$  probed). Finally, the magenta points correspond to the S-stars orbiting close to Sgr A\*, following [191, 193]. Figure 1 clearly allows us to appreciate how the shadows of Sgr A\* and M87\* probe gravity and, more generally, fundamental physics, in two completely different strong gravitational field regimes, reaffirming once more the value in performing the tests we shall conduct in this paper. We note that the tests we shall perform are parametric tests of theories of gravity and fundamental physics scenarios. In other words, such tests are performed either on specific known metric solutions of such theories, or anyhow on explicit metrics which are introduced phenomenologically (and whose origin in some cases is well rooted into specific underlying theories or frameworks). Another possibility, which we shall not pursue here, is to perform similar tests on theory-agnostic metrics which, while not arising from any specific theory, can be mapped to different theories and thus may effectively constrain a number of theories simultaneously (see e.g. [52, 194, 195] in a related context).

The rest of this paper is then organized as follows. In section 2 we discuss the methodology and assumptions entering into the computation of the sizes of BH shadows, and give a very brief overview of the fundamental physics scenarios we consider. Section 3 is divided into a large number of subsections (from sections 3.1 to 3.41), one for each of the models and



**Figure 1.** Regimes probed by various gravitational systems of interest, represented on the Baker–Psaltis–Skordis 'curvature-potential' diagram. The shadows of Sgr A\* and M87\* are indicated by the respective icons, the region bounded by the green dashed curve encloses other potential more distant EHT sources, and the magenta points correspond to the S-stars orbiting close to Sgr A\*. Credits for Sgr A\* and M87\*'s images are due to the © EHT Collaboration and © ESO (sources: 'Astronomers Reveal First Image of the Black Hole at the Heart of Our Galaxy' and 'First Image of a Black Hole' respectively—both images are licensed under a Creative Commons Attribution 4.0 International License).

fundamental physics scenarios considered (which in some cases contain various sub-classes), for which we report constraints obtained adopting the previously discussed methodology. In section 4 we provide a brief overall discussion of our results, and a brief overview of complementary probes and future prospects. Finally, in section 5 we draw concluding remarks and outline future directions.

#### 2. Methodology

The methodology we shall follow relies on comparing the observed angular radius of the ringlike feature in the EHT horizon-scale image of Sgr A\* with the theoretically computed angular radius of the shadows of BHs (or other alternative compact objects) within each fundamental physics scenario we consider, with prior knowledge of Sgr A\*'s mass-to-distance ratio. Requiring consistency between the two quantities, within the uncertainty allowed by the EHT observations, allows us to set constraints on the parameters describing the space-time in question. This simple methodology has been discussed several times in the past (see for example [62, 63]) and has been successfully applied to the EHT image of M87\* by the EHT collaboration themselves in [144], though for reasons discussed earlier in section 1, the application to Sgr A\* is in principle more robust. Note that, much as all related works in the literature, we will be using the observationally provided values for the mass-to-distance ratio and shadow radius implicitly assuming that their inferred values are independent of both the underlying space-time metric and/or theory of gravity. This allows us to use the same values for these quantities when moving across the different scenarios we consider.

Survey	$M( imes 10^6 M_{\odot})$	D(kpc)	References
Keck VLTI	$\begin{array}{r} 3.951  \pm  0.047 \\ 4.297  \pm  0.012  \pm  0.040 \end{array}$	$\begin{array}{c} 7.953  \pm  0.050  \pm  0.032 \\ 8.277  \pm  0.009  \pm  0.033 \end{array}$	[201] [202]

Table 1. Mass and distance to Sgr A\* as inferred from the Keck and VLTI instruments.

To apply this methodology, we require two ingredients. The first is Sgr A\*'s mass-todistance ratio. The mass and distance to Sgr A\*, M, and D, have been studied in detail over the past decades exploiting stellar cluster dynamics, and in particular the motion of S-stars, individual stars resolved within 1" of the Galactic Center. A significant role has been played by S0-2 which, with a K-band magnitude of  $\sim$ 14, period of  $\sim$ 16 years, and semimajor axis of  $\sim$ 125 mas, is the brightest star with a relatively close orbit and short period close to the Galactic Center. Its orbit has been exquisitely tracked by two sets of instruments/teams<sup>25</sup>, which we shall refer to as 'Keck' and 'VLTI' (standing for 'Very Large Telescope Interferometer'), respectively. Following [20], we adopt the (correlated) mass and distance estimates given in table 1 [201, 202], where uncertainties are quoted at 68% confidence level (C.L.), reporting (where available) systematic uncertainties. We refer the reader to section 2.1 of [20] for detailed discussions on these measurements.

The second ingredient we require is a calibration factor connecting the size of the bright ring of emission with the size of the corresponding shadow, which quantifies how safe it is to use the size of the bright ring of emission as a proxy for the shadow size. This calibration factor depends on the near-horizon physics of image formation and, as already anticipated earlier, is expected to be very close to unity for optically thin, geometrically thick radiatively inefficient advection-dominated accretion flows, such as the one surrounding Sgr A<sup>\*</sup>. In practice, the calibration factor accounts multiplicatively for various sources of uncertainty, ranging from formal measurement uncertainties, to fitting/model uncertainties, to theoretical uncertainties pertaining to the emissivity of the plasma (see section 3 of [20]).

Detailed studies of various sources of uncertainty have been conducted by the EHT in [20] and used to determine the above calibration factor. Folding in the calibration factor with uncertainties in Sgr A\*'s mass-to-distance ratio, and the angular diameter of Sgr A\*'s bright ring of emission, the EHT inferred  $\delta$ , the fractional deviation between the inferred shadow radius  $r_{\rm sh}$  and the shadow radius of a Schwarzschild BH of angular size  $\theta$  ( $\theta_{\rm sh} = 3\sqrt{3}\theta_{\rm g}$ ),  $r_{\rm sh \ Schwarzschild} = 3\sqrt{3}M$ . In practice,  $\delta$  is given by:

$$\delta \equiv \frac{r_{\rm sh}}{r_{\rm sh,Schwarzschild}} - 1 = \frac{r_{\rm sh}}{3\sqrt{3}M} - 1.$$
(4)

The inferred value of  $\delta$  depends on the mass-to-distance ratio assumed, with the Keck and VLTI measurements resulting in the following estimates [20]:

- Keck: δ = -0.04<sup>+0.09</sup><sub>-0.10</sub>;
  VLTI: δ = -0.08<sup>+0.09</sup><sub>-0.09</sub>;

with a good agreement between the two bounds.

<sup>25</sup> The motion of the S0-2 star has been used to set constraints on various aspects of fundamental physics, see e.g. [196-200].

In order to be as conservative as possible, and to simplify our later discussion, we take the average of the Keck- and VLTI-based estimates of  $\delta$ , treating them as uncorrelated, as they are obtained from two different instruments/surveys. This leads to the following estimate of  $\delta$  which we shall adopt throughout this work:

$$\delta \simeq -0.060 \pm 0.065,$$
 (5)

which, under the assumption of Gaussianity (itself supported by the shapes of the posteriors shown in figure 12 of [20]), trivially leads to the following  $1\sigma$  and  $2\sigma$  intervals for  $\delta$ :

$$-0.125 \lesssim \delta \lesssim 0.005 \quad (1\sigma),$$
  
$$-0.19 \lesssim \delta \lesssim 0.07 \quad (2\sigma). \tag{6}$$

As is clear from the previous Keck- and VLTI-based estimates which we have averaged, there is overall a very slight preference for Sgr A\*'s shadow being *marginally smaller* than the prediction  $3\sqrt{3}M$  for a Schwarzschild BH of given mass, with a  $\leq 68\%$  C.L. preference for  $\delta < 0$ . At the level of shadow radius  $r_{\rm sh}$  and once more assuming Gaussian uncertainties, after expressing  $r_{\rm sh}/M = 3\sqrt{3}(\delta + 1)$  by inverting equation (4), it is easy to see that the bounds reported in equation (6) translate to the following  $1\sigma$  constraints on  $r_{\rm sh}/M$ :

$$4.55 \lesssim r_{\rm sh}/M \lesssim 5.22\,,\tag{7}$$

as well as the following  $2\sigma$  constraints:

$$4.21 \lesssim r_{\rm sh}/M \lesssim 5.56. \tag{8}$$

These constraints are in good agreement with those reported in [20], though with slightly smaller uncertainties (by a factor of  $\approx \sqrt{2}$ ) as a result of having taken the average between the Keck- and VLTI-based estimates. In what follows, we shall use the bounds in equations (7) and (8) to constrain the parameters governing space-times beyond the Schwarzschild BH.

For simplicity, throughout this work we shall restrict ourselves to static spherically symmetric metrics, i.e. neglecting the effect of spin. The reason behind this choice is two-fold. First, the effect of spin on the shadow radius is small: we illustrate this in figure 2 for the case of a Kerr BH, where we plot the predicted shadow radius as a function of the dimensionless spin  $a_{\star}$ , for different values of the observer's inclination angle *i*. Clearly, the effect is more pronounced for high inclination angles (i.e. for almost edge-on viewing), but remains small ( $\leq 12\%$ ).<sup>26</sup>

Second, and perhaps most importantly, there is at present no clear consensus on the value of Sgr A\*'s spin and inclination angle. The EHT images are in principle consistent with large spin and low inclination angle, but are far from being inconsistent with low spin and large inclination angle [15, 19]. To be precise, however, this is strictly speaking only true for the Kerr metric, which has been assumed in deriving these results. Independent works based on radio, infrared, and x-ray emission, as well as millimeter VLBI, exclude extremal spin  $(1 - a_* \ll 1)$ , but have been unable to place strong constraints otherwise [203–205]. Estimates based on semi-analytical models, magnetohydrodynamics simulations, or flare emissions, have reported constraints across a wide range of spin values [206, 207]. One of the most recent dynamical estimates of Sgr A\*'s spin was reported in [208] by Fragione and Loeb, based on the impact on the orbits of the S-stars of frame-dragging precession, which would tend to erase the orbital

 $<sup>^{26}</sup>$  It is worth remarking that this is strictly speaking true for the Kerr space-time. We will adopt the assumption (implicitly also made in [144]) that the effect of spin remains small even in alternative metrics. Whether this is actually true should be checked on a case-by-case basis, although for many of the cases we will study, the rotating version of the metric is not available.



**Figure 2.** Shadow radius in units of BH mass,  $r_{\rm sh}/M$ , for a Kerr BH (solid curves), as a function of the dimensionless spin  $a_{\star}$ , for various values of the observer's inclination angle *i* as given by the color coding (with  $i = 0^{\circ}$  and  $i = 90^{\circ}$  corresponding to face-on and edge-on viewing respectively). The dashed cyan line indicates the size of a Schwarzschild BH, i.e. the limit  $a_{\star} \rightarrow 0$ , for which  $r_{\rm sh}/M = 3\sqrt{3}$ . It is clear that the effect of spin on the BH shadow radius is small, particularly at low values of the spin.

planes in which the S-stars formed and are found today: observations of the alignment of the orbital planes of the S-stars today require Sgr A\*'s spin to be very low,  $a_{\star} \leq 0.1$  (see also [209]). Some of these estimates rely on an assumed metric, although it is worth pointing out that the result of [208] does not. Constraints on the inclination angle are even more uncertain. The inconsistency across different estimates of Sgr A\*'s spin prompts us to a conservative approach where we neglect the effect of spin, while taking the very recent estimate of [208] as indication that the spin may be low: for  $a_{\star} \leq 0.1$ , as figure 2 clearly shows, the effect of spin on the shadow radius is negligible at all inclination angles.

Finally, simple shadow-based observables which are most sensitive to the spin are those related to the shadow's circularity, such as the deviation from circularity  $\Delta C$  discussed in [73] (and adopted by several works which used the shadow of M87\* to constrain fundamental physics), which for a Kerr BH should be  $\leq 4\%$ , increasing with spin and inclination angle. However, the sparse interferometric coverage of the EHT's 2017 observations of Sgr A\*, the associated significant uncertainties in circularity measurements, and the short variability timescale of Sgr A\*, prevented the collaboration from quantifying the circularity of the shadow based on observational features [18, 20, 210]. In closing, we note that most of the space-times considered in [20] are also spherically symmetric, for reasons similar to the ones we outlined, providing an independent validation of our approach (although with all the caveats discussed above).

#### 2.1. Black hole shadow radius in spherically symmetric space-times

Here we briefly review the calculation of the shadow radius in spherically symmetric spacetimes. If such a space-time possesses a photon sphere, the gravitationally lensed image thereof as viewed by an observed located at infinity will constitute the BH shadow<sup>27</sup>. Let us consider a generic static, spherically symmetric, asymptotically flat space-time, i.e. one admitting a global, non-vanishing, time-like, hypersurface-orthogonal Killing vector field, with no off-diagonal components in the matrix representation of the metric tensor. Without loss of generality, the space-time line element in Boyer–Lindquist coordinates can be expressed as:

$$ds^{2} = -A(r)dt^{2} + B(r)dr^{2} + C(r)d\Omega^{2}, \qquad (9)$$

where  $d\Omega$  is the differential unit of solid angle<sup>28</sup>. In the following, we shall henceforth refer to the function  $A(r) = -g_{tt}(r)$  as the 'metric function', and occasionally to  $C(r) = g_{\theta\theta} = g_{\phi\phi}/\sin^2\theta$  as the 'angular metric function' (though we note that the latter terminology is not a widespread one). If the space-time is asymptotically flat,  $A(r) \to 1$  as  $r \to \infty$  in equation (9). Note also that in most cases of interest  $C(r) = r^2$ , although here we choose to keep the discussion as general as possible.

Let us then define the function h(r) (see e.g. [43]):

$$h(r) \equiv \sqrt{\frac{C(r)}{A(r)}}.$$
(10)

If the metric in equation (9) possesses a photon sphere, the radial coordinate thereof,  $r_{ph}$ , is given by the solution to the following implicit equation:

$$\frac{\mathrm{d}}{\mathrm{d}r}\left[h^2(r_{\mathrm{ph}})\right] = 0\,,\tag{11}$$

which can be arranged to the following form:

$$C'(r_{\rm ph})A(r_{\rm ph}) - C(r_{\rm ph})A'(r_{\rm ph}) = 0.$$
(12)

Here, the prime denotes a derivative with respect to *r*. If, as in the majority of cases we will consider, the angular metric function is  $C(r) = r^2$ , then equation (12) reduces to the following well-known expression:

$$A(r_{\rm ph}) - \frac{1}{2}r_{\rm ph}A'(r_{\rm ph}) = 0.$$
(13)

The shadow radius  $r_{\rm sh}$  corresponds to the gravitationally lensed image of the surface defined by  $r_{\rm ph}$ , and is therefore given by (see e.g. [41–43, 135, 211]):

$$r_{\rm sh} = \sqrt{\frac{C(r)}{A(r)}}\Big|_{r_{\rm ph}}.$$
(14)

For the 'standard' angular metric function  $C(r) = r^2$  equation (14) reduces to the following well-known expression:

$$r_{\rm sh} = \frac{r}{\sqrt{A(r)}} \bigg|_{r_{\rm ph}}.$$
(15)

<sup>&</sup>lt;sup>27</sup> While a space-time with a photon sphere can cast a shadow, the existence of a photon sphere is strictly speaking not a mandatory requirement for a space-time to cast a shadow, see e.g. [74].

 $<sup>^{28}</sup>$  We are implicitly assuming that *r* matches the quantity *r* provided by astrophysical determinations (which is mathematically consistent with our later treatment of the shadow radius).

In the case of the Schwarzschild BH, whose metric function is A(r) = 1 - 2M/r and whose angular metric function is  $C(r) = r^2$  [1], combining equations (13) and (15) leads to the well-known result  $r_{\rm sh} = 3\sqrt{3}M$ . For obvious symmetry reasons, the BH shadow for a spherically symmetric space-time is a circle of radius  $r_{\rm sh}$  on the plane of a distant observer, irrespective of the inclination angle.

Note that the computation we have outlined is strictly valid for *asymptotically flat* spacetimes only. For other metrics, including but not limited to those matching on to the cosmological accelerated expansion at large distances and hence asymptotically de Sitter, the size of the BH shadow can explicitly depend on the radial coordinate of the distant observer, and on whether the observer is static or comoving (see e.g. [43, 212] for extensive discussions on this point). An example of such a situation occurs with the Kottler metric [213]. Given Sgr A\*'s proximity to us, for practical purposes we focus on a static observer (i.e. one which is not in the Hubble flow with respect to the BH) located at a distance  $r_0$ . For simplicity we also focus on the case where the angular metric function is  $C(r) = r^2$ . Then, the angular size of the BH shadow  $\alpha_{sh}$  is given by (see e.g. [43]):

$$\sin^2 \alpha_{\rm sh} = \frac{r_{\rm ph}^2}{A(r_{\rm ph})} \frac{A(r_{\rm O})}{r_{\rm O}^2} \,. \tag{16}$$

In the physically relevant small-angle approximation, it is easy to see that the shadow size is given by (see [43] for more detailed discussions):

$$r_{\rm sh} = r_{\rm ph} \sqrt{\frac{A(r_{\rm O})}{A(r_{\rm ph})}}.$$
(17)

The explicit dependence of the shadow size on the observer's position is clear in equation (17). It is trivial to see that for a distant observer viewing a BH described by an asymptotically flat metric, equation (17) reduces to equation (15), since  $A(r_0) \approx 1$  as long as the observer is sufficiently distant from the BH. For instance, for the Schwarzschild metric this condition is satisfied as long as  $r_0 \gg M$ , i.e. the distance between the BH and the observer is much larger than the BH gravitational radius. This condition is satisfied for the case of Sgr A\*, whose distance from us of  $\approx 8 \text{ kpc}$  should be compared to its gravitational radius  $r_g \approx \mathcal{O}(10^{-7}) \text{ pc}.$ 

As a further caveat, it is also worth noting that the computation outlined in equations (10)–(15) does not (necessarily) hold in theories with electromagnetic Lagrangian other than the Maxwell one (for instance non-linear electrodynamics (NLED)), since in this context photons do not necessarily move along null geodesics of the metric tensor, but along null geodesics of an effective geometry [214, 215], a fact which has only been recently appreciated in the literature. We will consider certain classes of NLED theories in this work, carefully accounting for the effective geometry.

As a final caveat, it is worth pointing out that the 'mathematical' result for the shadow radius obtained through the procedure discussed previously is 'astrophysically' relevant only if the resulting shadow size is located outside the event horizon for a BH, or the throat for a wormhole (WH). In the following, we have explicitly checked that this is indeed the case for all space-times discussed, or added caveats as appropriate.

In what follows, we will compute the shadow radius of various space-times of interest, and then compare them to the image of Sgr A<sup>\*</sup> by imposing the bounds of equations (7) and (8)<sup>29</sup>.

<sup>&</sup>lt;sup>29</sup> For each space-time, we typically explicitly provide only  $A(r) = -g_{tt}$  [and C(r) if  $\neq r^2$ ], as this is the metric component directly entering in the shadow radius calculation. For most of the space-times we study, the condition  $g_{tt}g_{rr} = -1$  is fulfilled: as discussed in [216], this occurs if and only if the radial null-null components of  $R_{\mu\nu}$  vanish

These, as discussed earlier, already take into account various observational, theoretical, and modeling uncertainties, including the potential multiplicative offset between the radius of the bright ring of emission and the underlying shadow radius. We will show that the EHT image of Sgr A\* sets particularly stringent constraints on theories and frameworks which predict a shadow radius *larger* than the Schwarzschild radius  $3\sqrt{3}M$ . Although the effect of new physics in many of the cases we will consider is that of decreasing the shadow radius with respect to the Schwarzschild case, a few will actually turn out to increase it, and these will be the models whose parameters are most tightly constrained. The key difference between these two classes of scenarios can easily be understood in terms of the effective potential experienced by photons which, assuming spherical symmetry and considering motion in the equatorial plane  $(\theta = \pi/2, \text{ which simplifies the calculations and is consistent with the assumed symmetries), is proportional to <math>V_{\text{eff}}(r) \propto A(r)/r^2$ . The shadow radius is then directly related to the maximum of the effective potential, and this can easily be shown to move towards the left/right within the scenarios leading to a decrease/increase of the shadow radius with respect to the Schwarzschild case.

#### 2.2. Brief overview and motivation of scenarios and space-times considered

Before moving on to actually performing our analysis, we will give a very brief overview of the space-times we consider, along with motivation for moving beyond the Schwarzschild space-time. We will consider a wide range of gravity theories, fundamental physics scenarios, and space-times beyond those of the Schwarzschild BH, for which  $r_{\rm sh} = 3\sqrt{3}M$ . The NHT [64–68] states that the only possible stationary, axisymmetric, and asymptotically flat BH solutions of the 4D electrovacuum Einstein–Maxwell equations are described by the Kerr–Newman family of metrics [217, 218]. Recall that the latter describes electrically charged, rotating BHs, and reduces to the Kerr metric in the absence of electric charge, to the Reissner–Nordström (RN) metric in the absence of rotation, and finally to the Schwarzschild metric in the absence of both charge and rotation. As John Archibald Wheeler phrased it, *'black holes have no hair'*, with the term 'hair' referring generically to parameters other than the BH mass M, spin J, and electric charge Q, which are required for a complete description of the BH solution.

At present, there is no sign of tension between the NHT and observations of astrophysical BHs. Why, then, is going beyond the Kerr–Newman family of metrics a well-motivated endeavor? The first reason is tied to the fact that a wide variety of theoretical and observational issues hint towards the possibility that our understanding of gravity as provided by general relativity (GR) is likely to be incomplete. These hints range from the quest for unifying gravity and QM, to the conflict between unitary evolution in QM and Hawking BH radiation as encapsulated in the BH information paradox [219], and finally to cosmological observations requiring the existence of dark matter (DM) and dark energy, likely a phase of cosmic inflation, and possibly a phase of early dark energy around the time of recombination. In the presence of new physics which eventually will address these issues, it is perfectly reasonable to assume that the NHT may only be an approximation, valid to the current level of precision. In fact, several

or, equivalently, if the restriction of  $R_{\mu\nu}$  to the *t*-*r* subspace is proportional to  $g_{\mu\nu}$  (which on solutions of the Einstein equations implies that the radial pressure is equal to minus the energy density). The only space-times we study for which this condition is violated are: the Morris–Thorne traversable wormhole (section 3.10), the Damour–Solodukhin wormhole (section 3.11), the Joshi–Malafarina–Narayan naked singularity (section 3.13), the naked singularity surrounded by a thin shell of matter (section 3.14), the BH in Clifton–Barrow *f*(*R*) gravity (section 3.18), the Sen BH (section 3.23), the Einstein–Maxwell-dilaton-1 BH (section 3.25), the BH in loop quantum gravity (section 3.30), the DST BH (section 3.33), the BH in bumblebee gravity (section 3.41.2), and the Casimir wormhole (section 3.41.4).

theoretical approaches towards addressing the above issues posit the existence of new fields (particularly scalar fields) or modifications to GR (often introducing effective scalar degrees of freedom), see e.g. [220–253] for an inevitably incomplete selection of examples. All of these scenarios can very naturally lead to violations of the NHT<sup>30</sup>. Moving beyond astrophysical systems, the study of controlled violations of the NHT is well-motivated by developments in our understanding of the gauge/gravity duality [257], with important applications to various condensed matter systems [258], including holographic superconductors [259] and quantum liquids [260].

The second reason is possibly even more fundamental, and is related to the well-known Penrose–Hawking singularity theorems, i.e. the fact that in GR continuous gravitational collapse leads to the inevitable, but at the same time arguably undesirable, appearance of singularities [2, 261]. For instance, the Kerr–Newman family of metrics possesses a well-known physical (non-coordinate) singularity at r = 0. The cosmic censorship conjecture [262, 263] notwithstanding, the mere existence of singularities has prompted a long-standing search for 'regular' BH solutions, which regularize the central singularity (see e.g. [264–318] for examples). Therefore, testing the metrics of regular BHs, or BH mimickers which address at least in part the existence of singularities, is a very well-motivated direction, particularly at present time with the availability of horizon-scale BH images. Note that many of these regular BH solutions can be obtained as solutions to the Einstein field equations coupled to a suitable NLED source. Finally, as the nature of DM, dark energy, and the origin of structure (whether through a phase of inflation or an alternative mechanism) is not fully understood at present, one should be open-minded to the possibility that one or more of these phenomena may be connected to potential modifications of GR.

With these considerations in mind, in this work we will consider a wide range of spacetimes beyond the Schwarzschild BH, testing them against the EHT horizon-scale image of Sgr  $A^*$ . The space-times we consider broadly speaking fall within these (by no means mutually exclusive) categories:

- Regular BHs, whether arising from specific theories or constructed in a phenomenological setting;
- 2. BHs in modified theories of gravity, modified electrodynamics, and string-inspired settings;
- 3. BHs in theories with additional matter fields (usually scalar fields), typically bringing about violations of the NHT;
- 4. BH mimickers such as WHs (or effective WH geometries) and naked singularities (NSs);
- Modifications induced by novel fundamental physics frameworks (e.g. generalized or extended uncertainty principles (EUPs), non-commutative geometries (NCGs), or entropy laws beyond the Bekenstein–Hawking entropy).

Of course, several space-times we will consider fall within more than one of the above categories at the same time (especially 1. and 2., or 1. and 3.). Most of these space-times are described by one or more extra parameters, which in most cases we shall generically refer to

<sup>&</sup>lt;sup>30</sup> Note that, although one would in principle expect the effects of dark energy and modified gravity models invoked to explain cosmic acceleration to appear on scales  $\gtrsim$ 150Mpc, whereas Sgr A\* is 8kpc distant from us, in practice these models almost always result in unwanted effects (e.g. fifth forces) on local scales, unless screening mechanisms are invoked [254]. These screening mechanisms typically kick in at scales intermediate between cosmological [ $\mathcal{O}(\text{Gpc})$ ] and local [ $\mathcal{O}(\text{pc})$ ] ones, so quite naturally in the  $\mathcal{O}(\text{kpc})$  range, lending motivation to the search for novel phenomenological signatures in this regime. In addition, it is worth noting that several 'quintessential inflation' models exist wherein inflation and cosmic acceleration are unified as the result of a single scalar degree of freedom [255, 256].

as 'hair', 'hair parameters', or 'charges', in the latter case following the language adopted by [20, 144].

Before moving on to the results, one final clarification is in order. In all the cases we will consider, the hair parameters discussed above essentially fall into one of two categories. In the first case, they are 'universal', i.e. the same for each BH in the Universe regardless of its other hairs (mass, spin, electric charge). In the second case, they are 'specific', i.e. they can vary from BH to BH. With a slight abuse of language, we shall refer to these as being *universal hairs* and *specific hairs* respectively: note that the two should not be confused with the concepts of primary hair and secondary hair, although they are to some extent related.

An example of universal hair could be one which is exclusively determined by a parameter (e.g. a coupling) of the underlying Lagrangian of the theory (which is therefore uniquely determined), or by another fundamental parameter introduced at a phenomenological level. A very simple example of specific hair is the electric charge of a BH appearing in the well-known RN metric: this is a conserved charge which essentially emerges as an integration constant, and which can therefore take different values from BH to BH. As far as we can tell, the distinction between these two types of hairs has not been sufficiently emphasized in the literature on tests of fundamental physics from BH shadows.

We note that the comparison between constraints on specific hairs obtained from BH shadows and other astrophysical probes (for instance gravitational waves, stellar motions, and cosmology) is meaningful only if the constraints are referred to the same source. On the other hand, in the case of universal hairs such a comparison can be made even without reference to any specific source. In what follows, for each BH solution we will study, we will clearly indicate whether the additional hair parameter which controls deviations from the Schwarzschild metric is a specific hair or an universal hair.

#### 3. Results

#### 3.1. Reissner-Nordström black hole and naked singularity

The first geometry we consider beyond the Schwarzschild BH, and arguably the simplest extension thereof, is the RN metric describing an electrically charged, non-rotating BH. Explicitly introducing the BH mass (which we will later set to M = 1 for convenience), and recalling that we are setting  $4\pi\epsilon_0 = \hbar = c = G_N = 1$ , the metric function of the RN BH with electric charge Q is given by [319–322]:<sup>31</sup>

$$A(r) = 1 - \frac{2M}{r} + \frac{Q^2}{r^2},$$
(18)

where Q describes a specific hair. Henceforth, if not explicitly stated, the angular metric function will be implicitly assumed to be given by  $C(r) = r^2$ . Setting M = 1,<sup>32</sup> it is easy to show that this space-time possesses an event horizon only for  $Q \leq 1$ . However, even for larger values of the electric charge the space-time, which is now a NS, can still possess a photon sphere (and thereby cast a shadow-like feature), provided  $1 < Q \leq \sqrt{9/8} \approx 1.06$ : this region of parameter space describes what is commonly referred to as the 'RN naked singularity'.

<sup>&</sup>lt;sup>31</sup> For completeness, given the importance of the Reissner–Nordström space-time, we note that the metric function in SI units reads  $A(r) = 1 - 2G_N M/c^2 r + GQ^2/4\pi\epsilon_0 c^4 r^2$ .

<sup>&</sup>lt;sup>32</sup> This operation effectively corresponds to rescaling the Boyer–Lindquist radial coordinate and electric charge as  $\tilde{r} = r/M$  and  $\tilde{Q} = Q/M$  respectively. Throughout this work we will drop the tildes, and whether we are referring to the original or rescaled quantities should be evident from the context.



**Figure 3.** Shadow radius  $r_{sh}$  of the Reissner–Nordström black hole (solid curve) and naked singularity (dashed curve) with metric function given by equation (18) and in units of the BH mass *M*, as a function of the normalized electric charge Q/M, as discussed in section 3.1. The dark gray and light gray regions are consistent with the EHT horizon-scale image of Sgr A<sup>\*</sup> at 1 $\sigma$  and  $2\sigma$  respectively, after averaging the Keck and VLTI mass-to-distance ratio priors for Sgr A<sup>\*</sup> (see equations (7) and (8)). The red regions are instead excluded by the same observations at more than  $2\sigma$ .

Using equations (13) and (15), we straightforwardly obtain the following expression for the radius of the shadow cast by the RN space-time, which is valid within both the BH and NS regimes:

$$r_{\rm sh} = \frac{\sqrt{2} \left(3 + \sqrt{9 - 8Q^2}\right)}{\sqrt{4 + \frac{\sqrt{9 - 8Q^2} - 3}{Q^2}}}.$$
(19)

It is at first glance not obvious that equation (19) does indeed reduce to the Schwarzschild result of  $r_{\rm sh} = 3\sqrt{3}$  in the  $Q \rightarrow 0$  limit, due to the apparent divergence in the denominator. However, the proper way of taking this limit is to Taylor-expand the denominator: when doing so, one indeed finds that the denominator of equation (19) tends to  $\sqrt{8/3}$ , whereas the numerator straightforwardly tends to  $6\sqrt{2}$ , leading overall to the correct limit of  $3\sqrt{3}$ .

The evolution of the shadow radius as a function of the electric charge is shown in figure 3, for both the BH (solid) and NS (dashed) regimes of the RN space-time, alongside the observational constraints imposed by the EHT image of Sgr A<sup>\*</sup> (equations (7) and (8)). We see that as the electric charge increases, the shadow radius decreases, which can be understood by studying how the electric charge affects the effective potential felt by test particles. This is a feature which is common to a number of extensions of the Schwarzschild metric, as we shall see throughout the paper: most extensions will lead to a shadow radius which *decreases* with increasing charge/hair parameter (although notable exceptions exist, which we shall also discuss in this paper).

From figure 3, we observe that the EHT observations set the  $1\sigma$  upper limit  $Q \leq 0.8M$ , and the  $2\sigma$  upper limit  $Q \leq 0.95M$ . Therefore, within more than 2 standard deviations, the EHT observations rule out the possibility of Sgr A\* being an extremal RN BH (Q = M). Of course,

the NS regime ( $M < Q \le \sqrt{9/8}M$ ) is completely ruled out: therefore, the EHT observations rule out the possibility of Sgr A\* being one of the simplest possible NSs (although other NS space-times are allowed as we shall see later). Finally, re-inserting SI units, we see that the previous  $2\sigma$  upper limit translates to  $Q \le 0.95\sqrt{4\pi\epsilon_0}GM \sim 6.5 \times 10^{26}$  C. In practice, this limit is significantly weaker than the limit  $Q \le 3 \times 10^8$  C, or equivalently  $Q \le \mathcal{O}(10^{-19})$  in units of mass of Sgr A\*, obtained from astrophysical considerations in [323, 324] (see also [325–329]).

#### 3.2. Bardeen magnetically charged regular black hole

We now consider regular BH solutions, which are free of the r = 0 core singularity present within the Kerr–Newman family of metrics. Several approaches towards constructing regular BH metrics replace the Kerr–Newman core with a different type of core, such as a de Sitter or Minkowski core, or phenomenologically smearing the core over a larger surface. The Bardeen space-time is one of the first regular BH solutions to the Einstein equations ever proposed [330], and describes the space-time of a singularity-free BH with a de Sitter core [331]. The Bardeen space-time carries a magnetic charge  $q_m$ , and can be thought of as a magnetic monopole satisfying the weak energy condition [332]. It also emerges in the context of GR coupled to a very specific NLED Lagrangian [333], and can further be interpreted as a quantum-corrected Schwarzschild BH [334].

The metric function for the Bardeen magnetically charged BH is given by [330]:

$$A(r) = 1 - \frac{2Mr^2}{(r^2 + q_{\rm m}^2)^{3/2}},$$
(20)

with  $q_{\rm m}$  characterizing a specific hair and satisfying the condition  $q_{\rm m} \leq \sqrt{16/27} \approx 0.77$ . We compute the corresponding shadow radius numerically, and show its evolution as a function of the magnetic charge  $q_{\rm m}$  in figure 4. As with the RN BH, we see that increasing  $q_{\rm m}$  decreases the shadow radius. Within  $2\sigma$ , the EHT observations are consistent with Sgr A\* being a Bardeen BH for any (allowed) value of the magnetic charge, including the extremal value  $q_{\rm m} = \sqrt{16/27}M$ .

A comment, relevant for several other space-times, is in order here. Since the Bardeen regular BH can emerge within a specific NLED theory, there is potentially some ambiguity as to how one computes the corresponding shadow: one can compute it (*a*) using the 'standard' formulae (equations (12)–(14)), thus requiring only knowledge of the metric function equation (20), or (*b*) accounting for corrections to the effective geometry arising from the underlying NLED theory. Approach (*a*) can be considered model-agnostic, in the sense that the metric is taken as a phenomenological input without further questioning its origin, whereas approach (*b*) assumes a specific microscopical origin for the metric. For the Bardeen BH, we have chosen to adopt approach (*a*) for three reasons:

- Historically, Bardeen first proposed this specific metric on a phenomenological basis as a toy model, and only at a later stage was the associated NLED theory obtained;
- The corresponding underlying NLED theory is arguably quite baroque, and was 'reconstructed' to give the desired solution;
- NLED is not the only possible microscopical origin for the Bardeen BH.

Regarding the second point in particular, it is worth noting that it is almost always possible to reconstruct a possible NLED source for a given (regular or non) BH metric. The corresponding Lagrangian is not always guaranteed to be particularly motivated, and in fact more often than not will be quite baroque. While this is somewhat a matter of taste, in our view a more



**Figure 4.** Same as in figure 3 for the Bardeen magnetically charged regular BH with metric function given by equation (20), as discussed in section 3.2.

aesthetically appealing approach is a 'top-down' one starting from a specific well-motivated NLED Lagrangian, and finding the corresponding solution. For this reason, if a specific metric can be supported by a NLED source, but the latter is quite baroque or was reconstructed *a posteriori*, we choose to compute the shadow following the phenomenological/model-agnostic approach (*a*). This is the case for the Bardeen BH, which is the reason why only one curve, obtained from the standard formulae equations (12)–(14), is present on figure 4, and similar considerations hold for other metrics we study later.

#### 3.3. Hayward regular black hole

Another well-known regular BH solution has been proposed by Hayward [335]. This spacetime replaces the r = 0 GR singularity by a de Sitter core with effective cosmological constant  $\Lambda = 3\ell^2$ . In doing so, one is assuming that an effective cosmological constant plays an important role at short distances, with Hubble length  $\ell$ . Such a behavior, while introduced at a phenomenological level for the Hayward BH, has been justified within the context of the equation of state of matter at high density [331, 336], or an upper limit on density or curvature [337–339], the latter expected to emerge within a quantum theory of gravity [340].

The metric function for the Hayward regular BH is given by [335]:

$$A(r) = 1 - \frac{2Mr^2}{r^3 + 2\ell^2 M},$$
(21)

with the Hubble length associated to the effective cosmological constant satisfying  $\ell \leq \sqrt{16/27}M$  and, as originally envisaged by Hayward, characterizing an universal hair (although, since the space-time is introduced in a phenomenological way, nothing prevents the hair from being specific). We compute the corresponding shadow radius numerically, showing its evolution as a function of the Hubble length  $\ell$  in figure 5. Again, we see that increasing  $\ell$  decreases the shadow radius. Within the present constraints, we see that the EHT observations are consistent with Sgr A\* being a Hayward BH for any (allowed) value of the Hubble length associated to the effective cosmological constant, including the extremal value  $\ell = \sqrt{16/27}M$ .



**Figure 5.** Same as in figure 3 for the Hayward regular BH with metric function given by equation (21), as discussed in section 3.3.

#### 3.4. Frolov regular black hole

A well-known generalization of the Hayward BH, with the same quantum gravity-inspired theoretical motivation and featuring an additional charge parameter, has been proposed by Frolov [341]. The metric function for the Frolov BH is given by:

$$A(r) = 1 - \frac{(2Mr - q^2)r^2}{r^4 + (2Mr + q^2)\ell^2},$$
(22)

where the additional charge parameter q characterizes a specific hair and satisfies  $0 < q \le 1$ , and the length below which quantum gravity effects become important satisfies  $\ell \le \sqrt{16/27}$ as with the Hayward BH. In principle, q admits an interpretation in terms of electric charge measured by an observer situated at infinity, where the metric is asymptotically flat.

Here, we focus on the charge q, while fixing  $\ell = 0.3$ . This is similar to the choice reported in [20], which instead fixed  $\ell = 0.4$ , and can be motivated by the fact that the shadow radius changes slowly with  $\ell$ , as is clear from figure 5. Therefore, the limits on q we will report can be interpreted as constraints along a particular slice of the full parameter space (see also the earlier work of [103] by some of us). We compute the corresponding shadow radius numerically, showing its evolution as a function of the charge q in figure 6. We see that, for  $\ell = 0.3$ , the EHT observations set the upper limits  $q \leq 0.8M (1\sigma)$  and  $q \leq 0.9M (2\sigma)$ , qualitatively similar to the constraints on the electric charge of the RN metric Q we obtained in section 3.1. Recall once more that these constraints are valid for  $\ell = 0.3$ . From the results reported in section 3.3 and figure 5 for the Hayward BH, we can expect that fixing  $\ell$  to a smaller/larger value would result in weaker/tighter constraints on q, as increasing  $\ell$  leads to a slight decrease in the shadow size: in other words, we can expect a negative correlation/parameter degeneracy between  $\ell$ and q.



**Figure 6.** Same as in figure 3 for the Frolov regular BH with metric function given by equation (22), where we have fixed the Hubble length associated to the effective cosmological constant to  $\ell = 0.3$ , as discussed in section 3.4.

#### 3.5. Magnetically charged Einstein-Bronnikov regular black hole

NLED theories are well-motivated extensions of Maxwell's Lagrangian in the high-intensity regime [342], and appear in the low-energy limit of various well-known theories, including Born–Infeld electrodynamics [343], as well as several string or supersymmetric constructions [344–348]. Another important class of regular BHs arises in the context of GR coupled to Bronnikov NLED, where the Lagrangian of the latter is given by [349]:

$$\mathcal{L}(U) = F_{\mu\nu}F^{\mu\nu}\cosh^{-2}\left[a\left(F_{\alpha\beta}F^{\alpha\beta}/2\right)^{1/4}\right].$$
(23)

We recall that the relativistic invariants U and W are constructed from the electromagnetic field-strength tensor  $F_{\alpha\beta}$  and its dual  $F^{\star}_{\alpha\beta}$  as follows:

$$U = F^{\alpha\beta}F_{\alpha\beta}, \quad W = F^{\alpha\beta}F^{\star}_{\alpha\beta},$$
  
$$F^{\star}_{\alpha\beta} = \frac{1}{2}\epsilon_{\alpha\beta\mu\nu}F^{\mu\nu}, \quad F_{\mu\nu} = \partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu}, \qquad (24)$$

where  $\star$ ,  $\epsilon_{\alpha\beta\mu\nu}$ , and  $A_{\mu}$  denote respectively the Hodge dual operator, the Levi-Civita symbol, and the electromagnetic gauge field.

Einstein–Bronnikov gravity admits regular BHs only if these carry magnetic and not electric charge, i.e. choosing a purely magnetic configuration for the gauge field:

$$A_{\mu} = q_{\rm m} \cos \theta \delta^{\phi}_{\mu}, \tag{25}$$

where  $q_{\rm m}$  is the magnetic charge. In this case, the central singularity is removed if the parameter  $a = q_{\rm m}^{3/2}/M$  is different from zero. The metric function can be written explicitly as [349]:

$$A(r) = 1 - \frac{2M}{r} \left( 1 - \tanh \frac{q_{\rm m}^2}{2Mr} \right), \qquad (26)$$

where  $q_{\rm m}$  characterizes a specific hair. As anticipated earlier, however, in NLED theories photons do not move along null geodesics of the metric tensor, but of an effective geometry which depends on the details of the NLED theory. Therefore, knowledge of the metric function alone is in principle *not* sufficient to compute the shadow cast by BHs within NLED theories.

To proceed, we follow Novello's method [214, 215] to compute the effective geometry. Since these can now be considered standard references in the field, here we very succinctly review the basic steps of Novello's method, and simply refer the reader to [214, 215] for further details. Novello's method is itself based on Hadamard's method to obtain the equations for the propagation of the electromagnetic field, where the discontinuity surface  $\Sigma$  in the electromagnetic field is identified with the wavefronts. In short, one starts from the Bianchi identities involving the field-strength tensor  $F_{\mu\nu}$  and its dual. It is assumed that the field-strength is continuous across  $\Sigma$ , while its first derivative is discontinuous, with the discontinuity parametrized by  $f_{\alpha\beta}k_{\mu}$ , where  $k_{\mu}$  is the propagation vector. These conditions are then applied to the Bianchi identities and the resulting equation(s), taking the form of a cyclic identity (involving the tensor f and the propagation vector) with indices, say,  $_{\alpha\beta\mu}$ , are contracted with  $F^{\alpha\beta}k^{\mu}$ to obtain a scalar equation. These equations are evaluated on-shell and, after simple manipulations, can be expressed in the form  $g_{\mu\nu}^{\text{eff}}k^{\mu}k^{\nu} = 0$ , showing that photons move along null geodesics of  $g_{\mu\nu}^{\text{eff}}$ , itself depending on the NLED Lagrangian.

For the intermediate steps leading to the computation of the effective geometry, we refer the reader to equations (2.10)–(2.21) of [104] by some of us (see also [350]), where these steps are explicitly spelled out. We compute the shadow radius numerically, following the procedure outlined in [104]. In figure 7, we show the evolution of the shadow size as a function of the only free parameter,  $|q_m|$ . We see once more that increasing the magnetic charge  $q_m$  acts to decrease the shadow radius. We also see that the EHT observations restrict the magnetic charge to  $|q_m| \leq 0.8M (1\sigma)$  and  $|q_m| \leq M (2\sigma)$ , while noting that within Bronnikov NLED the magnetic charge is in principle allowed to take values  $q_m > M$ .

#### 3.6. Ghosh-Culetu-Simpson-Visser (GCSV) regular black hole

The GCSV BH was introduced in [351–353], including work by one of us, as toy model of a regular BH in the sense of Bardeen, where the core is asymptotically Minkowski rather than de Sitter<sup>33</sup>: stated differently, the associated energy density and pressure asymptote to zero rather than to a final value determined by the effective cosmological constant of the de Sitter case (see also [131, 354–362]). While the GCSV BH is introduced in a purely phenomenological way, Simpson and Visser (SV) argued in [353] that this space-time is mathematically interesting due to its tractability (the curvature tensors and invariants take forms which are much simpler than those of the Bardeen, Hayward, and Frolov BHs), and physically interesting because of its non-standard asymptotically Minkowski core. The metric function describing the GCSV space-time is given by [351–353]:

<sup>&</sup>lt;sup>33</sup> To the best of our knowledge, this specific metric does not carry a particular name, and here it is the first time it is being referred to as 'GCSV BH'.



**Figure 7.** Same as in figure 3 for the Einstein–Bronnikov magnetically charged BH with metric function given by equation (23), as discussed in section 3.5.

$$A(r) = 1 - \frac{2Me^{-g^2/2Mr}}{r},$$
(27)

where depending on the underlying origin for this space-time, g can characterize either an universal or specific hair. As shown by some of us in [363, 364], the space-time described by equation (27) can be obtained in the context of GR coupled to a suitable NLED source, with g associated to the NLED charge.

We compute the shadow size numerically, and show its evolution against |g| in figure 8. As for all solutions considered so far, we see that increasing the hair parameter decreases the shadow size, as also observed in the earlier work of [131] by some of us. The EHT observations set the limits  $|g| \leq 0.8M (1\sigma)$  and  $|g| \leq M (2\sigma)$ . We are not aware of any *a priori* theoretical restriction on the value of *g*, given that the metric has been introduced as a toy model.

#### 3.7. Kazakov-Solodukhin (KS) regular black hole

As a further example of regular BH, we consider the KS BH, which arises within a stringinspired model where spherically symmetric quantum fluctuations of the metric and the matter fields are governed by the 2D dilaton-gravity action [365]. The KS BH then provides a wellmotivated example of quantum deformation of a Schwarzschild BH. The metric function for this space-time is given by:

$$A(r) = -\frac{2M}{r} + \frac{\sqrt{r^2 - \ell^2}}{r},$$
(28)

where the hair parameter  $\ell > 0$  sets the scale over which quantum deformations of the Schwarzschild BH shift the central singularity to a finite radius, and characterizes an universal hair. Physically speaking, what effectively occurs is that the singularity is smeared out over



**Figure 8.** Same as in figure 3 for the Ghosh–Culetu–Simpson–Visser BH with metric function given by equation (27), as discussed in section 3.6.

a two-dimensional sphere of area  $4\pi\ell^2$ . While  $\ell$  is expected to be of the order of the Planck length  $\ell_{\text{Pl}}$ , in a phenomenological approach it can in principle take any positive value, as done for instance in [366].

We find that a closed-form expression for the shadow radius can be provided, and is given by:

$$r_{\rm sh} = \frac{\sqrt[4]{\left(9 + 3\ell^2 + 3\sqrt{9 + 2\ell^2}\right)^3}}{\sqrt{-4\sqrt{2} + 2\sqrt{9 + \ell^2 + 3\sqrt{9 + 2\ell^2}}}},$$
(29)

which correctly reduces to  $3\sqrt{3}$  in the limit  $\ell \to 0$ . The evolution of the shadow size  $r_{\rm sh}$  is shown against  $|\ell|/M$  in figure 9. The KZ BH provides the first example, among those considered, of a space-time where the size of the BH shadow increases with the hair parameter. As per our previous discussion, it is easy to show that the location of the maximum of the effective potential for the KS space-time moves to the right with respect to the Schwarzschild case for any value of  $\ell > 0$ . The EHT observations set the limits  $|\ell| \leq 0.2M (1\sigma)$  and  $|\ell| \leq M$  $(2\sigma)$ . The physical meaning of this limit is that the singularity cannot be spread out across a scale larger than the gravitational radius, and in any case has to be confined within the event horizon. This example also highlights how the EHT near-horizon image of Sgr A\* places particularly tight limits on fundamental physics scenarios which lead to a larger shadow compared to the Schwarzschild BH.

#### 3.8. Ghosh-Kumar (GK) black hole

Another interesting BH solution, which we shall refer to as the GK BH, was proposed by some of us in [367], and is described by the following metric function:

$$A(r) = 1 - \frac{2M}{\sqrt{r^2 + k^2}},$$
(30)



**Figure 9.** Same as in figure 3 for the Kazakov–Solodukhin BH with metric function given by equation (28), as discussed in section 3.7.

where depending on the underlying origin for this space-time, k can characterize either an universal or specific hair. Although the above metric function is regular throughout the whole space-time, there is nonetheless a scalar polynomial singularity at r = 0, which can be seen by computing the Kretschmann and Ricci scalars [367]. In addition, the space-time exhibits an interesting reflection symmetry  $r \rightarrow -r$ . This can be viewed as a minimal deformation of the Schwarzschild BH, and is similar to the SV space-time we will consider later in section 3.9, albeit with a crucial difference in the angular metric function, which here is the standard  $C(r) = r^2$ .

We find that a closed-form expression for the shadow radius exists and is given by:

$$r_{\rm sh} = \frac{\sqrt{\mathcal{X} - k^2 \left(1 + \frac{2}{\mathcal{X}}\right) + \frac{9}{\mathcal{X}} + 3}}{\sqrt{1 - \frac{2}{\sqrt{\mathcal{X} + \frac{9}{\mathcal{X}} - 2\frac{k^2}{\mathcal{X}} + 3}}}},$$
(31)

where the quantity  $\mathcal{X}$  is defined as:

$$\mathcal{X} \equiv \sqrt[3]{\frac{k^4 - 18k^2 + \sqrt{k^6(k^2 - 4)} + 54}{2}}.$$
(32)

It is easy to show that, with  $\mathcal{X}$  defined as in equation (32),  $r_{\rm sh}$  correctly approaches  $3\sqrt{3}$  in the limit  $k \to 0$ , where  $\mathcal{X} \to 3$ . We show the evolution of the shadow size against |k|/M in figure 10. We see that increasing |k| decreases the shadow size, as with most BH solutions studied so far. We find that the EHT observations set the upper limits on the NLED charge  $|k| \leq 1.4M$  (1 $\sigma$ ) and  $|k| \leq 1.6M$  (2 $\sigma$ ). To the best of our knowledge, these are the first ever constraints on this space-time<sup>34</sup>.

 $<sup>^{34}</sup>$  In [367], it was shown that this space-time can emerge within a specific NLED theory, where k is related to the NLED charge. Following similar considerations as for the Bardeen BH in section 3.2, here we have computed the



**Figure 10.** Same as in figure 3 for the Ghosh–Kumar BH with metric function given by equation (30), as discussed in section 3.8.

#### 3.9. Simpson-Visser regular black hole and traversable wormhole

We now start moving away from regular BH metrics, considering a metric encompassing both a regular BH as well as a traversable WH regime. A WH is a (typically tunnel-like) spacetime structure with a non-trivial topology, which can connect either two distant regions of the same Universe, or two different Universes (see e.g. [368] for a review of astrophysical WHs). In most cases, the field content needed to sustain the WH geometry requires a violation of the null energy condition [369, 370], either in the form of exotic matter or a modification to Einstein's gravity<sup>35</sup>. The simplest WH configurations feature two 'mouths' connected by a 'throat', but more complex structures are possible.

In [417], SV proposed a one-parameter extension of the Schwarzschild space-time which interpolates between the latter and a traversable WH, while passing through a regular BH geometry. The SV space-time is described by the following metric functions:

$$A(r) = 1 - \frac{2M}{\sqrt{r^2 + a^2}}, \quad C(r) = r^2 + a^2, \tag{33}$$

where the new parameter is required to satisfy a > 0, and characterizes a specific hair. For a = 0, equation (33) reduces to the Schwarzschild space-time. On the other hand, for 0 < a < 2M, the metric describes the space-time of a regular BH with a one-way space-like throat. For a = 2M, the space-time corresponds to that of a one-way WH with an extremal null throat. Finally, for a > 2M, equation (33) describes a traversable WH with a two-way time-like throat. We also note that when M = 0, equation (33) reduces to the metric of the Ellis–Bronnikov WH [418, 419]. The SV space-time and extensions thereof have received significant attention

shadow size following the phenomenological/model agnostic approach where we do not question the underlying origin of the GK BH.

 $<sup>^{35}</sup>$  See for instance [371–416] for a diverse range of examples of WH solutions within modified theories of gravity which violate the null energy condition in a controlled way.



**Figure 11.** Same as in figure 3 for the Simpson–Visser regular BH and traversable wormhole with metric functions given by equation (33), as discussed in section 3.9.

in the literature, attracting several follow-up studies, see [420–449] for examples of representative works.

As one sees from equation (33), the SV space-time is the first example of space-time we consider where  $C(r) \neq r^2$ . As a consequence, the coordinate *r* is in general *not* the areal radius, unless a = 0. This is also the key difference between the SV space-time and the GK BH we studied previously, which are otherwise very similar, and underlies several peculiar features of the SV space-time we shall now discuss. The properties of the SV space-time are easy to compute analytically. To compute the location of the photon sphere we use equation (12), which upon inserting A(r) and C(r) as given by equation (33), and working in units where M = 1, reduces to:

$$2r\left(1-\frac{2}{\sqrt{r^2+a^2}}\right) - (r^2+a^2)\frac{2r}{\sqrt{(r^2+a^2)^3}} = 2r\left(1-\frac{3}{\sqrt{r^2+a^2}}\right) = 0, \quad (34)$$

from which we trivially find the photon sphere location as being  $r_{\rm ph} = \sqrt{9 - a^2}$ , as also obtained in [417]. Note that these results are only valid for a < 3: for  $a \ge 3$ , the space-time does not possess a photon sphere (although, we stress, this does not necessarily mean that it cannot cast a shadow).

To obtain the shadow size, we use the more complete expression given by equation (14), finding that  $r_{sh} = 3\sqrt{3}$ , with no dependence on the parameter *a*. Therefore, the shadow cast by the SV space-time is in perfect agreement with the EHT observations within all four the singular BH, regular BH, one-way WH, and traversable WH regimes, as shown in figure 11. While phenomenological, this is therefore the first example (among those we have considered) of a BH mimicker space-time which could be a viable candidate in light of Sgr A\*'s shadow<sup>36</sup>.

<sup>&</sup>lt;sup>36</sup> While the SV space-time was introduced on purely phenomenological grounds in [417], it can emerge as a solution of GR coupled to a suitable NLED source and a minimally coupled phantom scalar field with an appropriate potential,

On the other hand, complementary x-ray reflection spectroscopy constraints [451] obtained by some of us set the very tight limit  $a \leq 0.5M$  at 90% C.L. [308], which is particularly tight since it is obtained from observations of the stellar-mass x-ray binary candidate BH EXO 1846-031, whose mass is 6 orders of magnitude lower than Sgr A<sup>\*</sup>.

The somewhat surprising result that the shadow cast by the SV space-time is identical in size to that of the Schwarzschild BH deserves further comments. The fact that the SV and Schwarzschild space-times are *shadow-degenerate* had already been noted in [452], and the easiest way of seeing this is to perform a coordinate transformation and express the metric described by equation (33) in terms of the areal radius  $R = \sqrt{r^2 + a^2}$ . In these new coordinates,  $g_{tt}$ ,  $g_{\theta\theta}$ , and  $g_{\phi\phi}$  are exactly of the Schwarzschild form, whereas the only deviations from the Schwarzschild form appear in  $g_{RR}$ , which however does not enter in the computation of the shadow radius as is clear from equations (10)–(15). More generally, [452] shows that the condition for a space-time to be shadow-degenerate with the Schwarzschild space-time is that the dominant photon sphere of both has the same impact parameter, after correcting for the potentially different redshift of comparable observers in the two space-times. Besides the SV space-time, [452] provides another example of a space-time which is shadow-degenerate with the Schwarzschild space-time, termed 'Class II'.

Finally, as anticipated earlier, we note that the absence of a photon sphere for a > 3 does not necessarily imply that the SV space-time cannot cast a shadow in this regime. In fact, as noted in [453], for a > 3 the WH throat can act as photon sphere. For the specific case of a = 3, two photon spheres and an anti-photon sphere at the throat degenerate into an effective marginally unstable photon sphere. This allows the SV space-time to cast a shadow. Given the significantly more complicated behavior of the system in the  $a \ge 3$  regime, here we choose not to cover it further, and refer the reader to [453] for more detailed discussions on the subject.

#### 3.10. Morris-Thorne (MT) traversable wormhole

We now consider what is arguably the simplest example of non-rotating, asymptotically flat, traversable WH, described by the well-known MT metric [454]:

$$ds^{2} = -e^{\Phi(r)}dt^{2} + \frac{dr^{2}}{1 - \Psi(r)} + r^{2}d\Omega^{2}, \qquad (35)$$

where  $\Phi(r)$  is referred to as the redshift function, and  $\Psi(r)$  as the shape function (although the reader should note that different definitions and conventions for the redshift and shape functions exist in the WH literature). One of the simplest choices for these two in principle free functions consistent with a WH geometry is to set  $\Phi = -r_0/r$  and  $\Psi(r) = (r_0/r)^{\gamma}$  [454– 458], where  $r_0$  is the radial coordinate of the WH throat, and  $\gamma$  is a constant which we shall henceforth fix to  $\gamma = 1$ . For this choice of redshift and shape functions, it is easy to show that the Arnowitt–Deser–Misner (ADM) mass [459, 460] is  $M = r_0/2$ , so the metric function can be rewritten as:

$$A(r) = e^{-\frac{2M}{r}}.$$
(36)

as shown for the first time by one of us in [450]. Following similar considerations as for the Bardeen BH in section 3.2, here we have computed the shadow size following the phenomenological/model agnostic approach where we do not question the underlying origin of the SV metric.



**Figure 12.** Same as in figure 3 for the Morris–Thorne traversable wormhole described by the metric in equation (35), whose shadow size is  $r_{sh} = eM$  and does not depend on any free parameters. Also shown are the  $1\sigma$ ,  $2\sigma$ ,  $3\sigma$ ,  $4\sigma$ , and  $5\sigma$  constraints derived from the EHT observations, given by different shades of gray. The red regions are excluded by the same observations at more than  $5\sigma$ . For comparison, the size of the shadow of a Schwarzschild BH with the same ADM mass is also shown (dashed black curve).

Importantly, the size of the WH shadow in units of the ADM mass is then uniquely determined and does not depend on additional hair parameters, given that the only free parameter in the theory ( $\gamma$ ) enters exclusively in  $g_{rr}$ , which does not affect the computation of the shadow<sup>37</sup>.

For the choice of redshift function we have adopted, the equations governing the location of the photon sphere and the corresponding shadow size simplify considerably. The equation for  $r_{\rm ph}$  takes the form  $r_{\rm ph}(1-r_{\rm ph}) = 0$ , for which the physical solution is  $r_{\rm ph} = 1$ . Then, the shadow size is given by  $r_{\rm sh} = 1/\sqrt{e^{-2}} = e$ . In other words, the size of the MT WH shadow in units of ADM mass is exactly equal to Euler's number,  $r_{\rm sh}/M = e \approx 2.718$ , i.e. almost a factor of 2 smaller than the size of the shadow of a Schwarzschild BH with the same ADM mass. By direct comparison, and as is visually clear in figure 12, the EHT measurements rule out at very high significance (>5 $\sigma$ ) the possibility of Sgr A\* being a MT traversable WH, for the particular well-motivated choice of redshift function  $\Phi = -r_0/r$ . Finally, we note that this result has been also been discussed by the EHT collaboration in [20].<sup>38</sup>

#### 3.11. Damour-Solodukhin (DS) wormhole

Another interesting WH solution was proposed by DS [461] (see also [462–466] for further relevant work), with the goal of studying the extent to which globally static WHs can mimic

<sup>&</sup>lt;sup>37</sup> Note that in equation (36) the radial coordinate is only defined down to the WH throat, and therefore the limit  $r \to 0$  (where formally  $A(r) \to 0$ ) should not be taken. The location of the WH throat is controlled by the shape function, which does not in itself affect the computation of the shadow, if not implicitly through the parameter  $\gamma$  which controls the ADM mass.

<sup>&</sup>lt;sup>38</sup> As per earlier discussions, one has to ensure that the size of the WH shadow is larger than that of the WH throat. For our choice of  $\gamma = 1$ , it is easy to show that the throat is located at  $r_0 = 2M$ . Since the shadow size is  $r_{sh} = eM > 2M$ , the shadow size we have computed is indeed astrophysically relevant.

observational features of BHs, including accretion of matter, no-hair properties, ringing of quasi-normal modes, and dissipative properties associated to the event horizon. The DS WH is described by the following metric [461]:

$$ds^{2} = -\left(1 - \frac{2M}{r} + \lambda^{2}\right)dt^{2} + \frac{dr^{2}}{1 - \frac{2M}{r}} + r^{2}d\Omega^{2},$$
(37)

and differs from the Schwarzschild BH because of the presence of the dimensionless parameter  $\lambda$ , which characterizes an universal hair. When  $\lambda \neq 0$ , equation (37), the would-be horizon at r = 2M actually corresponds to a throat joining two isometric regions. Note that the would-be Killing horizon where  $g_{tt} = 0$  would actually be located at  $r = 2M/(1 + \lambda^2) < 2M$ , so behind the throat. As discussed in [461], this space-time provides an example of Lorentzian WH<sup>39</sup>, where the throat describes a brane-like structure (a high-tension distribution), with radial and tangential tensions proportional to  $1/\lambda^2$  but with zero energy density, localized in a thin shell around the throat. We also note that a similar space-time was studied in 3D in [471], with the goal of restoring Poincaré recurrence in BHs: in this specific case,  $\lambda \sim e^{-4\pi GM^2}$  is exponentially small, so as to mimic the expected dependence of the Poincaré recurrence time on the BH entropy. In [461], it was argued that choosing exponentially small values of  $\lambda$  allows the resulting WH space-time to mimic both classical and quantum properties of Schwarzschild BHs.

In the following, we treat  $\lambda$  as a free parameter and simply ask what values are compatible with the observed shadow of Sgr A\*, assuming that the latter is indeed described by a DS WH. Note that as soon as  $\lambda \neq 0$  the space-time is no longer asymptotically flat, so in order to compute its shadow we adopt the more general computation given by equations (16) and (17). Following this procedure, we can easily obtain a closed-form expression for the shadow radius as follows:

$$r_{\rm sh} \simeq \frac{3\sqrt{3}}{1+\lambda^2}.\tag{38}$$

where the  $\simeq$  sign is included as we have made the approximation  $r_0 \gg 1$  (in units M = 1), valid for Sgr A<sup>\*</sup>. We plot the evolution of the shadow size against  $\lambda$  in figure 13, where we see that increasing  $\lambda$  decreases the size of the WH shadow. In particular, we find that the EHT observations set the upper limits  $\lambda \lesssim 0.4 (1\sigma)$  and  $\lambda \lesssim 0.5 (2\sigma)$ . To the best of our knowledge, these are the first robust observational constraints on  $\lambda$ . We note of course that our limits are consistent with the exponentially small values of  $\lambda$  suggested in [461, 471], for which the impact on the shadow size is negligible as we could have expected.

#### 3.12. Janis-Newman-Winicour (JNW) naked singularity

Having discussed a few interesting WH solutions, we now turn our attention to another fascinating class of BH mimickers: NSs (see [472] for a review). The cosmic censorship conjecture, taken at face value, appears to forbid the existence of NSs [2]. In addition, the end result of gravitational collapse is commonly believe to be a BH, with the central singularity (if any) surrounded by an event horizon, a result which seemingly enjoys support from Birkhoff's theorem as well [473]. The first gravitational collapse solution of a spherically symmetric and homogeneous dust cloud (the so-called OSD collapse) was derived in 1939 by Oppenheimer,

<sup>&</sup>lt;sup>39</sup> That this space-time describes a WH and not a BH can explicitly be seen adopting the Kodama–Hayward invariant formalism (see e.g. [309, 467–470]), and verifying whether the Kodama energy at the would-be horizon is 0 (BH) or >0 (WH).



**Figure 13.** Same as in figure 3 for the Damour–Solodukhin wormhole with metric function given by equation (37), as discussed in section 3.11.

Snyder, and Datt [474, 475], and does indeed recover the Schwarzschild BH as final state. Within the OSD collapse scenario, trapped surfaces will form around the center before the formation of the central space-like singularity, which is therefore causally disconnected from all other points of space-time.

However, the OSD collapse model (as well as subsequent refinements thereof) is highly idealized in its assuming homogeneous density and zero pressure within the massive collapsing star: this is not expected to be the case for realistic astrophysical stars. Under physically more realistic situations, the introduction of inhomogeneities in the matter density profile and a non-zero pressure *within GR* leads to the possibility of dynamical collapse resulting in a strong but non-space-like curvature singularity, not hidden behind an event horizon [476–485]. Therefore, given suitable initial conditions, continuous gravitational collapse of an inhomogeneous matter cloud can lead to the formation of a class of exotic horizonless compact objects known as NSs, which have received significant attention, particularly in the recent literature (see e.g. [486–505]).

One of the simplest known NS solutions is the JNW NS. This is a solution of the Einstein– Klein–Gordon equations for a massless scalar field [506] (see also [507–510]). Despite being a NS, it possesses a photon sphere and therefore casts a shadow. The JNW space-time is characterized by a scalar charge which one can continuously adjust to recover the Schwarzschild solution. Given its importance, this space-time has been the subject of a large number of studies, particularly regarding the possibility of observationally distinguishing it and other NSs from the Schwarzschild BH, see e.g. [102, 485, 511–519].

The JNW space-time can be characterized by the following metric functions [506]:

$$A(r) = \left[1 - \frac{2M}{r(1 - \overline{\nu})}\right]^{1 - \overline{\nu}}, \quad C(r) = r^2 \left[1 - \frac{2M}{r(1 - \overline{\nu})}\right]^{\overline{\nu}}.$$
 (39)



**Figure 14.** Same as in figure 3 for the JNW naked singularity with metric functions given by equation (39), as discussed in section 3.12.

Here,  $\overline{\nu} = 1 - \sqrt{1 + q/M}$ , q is the scalar charge of the NS, and the scalar field radial profile is given by the following:

$$\phi(r) = \frac{q}{2\sqrt{M^2 + q^2}} \left(1 - \frac{2\sqrt{M^2 + q^2}}{r}\right).$$
(40)

Overall, the parameter  $\overline{\nu}$  characterizes a specific hair. Note that, since  $C(r) \neq r^2$ , the coordinate r appearing in equation (39) is not the areal radius, which can instead be obtained through an appropriate transcendental coordinate transformation. We also note that the JNW space-time can only cast a shadow for  $0 \leq \overline{\nu} \leq 0.5$ .

We compute the shadow radius numerically, and plot its evolution against the hair parameter  $\overline{\nu}$  in figure 14. There is a wide range of parameter space where Sgr A\*'s shadow is consistent with the corresponding compact object being a JNW NS. In fact, the EHT observations set rather weak limits on  $\overline{\nu}$ , with  $\overline{\nu} \lesssim 0.4M$  (1 $\sigma$ ) and  $\overline{\nu} \lesssim 0.45M$  (2 $\sigma$ ), barely excluding the extremal JNW NS,  $\overline{\nu} = 0.5$ . These limits can of course be translated into rather weak limits on the strength of the scalar field from equation (40). More importantly, among all the examples of BH mimickers we have considered so far, the JNW NS is one of the few which is in excellent agreement with the EHT observations given the metric tests we have performed in this work.

#### 3.13. Joshi-Malafarina-Narayan (JMN) naked singularity

Another well-known NS solution is the JMN NS, developed by one of us in [520]. To distinguish it from another NS solution developed by the same authors in [521] we shall refer to the solution considered here as JMN singularity of the first type, or JMN-1 for short, as common practice in the literature. The JMN-1 NS can be formed as the end state of gravitational collapse given zero radial pressure but non-zero tangential pressure, i.e. in the presence of an anisotropic fluid. As shown in [520], the non-zero tangential pressure can prevent trapped surfaces from forming around the high-density core region of the collapsing matter cloud, resulting in the formation of a central NS over the course of a large co-moving time (see also [522, 523]).

The JMN-1 space-time contains a compact region filled with an anisotropic fluid for  $r < R_b$ , which is then usually matched to an exterior space-time. The interior JMN-1 space-time is described by the following metric:

$$ds^{2} = (\chi - 1) \left(\frac{r}{R_{b}}\right)^{\frac{\chi}{(1-\chi)}} dt^{2} + \frac{dr^{2}}{(1-\chi)} + r^{2} d\Omega^{2}, \qquad (41)$$

whose associated ADM mass is  $M = \chi R_b/2$ , with  $\chi$  characterizing specific hair, and whose energy density  $\rho$ , radial pressure  $p_r$ , and tangential pressure  $p_{\theta}$  are given by:

$$\rho = \frac{\chi}{r^2}, \quad p_r = 0, \quad p_\theta = \frac{\chi}{4(1-\chi)}\rho,$$
(42)

satisfying all the energy conditions. It can be shown that the interior JMN-1 space-time can be matched smoothly at  $r = R_b$  to an exterior Schwarzschild space-time with the same mass M, since the induced metrics and extrinsic curvatures of the two space-times can be matched on the matching hypersurface (see [524]).

The shadow cast by the JMN-1 space-time was studied in a number of works by some of us, including [78, 485, 525]. It can be shown that the JMN-1 space-time does not support a photon sphere and cannot cast a shadow for  $0 < \chi < 2/3$  (which can be re-written in the somewhat more illuminating form  $R_b > 3M$ ). On the other hand, for  $2/3 \le \chi < 1$ , the JMN-1 space-time casts a shadow of size  $r_{\rm sh} = 3\sqrt{3}\chi R_b/2 = 3\sqrt{3}M$ , i.e. identical to that of a Schwarzschild BH with the same mass M. This is shown in figure 15. The physical reason for this is that 'shadow photons' are actually traveling in the exterior Schwarzschild metric, since the interior metric does not support a photon sphere, and therefore the shadow size is identical to that of a Schwarzschild BH, thus trivially ensuring consistency with the findings of [452]. The photon sphere associated to the exterior Schwarzschild metric  $(r_{\rm ph} = 3M)$  is astrophysically relevant only if it is located outside of the matching radius ( $R_b$ ): when setting  $M = \chi R_b/2$ , the condition  $3M \ge R_b$  becomes  $\chi \ge 2/3$ , recovering the previously discussed condition for the JMN-1 space-time to cast a shadow. Summarizing, based on the metric test considered in this work, the JMN-1 NS is completely indistinguishable from a Schwarzschild BH, and is in perfect agreement with the EHT observations, which therefore allow for the possibility of Sgr A\* being a NS.

In closing, we note that one of us studied the effect of spherical accretion upon a JMN-1 NS, finding that the corresponding images remain indistinguishable from those of a Schwarzschild BH, even when considering the accretion flow spectra at  $\approx 200 \text{ GHz}$  [485]. However, a subsequent study by some of us in [525] showed that the two can in principle be distinguished by the intensity distribution of light inside the shadow region: although extremely challenging, this opens up the intriguing possibility of discriminating a JMN-1 NS from a Schwarzschild BH with the same mass, which is otherwise not possible with the test considered in this work. Finally, we also note that in the  $0 < \chi < 2/3$  regime, for which no photon sphere exists, the central singularity can cast a peculiar full-moon-type image [485] which, although inconsistent with the EHT image of Sgr A<sup>\*</sup>, is certainly an interesting signature to look for in upcoming SMBH images.



**Figure 15.** Same as in figure 3 for the JMN-1 naked singularity with interior metric given by equation (41) and smoothly matched to an exterior Schwarzschild metric at  $r = R_b$ , as discussed in section 3.13.

#### 3.14. Naked singularity surrounded by a thin shell of matter

In all the space-times we have discussed so far, the existence of a shadow (if any) was tied to the existence of a photon sphere, which one could argue to be responsible for casting the shadow itself. However, as shown in a number of recent works, the existence of a photon sphere is a sufficient but not necessary condition for a space-time to cast a shadow. One particularly interesting example in this context is that where the existence of a thin-shell of matter is responsible for casting a shadow. Here, we consider a NS space-time surrounded by a thin shell of matter, studied by some of us in [526]. Specifically, we glue two JMN-1 NS space-times (see section 3.13), an interior one (JMN-1<sub>i</sub>) characterized by the parameter  $\chi_i$  and an exterior one (JMN-1<sub>e</sub>) characterized by the parameter  $\chi_e$ , at  $r = R_{b1}$ . The JMN-1<sub>e</sub> space-time is then further glued to an external Schwarzschild space-time with mass  $M = \chi_e R_{b2}/2$  at  $r = R_{b2}$ . Explicitly, the metric of the JMN-1<sub>i</sub> space-time is given by:

$$ds^{2} = (\chi_{e} - 1) \left(\frac{R_{b1}}{R_{b2}}\right)^{\frac{\chi_{e}}{(1 - \chi_{e})}} \left(\frac{r}{R_{b1}}\right)^{\frac{\chi_{i}}{(1 - \chi_{i})}} dt^{2} + \frac{dr^{2}}{(1 - \chi_{i})} + r^{2} d\Omega^{2},$$
(43)

whereas the metric of the JMN-1<sub>e</sub> space-time is given by:

$$ds^{2} = (\chi_{e} - 1) \left(\frac{r}{R_{b2}}\right)^{\frac{\Lambda_{e}}{(1 - \chi_{e})}} dt^{2} + \frac{dr^{2}}{(1 - \chi_{e})} + r^{2} d\Omega^{2}.$$
 (44)

As shown by some of us in [526] using Israel's junction conditions [527], the gluing procedure is possible even if  $\chi_i \neq \chi_e$ , provided a thin shell of matter exists at the time-like hypersurface  $r = R_{b1}$ . Note that  $\chi_i$  and  $\chi_e$  characterize specific hair.

If  $\chi_e < 2/3$ , based on our earlier discussions in section 3.13 it is clear that the space-time does not support the existence of a photon sphere in the usual sense. Nonetheless, if  $\chi_i > 2/3$ , the space-time can still cast a shadow, thanks to the presence of a cusp-like feature in the effective potential which leads to the existence of a critical impact parameter (much like in the



**Figure 16.** Same as in figure 3 for the naked singularity surrounded by a thin shell of matter with metric described by equations (43) and (44), as discussed in section 3.14.

case where a photon sphere is present), as shown in [526]. One can then show that the shadow size is given by:

$$r_{\rm sh} = \frac{R_{b1}}{\sqrt{1 - \chi_{\rm e}} \left(\frac{R_{b1}}{R_{b2}}\right)^{\frac{\chi_{\rm e}}{2(1 - \chi_{\rm e})}}},\tag{45}$$

which depends on the location of the thin shell of matter ( $R_{b1}$ ), the radius where the JMN-1<sub>e</sub> space-time is matched to the external Schwarzschild space-time ( $R_{b2}$ ), and the value of  $\chi$  corresponding to the JMN-1<sub>e</sub> space-time ( $\chi_e$ ), recalling that  $M = \chi_e R_{b2}/2$ , and subject to the conditions  $R_{b2} > R_{b1}$  and  $\chi_e < 2/3$  (or equivalently  $R_{b2} > 3M$ ), with the latter required for the space-time to not possess a photon sphere, and therefore for the shadow to be cast exclusively by the thin shell of matter.

For illustrative purposes, we consider an example where we fix  $\chi_e = 0.6$  and  $R_{b2} \approx 0.33$ (implying that  $\chi_e R_{b2} \approx 2$  for consistency, as we are working in units of *M*), so that the shadow size only depends on the physically most interesting parameter  $R_{b1}$ , i.e. the location of the thin shell of matter. We show the evolution of the shadow size against  $R_{b1}$  in figure 16. We immediately note that the space-time can cast a shadow only for non-zero values of  $R_{b1}$ 1.5*M*. This is consistent with our expectations: in fact, the limit  $R_{b1} \rightarrow 0$  corresponds to the case where the JMN-1<sub>i</sub> space-time, and correspondingly the thin-shell of matter, is not present. This case reduces to that of a single JMN-1 space-time already studied in section 3.13, for which we know that no shadow is cast for  $\chi < 2/3$  (see figure 15), which is clearly the case here since  $\chi_e = 0.6$ . This result confirms that it is genuinely the thin shell of matter which is responsible for the space-time casting a shadow, even in the absence of a photon sphere. Of course, it is clear from equation (45) that the shadow size depends on  $\chi_e$  and  $R_{b2}$  separately, so that different choices of  $\chi_e$  and  $R_{b2}$  such that  $\chi_e R_{b2} = 2$  would lead to results which differ from those we have shown here. Nonetheless, the qualitative features we have shown and discussed would remain unchanged: this is sufficient for the purposes of the present discussion with the aim of showing that a thin shell of matter can cast a shadow.

#### 3.15. Null naked singularity

In the absence of a photon sphere, a thin shell of matter is not the only possibility for casting a shadow. A relevant example in this sense was studied by some of us in [74], in the context of a novel NS space-time described by the following metric:

$$ds^{2} = -\frac{dt^{2}}{\left(1 + \frac{M}{r}\right)^{2}} + \left(1 + \frac{M}{r}\right)^{2} dr^{2} + +r^{2} d\Omega^{2}.$$
(46)

The above metric describes a space-time of ADM mass M which approaches the Schwarzschild space-time as  $r \rightarrow \infty$ , possesses a strong curvature singularity at r = 0 (as can be verified by computing the Kretschmann and Ricci scalars), and is manifestly horizon-less. Note that this space-time is not characterized by additional hairs. Since the singularity is null-like, we shall refer to the space-time described by equation (46) as 'null NS' (NNS). We note that this particular NNS space-time was introduced on purely phenomenological grounds, as a toy model to explore the possibility of a shadow being cast in the absence of a photon sphere and a thin shell of matter. We leave open the question of whether gravitational collapse from suitable initial conditions can lead to the NNS space-time as end product. We note, however, that the formation of null- and even time-like singularities from the gravitational collapse of matter clouds is known to be possible and has been studied extensively in the literature (see e.g. [478, 522, 528–535] for relevant works).

From equation (13), it is clear that the NNS does not possess a photon sphere—formally, the solution to equation (13) would be  $r_{ph} = 0$ . In other words, the minimum turning point radius for a photon fired from infinity is r = 0, where the effective potential for null geodesics has a finite value. Nonetheless, the impact parameter corresponding to the minimum turning point radius is non-zero, and this allows the NNS to cast a shadow of size  $r_{sh} = M$ . In other words, despite the absence of an event horizon, null geodesics reaching the NNS from infinity with an impact parameter b < M will be trapped closer and closer to the central singularity, and will not be able to return to the observer: this leads to the appearance of a central dark region in the observer's sky, which we associate to the NNS shadow. The situation is perhaps visually easier to grasp from figure 1(f) of [74], which shows examples of light trajectories around the NNS, making the physical origin of the shadow clear.

The shadow cast by the NNS is of course too small to be consistent with the EHT observations of Sgr A<sup>\*</sup>, as is clear from figure 17. Therefore, much as the MT traversable WH, the EHT observations rule out the possibility of Sgr A<sup>\*</sup> being a NNS at very high ( $>5\sigma$ ) significance. Of course, as we have discussed earlier in sections 3.12–3.14, other NS space-times remain potentially viable descriptions of Sgr A<sup>\*</sup>, leaving open the intriguing possibility that the latter might not be a BH but an exotic horizonless object.

#### 3.16. Minimally coupled scalar field with a potential

So far we have discussed a few well-motivated regular BH solutions, as well as BH mimickers such as WHs and NSs. To continue our journey into horizon-scale tests of alternatives to the Schwarzschild metric, we now move on to BH solutions which violate the NHT in a controlled way, in the presence of additional fields or exotic interactions. As is well known, one of the most straightforward (and well-motivated) ways of violating the NHT is through the presence of additional matter fields, such as a scalar field. The following subsections will therefore be devoted to the study of a number of 'hairy' BH solutions arising broadly speaking in scenarios



**Figure 17.** Same as in figure 12 for the null naked singularity with metric given by equation (46), as discussed in section 3.15.

of 'new physics', which include but are not limited to the presence of additional fields, modified theories of gravity (these two scenarios are not mutually exclusive), or new high-energy physics models.

We begin by studying the simplest possible scenario where the NHT can be violated in a controlled way, considering a minimally coupled real scalar field  $\phi$  featuring a potential. The dynamics of the field are thus governed by the following Lagrangian:

$$\mathcal{L} \supset \frac{1}{2} \partial^{\mu} \phi \partial_{\mu} \phi - V(\phi) , \qquad (47)$$

where here and in what follows the symbol  $\supset$  indicates that the Lagrangian contains at least the terms on the right, plus potentially other terms which are irrelevant for the determination of the corresponding BH metric. Following [128, 536, 537], including work by some of us, we take the following form for the potential:

$$V(\phi) = \frac{1}{2\nu^4} e^{-2\sqrt{2}\phi} g(\phi),$$
(48)

where  $g(\phi)$  is given explicitly in [128]. The specific choice of potential is not particularly motivated from a high-energy theory perspective. However, in the context of BH physics, it is particularly interesting because it provides a toy example of a controlled violation of the NHT, which proceeds by explicitly introducing a new energy scale in the scalar sector. More specifically, the scalar field profile is given by [128, 536, 537]:

$$\phi(r) = \frac{1}{\sqrt{2}} \ln\left(1 + \frac{\nu}{r}\right) \tag{49}$$

where the scale  $\nu > 0$  governs the scalar field fall-off, and effectively acts as a hair parameter. The metric functions read (see e.g. [128, 536, 537]):

$$A(r) = -\frac{2r\nu + 6M(2r+\nu)}{\nu^2} - \frac{2r(r+\nu)(6M+\nu)}{\nu^3} \ln \frac{r}{r+\nu},$$
  

$$C(r) = r(r+\nu).$$
(50)


Figure 18. Same as in figure 3 for the BH with scalar hair, as discussed in section 3.16.

Although it is not straightforward to see from equation (50), the metric function A(r) does indeed reduce to the Schwarzschild limit 1 - 2M/r as the hair parameter vanishes. This is best seen by Taylor-expanding the metric function in  $\nu$ , yielding:

$$A(r) = 1 - \frac{2M}{r} + \frac{3M - r}{3r^2}\nu + \mathcal{O}(\nu^2), \qquad (51)$$

from which one easily sees that the limit  $\nu \to 0$  indeed recovers the Schwarzschild metric, and  $\nu$  characterizes an universal hair. We compute the shadow radius of BHs within this theory numerically, as done earlier in [128] by one of us. We show the evolution of the shadow size against the hair parameter  $\nu$  in figure 18, finding that the shadow size increases quickly with increasing hair parameter. As in other similar examples we have already seen, the EHT observations set particularly tight constraints on this scenario, with limits on the hair parameter of order  $\nu \leq 0.01M (1\sigma)$  and  $\nu \leq 0.4M (2\sigma)$ . This effectively constrains the energy scale at which violations of the NHT become important, within this particular theory.

## 3.17. Black holes and wormholes with conformal scalar hair

We now study another scenario featuring a scalar field, which leads to a controlled violation of the NHT. We consider a real scalar field  $\phi$ , conformally coupled to gravity, through the following Lagrangian:

$$\mathcal{L} \supset \partial^{\mu}\phi\partial_{\mu}\phi + \frac{1}{6}\phi^{2}R, \qquad (52)$$

where *R* is the Ricci scalar, and the 1/6 factor ensures that the coupling is conformal (in 4 dimensions). BH solutions within this theory have been obtained by Astorino [538] (see also [539–546] for BH and black brane solutions within related models), and contain a hair parameter *S* related to the scalar field profile, and directly connected to the value of the scalar field at the horizon. The hair parameter, or scalar charge, emerges as an integration constant and

characterizes primary hair, i.e. it cannot be expressed as a function of the other hair parameters. The metric function reads:

$$A(r) = 1 - \frac{2M}{r} + \frac{S}{r^2},$$
(53)

where the scalar charge *S* characterizes a specific hair and can take both positive and negative values, with  $S < M^2$  required in the former regime for the solution to describe a BH. The space-time described by equation (53) is closely related to the well-known 'BBMB black hole' [547, 548], and appears very similar to the RN metric [equation (18)]. However, there is a crucial difference in that the scalar hair parameter can take negative values, which are instead effectively precluded within the RN metric as  $Q^2 > 0$ . The S < 0 regime is in fact better known as *mutated* RN BH, and effectively describes the geometry of a traversable WH [549] (see also [128]). Formally, this corresponds to an analytic continuation of the RN metric for imaginary values of the electric charge (thus negative values of  $Q^2$ ), although we discourage the reader from taking this interpretation too far. We also note that the RN-like form of the metric is no coincidence, and is directly related to the conformal symmetry enjoyed by the Lagrangian in equation (52), see in addition [550] for earlier relevant results. Finally, it is worth noticing that a similar metric arises in the context of a conformal massive gravity model where Lorentz invariance is spontaneously broken (see [301, 551, 552]), although only for a very specific choice of the function  $\mathcal{F}$  appearing in equation (10) of [551].

Since the metric is RN-like, the shadow size takes the same functional form as equation (19):

$$r_{\rm sh} = \frac{\sqrt{2} \left(3 + \sqrt{9 - 8S/M^2}\right)}{\sqrt{4 + \frac{\sqrt{9 - 8S/M^2} - 3}{S/M^2}}},$$
(54)

valid within both the BH (S > 0) and WH (S < 0) regimes. We show the evolution of the shadow size against the scalar charge *S* in figure 19. We see that within the regime S > 0 the shadow size decreases with increasing scalar charge, analogously to the case with the electric charge in the RN metric. In this regime, the EHT observations set the limits  $S \leq 0.65M^2$  ( $1\sigma$ ) and  $S \leq 0.9M^2$  ( $2\sigma$ ). In the mutated RN BH/WH regime S < 0, the shadow size instead increases when increasing the absolute value of the scalar charge, quickly bringing the shadow size outside the range allowed by the EHT observations, which set the lower limits  $S \gtrsim -0.04M^2$  ( $1\sigma$ ) and  $S \gtrsim -0.4M^2$  ( $2\sigma$ ).

## 3.18. Clifton-Barrow f(R) gravity

Generalizations and extensions of Einstein's GR have a long and rich history, but undoubtedly one of the most studied theories in this sense is f(R) gravity, where the Ricci scalar R in the Einstein–Hilbert action is replaced by an arbitrary function f(R). Several f(R) models have been constructed in the literature with the aim of describing the late-time accelerated expansion of the Universe (and/or an early accelerated phase of inflation), which requires satisfying viability conditions including the positivity of effective gravitational couplings, stability of cosmological perturbations, asymptotic  $\Lambda$ CDM behavior in the high curvature regime, stability of the late-time de Sitter point, as well as observational constraints from the equivalence principle, Solar System, and cosmological tests. For an inevitably incomplete list of important



**Figure 19.** Same as in figure 3 for solution with conformally coupled scalar hair, leading to a BH (S > 0, solid curve) or a wormhole (S < 0, dashed curve) depending on the value of the hair parameter *S*, and with metric function given by equation (53), as discussed in section 3.17.

works on a wide range of aspects of f(R) gravity, with particular reference to BH solutions therein, we refer the reader to [553–608].

Clifton and Barrow considered a particularly interesting f(R) model in [609], whose action is given by:

$$S = \int \mathrm{d}^4 x \sqrt{-g} M_{\rm Pl}^{2-2\delta} R^{1+\delta} \,, \tag{55}$$

where  $\delta$  is a dimensionless parameter,  $M_{\text{Pl}}$  is the Planck mass, and GR is recovered in the limit  $\delta \rightarrow 0$ . This theory was argued to possess several appealing properties, including its being a one-parameter extension of GR not introducing any additional scale besides the already present Planck mass, its providing a testing ground for new developments in particle production, holography and gravitational thermodynamics, as well as its being one of the few f(R) models possessing simple exact solutions for Friedmann-Lemaître-Robertson-Walker (FLRW)-like cosmological models and exact static spherically symmetric solutions generalizing the Schwarzschild metric. Clifton and Barrow [609] does indeed identify such a solution, whose metric function is given by:

$$A(r) = r^{2\delta \frac{1+2\delta}{1-\delta}} - \frac{2M}{r^{\frac{1-4\delta}{1-\delta}}},$$
(56)

where  $\delta$  characterizes an universal hair. It is easy to see that the above metric is asymptotically flat only if  $-1/2 < \delta < 0$  or  $\delta > 1$ . We exclude the regime  $\delta > 1$ , as we find *a posteriori* that within this range the shadow size is by far too small and completely excluded by observations, and therefore focus on the region  $-1/2 < \delta < 0$ .



**Figure 20.** Same as in figure 3 for the BH in Clifton–Barrow f(R) gravity with metric function given by equation (56), as discussed in section 3.18.

We compute the shadow size numerically, and show its evolution against  $\delta$  in figure 20.<sup>40</sup> We see that the shadow size decreases with increasing  $|\delta|$  (i.e. with  $\delta$  becoming more negative), and that the EHT observations set the limits  $\delta \ge -0.15$  (1 $\sigma$ ) and  $\delta \ge -0.20$  (2 $\sigma$ ). In the  $\delta < 0$  regime, the existence of a stable matter-dominated period of evolution, which is itself a prerequisite for a viable cosmological model, requires  $\delta < -1/4$  [609]. This makes our limits particularly interesting, as they exclude this region, and therefore would appear to exclude this class of models, highlighting a very appealing complementarity between BH shadows and cosmological constraints.

### 3.19. Horndeski gravity

We now move on to a well-known and much more general class of scalar-tensor theories: Horndeski gravity. On very general grounds, scalar-tensor theories are an appealing playground for theorists as they encapsulate the key characteristics of several theories of modified gravity, which typically possess a scalar-tensor limit. Back in 1974, Horndeski wrote down the most general scalar-tensor theory with equations of motion at most of second order [610], with the latter requirement sufficient to ensure that the theory is free of the Ostrogradsky ghost [611].<sup>41</sup> After living almost ignored for over 30 years, the theory, which encompasses many well-known theories of gravity (including f(R) gravity) as specific cases, was 'rediscovered' in [622, 623], making it a beautiful example of 'sleeping beauty' theory which is now attracting a tremendous amount of attention (see e.g. [624–638]) after decades of dormancy.

<sup>&</sup>lt;sup>40</sup> We have found that a closed-form expression for the shadow radius exists, but is too cumbersome to report here. We have nonetheless verified that our numerical results perfectly reproduce those provided by this closed-form expression. <sup>41</sup> Note that having second order equations of motion is a sufficient but not necessary condition for the theory to be physically viable. As an example, the beyond Horndeski (GLPV) class of scalar-tensor theories has equations of motion of higher than second order, yet does not propagate unphysical degrees of freedom thanks to a degeneracy condition which allows to maintain the canonical number of degrees of freedom [612, 613] (see for instance [614–621] for further relevant works).

Here, we shall focus on the shift-symmetric subclass of Horndeski gravity, where the scalar field  $\phi$  enjoys a shift symmetry  $\phi \rightarrow \phi + \text{const.}$  This sector is of particular interest since a well-known no-go theorem by Hui and Nicolis dictates that static, asymptotically flat BH solutions thereof are isometric to the Schwarzschild BH, with a trivial (constant) profile for the scalar field [639]. Of course, this no-go theorem comes with several assumptions, which we shall here break (see e.g. [272, 640–644]) in order to play with hairy BH solutions. The shift-symmetric sector of Horndeski gravity we shall consider is described by the following action:

$$S = \int \mathrm{d}^4 x \sqrt{-g} \left( \mathcal{L}_2 + \mathcal{L}_3 + \mathcal{L}_4 + \mathcal{L}_5 \right), \qquad (57)$$

where the four Lagrangian terms  $\mathcal{L}_2$ ,  $\mathcal{L}_3$ ,  $\mathcal{L}_4$ , and  $\mathcal{L}_5$  are given by the following [610]:

$$\mathcal{L}_2 = G_2(X) \,, \tag{58}$$

$$\mathcal{L}_3 = -G_3(X) \Box \phi \,, \tag{59}$$

$$\mathcal{L}_{4} = G_{4}(X)R + G_{4,X}\left[(\Box\phi)^{2} - (\nabla_{\mu}\nabla_{\nu}\phi)^{2}\right],$$
(60)

$$\mathcal{L}_{5} = G_{5}(X)G_{\mu\nu}\nabla^{\mu}\nabla^{\nu}\phi - \frac{1}{6}G_{5,X}\left[(\Box\phi)^{3} - 3\Box\phi(\nabla_{\mu}\nabla_{\nu}\phi)^{2} + 2(\nabla_{\mu}\nabla_{\nu}\phi)^{3}\right],$$
(61)

which involve four arbitrary functions  $(G_2, G_3, G_4, \text{ and } G_5)$  of the kinetic term  $X \equiv -\partial^{\mu}\phi\partial_{\mu}\phi/2$ ,  $(\nabla_{\mu}\nabla_{\nu}\phi)^2 \equiv \nabla_{\mu}\nabla_{\nu}\phi\nabla^{\nu}\nabla^{\mu}\phi$ , and  $(\nabla_{\mu}\nabla_{\nu}\phi)^3 \equiv \nabla_{\mu}\nabla_{\nu}\phi\nabla^{\nu}\nabla^{\rho}\phi\nabla_{\rho}\nabla^{\mu}\phi$ , and where  $G_{4,X}$  and  $G_{5,X}$  are respectively the derivatives of the functions  $G_4$  and  $G_5$  with respect to *X*.

As discussed in [644], the no-go theorem of [639] essentially rests upon three assumptions:

- The derivatives with respect to *X* of the *G<sub>i</sub>*s only contain positive or zero powers of *X* in the limits *X*→0, *r*→∞;
- 2. The kinetic term *X* is canonical;
- The norm of the Noether current associated to the shift symmetry is finite down to the horizon.

Clearly, then, in order to find hairy BH solutions within Horndeski gravity, one must at the very least break one of the above assumptions—the solutions which we shall consider are obtained by breaking the first one. As shown in [644], doing so is possible at least for the following particular choices of the  $G_i$  functions:

$$G_2 \supseteq \sqrt{-X}, \quad G_3 \supseteq \ln |X|,$$
  

$$G_4 \supseteq \sqrt{-X}, \quad G_5 \supseteq \ln |X|.$$
(62)

A choice of the  $G_i$  functions satisfying equation (62) has the potential to admit non-Schwarzschild-isometric, asymptotically flat BH solutions, with a non-trivial scalar field profile, and with a regular Noether current. In the following, we shall consider two specific solutions obtained in [644, 645] for a particular choice of the  $G_i$  functions satisfying equations (62), which we shall overall refer to as 'sqrt quartic Horndeski gravity<sup>42</sup>', while simply referring to the solutions of [644] and [645] as 'Case 1' and 'Case 2' respectively.

<sup>&</sup>lt;sup>42</sup> We are obviously using 'sqrt' as a shorthand for 'square root'.

3.19.1. Case 1. The first BH solution in the shift-symmetric sector of Horndeski gravity we consider was studied in [644]. In the notation of equations (58)–(61), and within the system of units we have adopted, the particular choice of  $G_i$  functions is given by:

$$G_2 = \eta X, \quad G_4 = 1/16\pi + \beta \sqrt{-X}, \quad G_3 = G_5 = 0.$$
 (63)

Clearly,  $G_2$  corresponds to a canonical kinetic term, whereas it is the particular X-dependence of  $G_4$  (the  $1/16\pi$  term simply leads to an Einstein–Hilbert term in the action, since it is multiplied by R) which allows for hairy BH solutions, by satisfying the specific set of conditions laid out in equation (62).

Hairy BH solutions within this theory are described by the following metric function (again within the system of units we have adopted) [644]:

$$A(r) = 1 - \frac{2M}{r} - \frac{8\pi p_{\rm eff}}{r^2},$$
(64)

which is clearly asymptotically flat, and where we have defined the hair parameter  $p_{\text{eff}} \equiv \beta^2 / \eta$ which characterizes an universal hair. Note that, depending on the sign of  $\eta$  (which, in principle, can be either positive or negative),  $p_{\text{eff}}$  can carry either sign: in particular, positive  $\eta$ corresponds to positive  $p_{\text{eff}}$ , and vice-versa<sup>43</sup>.

Note that this BH solution is of the RN-like form. In [644], this was argued to be possibly related to the remnant of global conformal invariance enjoyed by the action defined by equation (63), as the extremal RN space-time itself is known to possess a discrete conformal isometry related to spatial inversion<sup>44</sup>. Within the negative  $p_{\text{eff}}$  regime,  $p_{\text{eff}} > -M^2/8\pi$  is required (else the solution describes a NS), whereas no such restrictions are present in the positive  $p_{\text{eff}}$  regime. Finally, we note that the positive  $p_{\text{eff}}$  regime is expected to described a mutated RN WH, as the S < 0 regime of the BH/WH with conformal scalar hair considered in section 3.17 (although we note that such an interpretation was not considered in [644]).

By virtue of the RN-like form of the metric, we can immediately borrow upon the previous results for RN-like space-times (sections 3.1 and 3.17) to obtain the shadow radius as a function of  $p_{\text{eff}}$  (see e.g. equation (54)):

$$r_{\rm sh} = \frac{\sqrt{2} \left(3 + \sqrt{9 + 64\pi p_{\rm eff}}\right)}{\sqrt{4 + \frac{3 - \sqrt{9 + 64\pi p_{\rm eff}}}{8\pi p_{\rm eff}}}},$$
(65)

which, we recall, is valid within both the  $p_{\rm eff} < 0$  and  $p_{\rm eff} > 0$  regimes. We show the evolution of the shadow size against the scalar charge  $p_{\rm eff} = \beta^2/\eta$  in figure 21, which of course is qualitatively identical to figure 19, albeit with the *x* axis inverted and rescaled. We see that the EHT observations set very stringent constraints on  $p_{\rm eff}$ . In the  $p_{\rm eff} > 0$  regime, we find the limits  $p_{\rm eff} \lesssim 0.002M^2$  (1 $\sigma$ ) and  $p_{\rm eff} \lesssim 0.01M^2$  (2 $\sigma$ ). On the other hand, in the  $p_{\rm eff} < 0$  regime,

<sup>&</sup>lt;sup>43</sup> As explicitly discussed in [644], although the kinetic term is phantom for  $\eta < 0$ , this is in itself not a guarantee of the instability of the theory, as the latter depends sensibly on the quartic term and hence on the value of  $\beta$ , see e.g. [646, 647]. To the best of our knowledge, an analysis of the stability of this specific Horndeski theory has not been carried out in the literature, and doing so would go well beyond the scope of our work. Nonetheless, we note that the BH solution is equally valid for the  $\eta > 0$  and  $\eta < 0$  regimes, provided  $p_{\text{eff}}$  satisfies  $p_{\text{eff}} > -M^2/8\pi$ , as noted in [644] (albeit with a different notation compared to us), and as we later impose.

<sup>&</sup>lt;sup>44</sup> This symmetry is best known as Couch–Torrence inversion symmetry, and maps  $(t,r) \rightarrow (t,M+M^2/(r-M))$  [648] (see also [649, 650]). Under this symmetry, the metric of an extremal RN BH transforms as  $g_{\mu\nu} \rightarrow \omega^2 g_{\mu\nu}$ , with  $\omega = M/(r-M)$ , making this a (discrete) conformal symmetry. Moreover, it can be shown that the symmetry interchanges the future event horizon  $\mathcal{H}^+$  with future null infinity  $\mathscr{I}^+$ , and similarly the past event horizon  $\mathcal{H}^-$  with past null infinity  $\mathscr{I}^-$ , allowing for an interpretation as spatial inversion symmetry.



**Figure 21.** Same as in figure 3 for the BH in the Horndeski gravity sector described by equation (63) and with metric function given by equation (64), as discussed in section 3.19.1.

we find the limits  $p_{\text{eff}} \gtrsim -0.005M^2 (1\sigma)$  and  $p_{\text{eff}} \gtrsim -0.01M^2 (2\sigma)$ . These constraints limit the combination  $\beta^2/\eta$ , with  $\beta$  governing the strength of the  $\sqrt{-X}$  term in  $G_4$  and  $\eta$  governing the strength of the kinetic term in  $G_2$ . A detailed study of the theoretical and model-building implications of these constraints as well as the complementarity with cosmological, astrophysical, and GW constraints (which is the subject of a rich and rapidly growing body of literature, see e.g. [651–680] for examples of these constraints and observational signatures of Horndeski gravity) would be very interesting, but goes significantly the scope of the paper, and is therefore left to future work.

3.19.2. Case 2. We now consider a different BH solution in the shift-symmetric sector of Horndeski gravity, studied in [645], and whose shadow was first computed in [154] by some of us. Using the same notation of equations (58)–(61), the authors begin from the following particular choice of  $G_i$  functions [645]:

$$G_{2} = \alpha_{21} X + \alpha_{22} (-X)^{\omega_{2}}, \quad G_{3} = -\alpha_{31} (-X)^{\omega_{3}}, G_{4} = 1/16\pi + \alpha_{42} (-X)^{\omega_{4}}, \quad G_{5} = 0,$$
(66)

which clearly generalizes equation (63), with the latter recovered for  $\alpha_{22} = \alpha_{31} = 0$  and  $\omega_4 = 1/2$ . Note also that the theory described by equation (66) is known to be weakly hyperbolic for  $\alpha_{21} \neq 0$  [681], but its stability has not been examined for  $\alpha_{21} = 0$ . The authors of [645] at a later stage focus on a particular choice of the  $\alpha$  parameters, which still preserves the regularity of the Noether current associated to the shift symmetry, and the finiteness of the energy. More specifically, the following choice of  $G_i$  functions is studied [645]:

$$G_2 = \alpha_{22} (-X)^{\frac{3}{2}}, \quad G_4 = 1/16\pi + \alpha_{42}\sqrt{-X},$$
  

$$G_3 = G_5 = 0.$$
(67)

The above is clearly not a generalization of equation (63): while the  $G_4$  functions are identical, there is a crucial difference in the  $G_2$  functions. Therefore, we should not expect the BH solution derived within this sector to (necessarily) reduce to the one studied in section 3.19.1.

The corresponding BH solution is indeed qualitatively very different from the one considered in section 3.19.1, and is described by the following metric function:

$$A(r) = 1 - \frac{2M}{r} + \frac{2Mh}{r} \ln\left(\frac{r}{2M}\right),\tag{68}$$

where the hair parameter *h* characterizes an universal hair and is related to the fundamental Lagrangian parameters  $\alpha_{22}$  and  $\alpha_{42}$  appearing in equation (67) through the following:

$$h \equiv \frac{2\sqrt{2}}{3\sqrt{3}} \alpha_{42} \sqrt{-\frac{\alpha_{42}}{\alpha_{22}}}.$$
 (69)

We see that the above space-time is asymptotically flat, while computing the Kretschmann and Ricci scalars signals the presence of a scalar polynomial singularity at r = 0. In the h > 0regime, which for consistency requires  $\alpha_{22} < 0$  and  $\alpha_{42} > 0$ , the space-time only contains one event horizon, analogously to the Schwarzschild metric. In the h < 0 regime (and more precisely for  $h \in [-2;0]$ ), which for consistency requires  $\alpha_{22} > 0$  and  $\alpha_{42} < 0$ , the space-time contains two horizons: an outer event horizon, and an inner Cauchy horizon.

Unlike the earlier [154] by some of us, where the analysis was restricted to the h < 0 regime, here we do not set any prior restriction on the sign of the hair parameter h. While slightly cumbersome, a closed-form expression for the shadow radius exists and is given by:

$$r_{\rm sh} = \frac{3h\mathcal{X}}{\sqrt{1 + \frac{2(-1+h\ln\mathcal{X})}{3h\mathcal{X}}}},\tag{70}$$

where the quantity  $\mathcal{X}$  is defined as:

$$\mathcal{X} \equiv W \left[ \frac{2e^{\frac{1}{3} + \frac{1}{h}}}{3h} \right],\tag{71}$$

with W(z) denoting the principal branch of the Lambert W function, also known as omega function or product logarithm function. In figure 22 we show the evolution of the shadow size against h. We see that the shadow radius quickly increases with increasing |h| in the h < 0regime, while decreasing more slowly with increasing h in the h > 0 regime. As a result, negative values of the hair parameter are very tightly constrained: we find that consistency with the EHT observations requires  $-0.01 \leq h \leq 0.40$  ( $1\sigma$ ) and  $-0.15 \leq h \leq 0.60$  ( $2\sigma$ ). As with the BH in Horndeski gravity studied in section 3.19.1, these constraints limit a particular combination of two fundamental Lagrangian parameters (in this case  $\alpha_{42}\sqrt{-\alpha_{42}/\alpha_{22}}$ ), and a detailed study of the theoretical and model-building implications of these constraints, as well as the complementarity with cosmological and astrophysical constraints, while interesting, is left to future work.

## 3.20. MOdified Gravity (scalar-tensor-vector gravity)

A particularly interesting modified theory of gravity was put forward by Moffat [682], and typically goes by the name of scalar-tensor-vector gravity, but is also often referred to as MOG (short for MOdified Gravity), as we shall do here. MOG is a covariant scalar-tensor-vector theory of gravity which introduces an additional massive scalar field and an additional massive vector field, while allowing the coupling strength of the latter, as well as the gravitational constant, to vary in space and time. The theory effectively adds a repulsive Yukawa force



**Figure 22.** Same as in figure 3 for the BH in the Horndeski gravity sector described by equation (67) and with metric function given by equation (68), as discussed in section 3.19.2.

to Newton's acceleration law. For the full action of MOG, we refer the reader to [682]. We also note that MOG has a long history of successfully explaining several astrophysical and cosmological observations: for an inevitably incomplete list of important works on MOG, see for example [683–707].

A particular BH solution within MOG was derived in [708], assuming that the gravitational coupling *G* is constant (although not necessarily taking its Newtonian value), i.e.  $\partial_{\mu}G = 0$  (see also [709–711]). In this configuration, the metric function of the resulting BH solution is given by [708]:

$$A(r) = 1 - \frac{2M(1+\alpha)}{r} + \frac{M^2\alpha(1+\alpha)}{r^2},$$
(72)

where  $\alpha$  is a parameter controlling the strength of the effective gravitational coupling  $G = G_N(1 + \alpha)$ , and characterizes an universal hair. We immediately notice that the space-time is of the RN form, with effective charge proportional to the BH mass. Clearly, one of the effects of MOG is modify the BH ADM mass as  $M \to M(1 + \alpha)$ , as already noticed in many works [690, 697, 708]. In the following, however, we shall opt for the same strategy adopted by many earlier works, and report the shadow size in units of the 'bare'/source/Newtonian mass M rather than the ADM mass. The rationale is that this is the ADM mass quantifies the energy measured by an observer at infinity, but dynamical measurements of the BH mass sufficiently close to the horizon will actually probe a quantity which is much closer to the Newtonian mass. This is indeed the case for the S-stars which underlie our mass-to-distance ratio prior, as explicitly discussed in [697]. In addition, one can envisage constraining  $\alpha$  through independent tests of the strength of Newton's constant, allowing to disentangle the Newtonian mass from the ADM mass.

Having made these considerations (see section 3.38.1 for further considerations of this sort in the context of another model), we compute the shadow size in units of M as done in various



**Figure 23.** Same as in figure 3 for the MOG BH with metric function given by equation (72), as discussed in section 3.20.

earlier works (we refer especially to [697]), and show its evolution against the MOG parameter  $\alpha$  in figure 23. Moreover, we note that a closed-form expression for the shadow radius can be found, and is given by the following:

$$r_{\rm sh} = \sqrt{2\alpha} \frac{3 + 3\alpha + \sqrt{(1+\alpha)(9+\alpha)}}{\sqrt{-3 + \alpha + \sqrt{(1+\alpha)(9+\alpha)}}},$$
(73)

which, although not obvious at first glance, does indeed reduce to  $3\sqrt{3}$  as  $\alpha \rightarrow 0$ , as can be seen by Taylor expanding equation (73) in  $\alpha$ . We see that the effect of increasing  $\alpha$  is that of very quickly increasing the shadow size. Despite the RN-like form of the space-time, this is a result of our choice of reporting the shadow size in units of *M* rather than  $M(1 + \alpha)$ , for the reasons discussed above.

We note that the EHT observations set extremely tight upper limits on  $\alpha$ , of the order of  $\alpha \lesssim 0.01 (1\sigma)$  and  $\alpha \lesssim 0.1 (2\sigma)$ . These limits are much tighter than the limit  $\alpha \lesssim 0.662$  obtained from the motion of the S2-star in [705], and safely exclude the range of values of  $\alpha$  required for MOG to explain the rotation curves of dwarf galaxies,  $2.5 \lesssim \alpha \lesssim 11.2$  [683]. On the other hand, they are competitive with the upper limits of  $\alpha \lesssim 0.1$  arising from neutron stars [693], and  $\alpha \lesssim 0.03$  coming from globular clusters [686]. Overall, the limits we have provided on the MOG parameter of order  $\alpha \lesssim 0.01$  are therefore among the tightest in the literature.

#### 3.21. Brane-world black holes

We now consider one of the most well-motivated high-energy physics models: the Randall– Sundrum II (RSII) brane-world model, which has been tested over a wide range of scales and using an enormous variety of probes (see e.g. [712–726]). This is an AdS<sub>5</sub> brane-world model where the extra dimension has an infinite size and negative bulk cosmological constant [727]. When projecting the full 5-dimensional BH solution of the theory onto our 4D space-time, BHs effectively inherit a tidal charge q which carries projected information about 5D bulk (and in particular the AdS<sub>5</sub> curvature radius  $\ell$ ), while characterizing the effect of the bulk on the brane. More specifically, the tidal charge controls the strength of projected Weyl tensor  $\mathcal{E}_{\mu\nu}$  which transmits tidal charge stresses from the bulk to the brane, and takes the form [728]:

$$\mathcal{E}_{\mu\nu} = -\left(\frac{q}{\widetilde{M}_{\rm Pl}^2}\right) \frac{1}{r^4} \left[u_{\mu}u_{\nu} - 2r_{\mu}r_{\nu} + h_{\mu\nu}\right],\tag{74}$$

where  $M_{\text{Pl}}$  is the 5-dimensional Planck mass,  $u_{\mu}$  is the velocity 4-vector,  $r_{\mu}$  is an unit radial vector, and  $h_{\mu\nu} = g_{\mu\nu} + u_{\mu}u_{\nu}$ . Physically speaking, a negative tidal charge amplifies gravitational effects of the bulk on the brane.

BH solutions within the RSII model have been studied in a large number of works (see e.g. [589, 729–738], as well as [739, 740]). The metric function for spherically symmetric solutions takes the following form:

$$A(r) = 1 - \frac{2M}{r} + \frac{4M^2q}{r^2},$$
(75)

where q characterizes a specific hair. We note that the metric function is again RN-like as with a few other metrics discussed earlier. Therefore, we can immediately borrow from the previous results, and write the shadow radius as a function of the tidal charge in a closed form:

$$r_{\rm sh} = \frac{\sqrt{2} \left(3 + \sqrt{9 - 32q}\right)}{\sqrt{4 + \frac{\sqrt{9 - 32q} - 3}{4q}}} \,. \tag{76}$$

In figure 24 we plot the size of the shadow radius as a function of the tidal charge: the plot is of course analogous to figures 19 and 21. It is clear that the EHT observations place very tight limits on the tidal charge,  $q \leq 0.15$  (1 $\sigma$ ) and  $q \leq 0.2$  (2 $\sigma$ ). The same observations also place very tight limits on the negative tidal charge regime,  $q \geq -0.01$  (1 $\sigma$ ) and  $q \geq -0.1$  (2 $\sigma$ ). This is in contrast with the shadow of M87<sup>\*</sup>, which instead exhibited a very slight preference for negative tidal charge, as noted earlier in [101, 124]. This is due to the fact that M87<sup>\*</sup>'s shadow size is slightly larger than what would be expected given its mass-to-distance ratio, or conversely one could argue that its mass-to-distance ratio is biased low. At any rate, this fact again highlights how Sgr A<sup>\*</sup>'s shadow places particularly tight constraints on scenarios which tend to enlarge the shadow radius compared to the Schwarzschild BH. We note that the negative tidal charge regime of the RSII BH would in principle appear to admit an interpretation in terms of mutated RN BH/WH (just as the space-time discussed in section 3.17), though to the best of our knowledge we are not aware of this possibility having being discussed in the literature on brane-world BHs.

#### 3.22. Magnetically charged Einstein-Euler-Heisenberg black hole

Earlier in section 3.5 we discussed regular magnetically charged BHs arising from Einstein– Bronnikov NLED. Another well-motivated NLED theory is Einstein–Euler–Heisenberg NLED [741], which appears in the low-energy limit of Born–Infeld electrodynamics [343], and is described by the following Lagrangian:

$$\mathcal{L}(U,W) = -\frac{1}{4}U + \frac{\mu}{4}\left(U^2 + \frac{7}{4}W^2\right).$$
(77)



**Figure 24.** Same as in figure 3 for the Randall–Sundrum II brane-world BH with metric function given by equation (75), as discussed in section 3.21.

Choosing the same gauge field configuration as in equation (25), and carefully accounting for the NLED-induced effective geometry, BH shadows within the Einstein–Euler–Heisenberg model were studied by some of us in [104], whose procedure we follow. For completeness, we report the metric function for magnetically charged BH solutions within this theory, given by:

$$A(r) = 1 - \frac{2M}{r} + \frac{q_{\rm m}^2}{r^2} - \frac{2\mu}{5} \frac{q_{\rm m}^4}{r^6}, \qquad (78)$$

where  $q_{\rm m}$  is the BH magnetic charge and characterizes a specific hair, and  $\mu$  is the NLED coupling appearing in equation (77). We note that as long as  $\mu \neq 0$ , values of the magnetic charge  $q_{\rm m} > 1$  are allowed.

We compute the shadow radius numerically, following [104] by some of us and once more carefully taking into account the effective geometry induced by the NLED coupling by adopting Novello's method [214, 215] (again, as in section 3.5), for the intermediate steps of this calculation we refer the reader to [104]. Note that we now have 2 free parameters:  $q_m$  and  $\mu$ . However, for simplicity, we fix the Lagrangian coupling to  $\mu = 0.3$ . The reason is two-fold: (*a*) as noted in [104], most of the effect on the shadow size come from the magnetic charge  $q_m$  rather than the NLED coupling  $\mu$ , and (*b*)  $\mu \sim 0.3$  is approximately the largest allowed coupling before the perturbative approach of the theory around the Maxwell Lagrangian for  $\mu \rightarrow 0$  ceases to be meaningful.

In figure 25 we plot the evolution of the shadow radius against  $q_m$ , at fixed  $\mu = 0.3$ . We immediately note that the behavior of the shadow radius, and therefore the constraints from the EHT observations, depend on whether we are in the  $q_m < 1$  or  $q_m > 1$  regime. For  $q_m < 1$ , we find the upper limits  $q_m \leq 0.7M (1\sigma)$  and  $q_m \leq 0.8M (2\sigma)$ . On the other hand, for  $q_m > 1$ , we observe a sharp discontinuity in the shadow radius, with the shadow size apparently diverging. This feature is due to the appearance of a singularity in the NLED-induced effective geometry: for more details see [104].

Leaving aside the issue of whether the  $q_m > 1$  regime is physical in first place, given this singularity, in this regime we find that the shadow size varies drastically as  $q_m$  changes. For



**Figure 25.** Same as in figure 3 for the Einstein–Euler–Heisenberg magnetically charged BH with metric function given by equation (77), where we have fixed the NLED coupling to  $\mu = 0.3$ , as discussed in section 3.22.

 $\mu = 0.3$ , consistency with the EHT observations requires  $q_{\rm m} \sim 1.25M$ . This number depends to some extent on the value to which  $\mu$  is fixed (see [104] for further discussions), although for values of  $\mu \lesssim 0.3$ , following the results of [104], we can expect consistency with the EHT observations for values of the magnetic charge  $1.1M \lesssim q_{\rm m} \lesssim 1.5M$ . We stress once again, however, that it is unclear whether we can consider this range of parameter space physical.

# 3.23. Sen black hole

The Sen BH is a non-regular space-time appearing in the low-energy limit of heterotic string theory [742], whose field content includes the electromagnetic field  $A^{\mu}$  with field-strength tensor  $F^{\mu\nu}$ , the second-rank antisymmetric tensor gauge field  $B_{\mu\nu}$  (also known as Kalb–Ramond field [743]), and the dilaton field  $\Phi$ ,<sup>45</sup> as described by the following action in the string frame [744] (see also [745–748] for other examples of BH and black *p*-brane solutions in the low-energy limit of string theory):

$$S = \int \mathrm{d}^4 x \sqrt{-g} \, e^{-\phi} \, \mathcal{L} \,, \tag{79}$$

where the Lagrangian  $\mathcal{L}$  is given by:

$$\mathcal{L} = R - \frac{1}{8} F^{\mu\nu} F_{\mu\nu} + \partial^{\mu} \Phi \partial_{\mu} \Phi - \frac{1}{12} H_{\kappa\lambda\mu} H^{\kappa\lambda\mu} , \qquad (80)$$

with  $H_{\kappa\lambda\mu}$  the third-rank tensor field (Kalb–Ramond 3-form) defined in terms of the tensor gauge field and the field-strength tensor of the electromagnetic field as:

<sup>&</sup>lt;sup>45</sup> In heterotic string theory the dilaton, the Kalb–Ramond field, and the metric tensor appear as a set of massless excitations of a closed string.



**Figure 26.** Same as in figure 3 for the Sen BH with metric function given by equation (82), as discussed in section 3.23.

$$H_{\kappa\mu\nu} \equiv \partial_{\kappa}B_{\mu\nu} + \partial_{\nu}B_{\kappa\mu} + \partial_{\mu}B_{\nu\kappa} - \frac{1}{4}\left(A_{\kappa}F_{\mu\nu} + A_{\nu}F_{\kappa\mu} + A_{\mu}F_{\nu\kappa}\right).$$
(81)

The metric function derived in this configuration was found by Sen [744] and reads:

$$A(r) = 1 - \frac{2M}{r + \frac{q_{\rm m}^2}{M}},$$
(82)

where  $q_{\rm m}$  is an effective charge (including both the electric charge and an effective charge associated with the dilaton field) measured by a static observer at infinity, which characterizes specific hair, and is bound by the requirement  $q_{\rm m} \leq \sqrt{2}$ . Various studies have been devoted to the Sen BH and its rotating generalization (the Kerr–Sen BH) in the literature, see for instance [749–758].

A closed-form expression exists for the shadow radius, and is given as follows:

$$r_{\rm sh} = \frac{3 - 2q_{\rm m}^2 + \sqrt{9 - 4q_{\rm m}^2}}{2\sqrt{\frac{\sqrt{9 - 4q_{\rm m}^2 + q_{\rm m}^2 - 3}}{q_{\rm m}^2}}}.$$
(83)

Although not obvious, equation (83) reduces to  $3\sqrt{3}$  as  $q_m \rightarrow 0$ , as can be seen by Taylor expanding it in  $q_m$ . We plot the evolution of the shadow radius against  $q_m$  in figure 26, from which we see that the shadow radius quickly decreases with increasing charge. We also see that the EHT observations set the upper limits  $q_m \leq 0.6M$  ( $1\sigma$ ) and  $q_m \leq 0.75M$  ( $2\sigma$ ), ruling out the possibility of Sgr A\* being an extremal Sen BH with  $q_m = \sqrt{2}M$ . We further note that complementary x-ray reflection spectroscopy constraints obtained by one of us set a much tighter limit,  $q_m \leq 0.6M$  at 90% C.L. [759], where as in section 3.9 *M* denotes the mass of EXO 1846-031.

### 3.24. Kalb-Ramond black hole

Remaining in string-inspired scenarios, another interesting BH solution has been derived in a scenario where the Kalb–Ramond field  $B_{\mu\nu}$  plays a key role in driving spontaneous breaking of Lorentz symmetry [743]. Specifically, the scenario considered by [760] envisages a field content including the Kalb–Ramond field  $B_{\mu\nu}$  and the Kalb–Ramond 3-form  $H_{\kappa\mu\nu}$ .<sup>46</sup> Inspired by the Standard Model Extension (an effective field theory containing all possible Lorentz breaking operators in addition to the Standard Model and GR Lagrangian terms [761– 763]) and by heterotic string theory [760], considered a scenario where the Kalb–Ramond field acquires a non-vanishing vacuum expectation value (VEV)  $b_{\mu\nu}$  (with  $b^2 = b_{\mu\nu}b^{\mu\nu}$ ), as well as a non-minimal coupling to the Ricci tensor with strength  $\xi_2$  (for the full action of the theory, see [760]). The non-zero VEV spontaneously breaks Lorentz invariance, leading to a host of interesting observational signatures [764–766].

BHs within this configuration have been studied by [760], and found to be described by the following metric function:

$$A(r) = 1 - \frac{2M}{r} + \frac{\Gamma}{r^{\frac{2}{k}}},$$
(84)

where  $\Gamma$  is an integration constant, and k is related to the Kalb–Ramond VEV and non-minimal coupling strength through  $k = |b|^2 \xi_2$  and therefore characterizes an universal hair. For illustrative purposes, here we choose to focus our attention on the parameter k which is of more direct interest (since it enjoys a direct physical interpretation) and is expected to be small. To this end, we fix  $\Gamma = 0.5$ : although there is no deep motivation for this value, this allows us to also compare our results to the earlier work of [108] by some of us.

We compute the size of the Kalb–Ramond BH numerically, and show its evolution against k in figure 27. As already found earlier in [108], we see that increasing k decreases the size of the BH shadow (see also [767]. For this particular choice of  $\Gamma$ , we find that the EHT observations set the upper limits  $k \leq 1.2$  (1 $\sigma$ ) and  $k \leq 1.5$  (2 $\sigma$ ). If independent limits on the Kalb–Ramond VEV are available, these limits can be translated to limits on the Kalb–Ramond non-minimal coupling strength (and vice-versa). However, we stress that these results only correspond to a slice of the full  $\Gamma$ –k parameter space, and that smaller (larger) values of  $\Gamma$  would lead to weaker (tighter) constraints on k (for more details see [108]).

In closing, it is worth noting that the space-time described by equation (84) is closely related to the 4D string-corrected BH solution found by one of us in [768], which generalizes the Callan–Myers–Perry BH [769] through a dimensional regularization-like procedure akin to the one considered in 4D Einstein–Gauss–Bonnet gravity, which we will discuss later in section 3.27. The solution derived in [768] corresponds to equation (84) with k = 1/2 (thus leading to a  $1/r^4$  correction to the Schwarzschild BH), and requires  $\Gamma \leq 27/16M^4 \sim 1.69M^4$ . However, it is easy to show (as we have explicitly checked) that within this particular spacetime, and given the above upper limit on  $\Gamma$ , the EHT observations are not able to set any meaningful limit on  $\Gamma$ : thus, we will not discuss this space-time further<sup>47</sup>.

<sup>&</sup>lt;sup>46</sup> Since the electromagnetic field does not appear within the field content considered by [760], the contributions related to  $A_{\mu}$  in equation (81) vanish.

<sup>&</sup>lt;sup>47</sup> Note that for this particular space-time we cannot translate the Reissner–Nordström constraints on Q to constraints on  $\Gamma$ , as the latter space-time corresponds to k = 1 (and  $\Gamma = Q^2$ ), whereas here we have k = 1/2.



**Figure 27.** Same as in figure 3 for the Kalb–Ramond BH with metric function given by equation (84), where we have fixed  $\Gamma = 0.5$ , as discussed in section 3.24.

## 3.25. Einstein-Maxwell-dilaton gravity (EMDG)

Still remaining in the string-inspired realm, one of the models in this context which enjoys particularly strong theory motivation is EMDG, which arises as an effective theory of the heterotic string at low energies once the additional six dimensions of the ten-dimensional manifold are compactified onto a torus [744], see also section 3.23. In EMDG, a dynamical dilaton field  $\Phi$ evolving under a potential  $V(\Phi)$  is included in the GR Lagrangian as:

$$\mathcal{L} \supset \frac{1}{16\pi} \left( \mathcal{R} + \frac{1}{2} (\nabla^{\mu} \Phi)^2 + V(\Phi) \right) + \frac{\alpha'}{8} \Phi F^{\mu\nu} F_{\mu\nu} , \qquad (85)$$

where  $\alpha'$  regulates the coupling of the dilaton to photons. Given a BH of dilaton charge Q, the corresponding metric function of the so-called Einstein–Maxwell-dilaton-1 BH is given by [745, 746, 770]:

$$A(r) = 1 - \frac{2M}{r} \left( \sqrt{1 + \frac{q^4}{4M^2r^2}} - \frac{q^2}{2Mr} \right),$$
(86)

with the requirement that the parameter  $q = Q\sqrt{\alpha'/8}$  associated to the dilaton charge, which characterizes a specific hair, satisfies  $q \leq \sqrt{2}$ .

A closed-form expression exists for the shadow radius, and is given as follows:

$$r_{\rm sh} = \frac{\sqrt{6\left(\tilde{q}+6\right) - q^2\left(q^2 + \sqrt{q^4 - 20q^2 + 36} + 16\right)}}{2\sqrt{2 - \frac{16}{2q^2 + \sqrt{2}\sqrt{q^4 - (\tilde{q}+16)q^2 + 6(\tilde{q}+6)}}}},$$
(87)

where the quantity  $\tilde{q}$  is defined as:

$$\tilde{q} \equiv \sqrt{q^4 - 20q^2 + 36}.$$
(88)



**Figure 28.** Same as in figure 3 for the Einstein–Maxwell-dilaton 1 BH with metric function given by equation (86), as discussed in section 3.25.

Although not obvious, equation (87) reduces to  $3\sqrt{3}$  as  $q \to 0$ , as can be seen by Taylor expanding it in q. We show the evolution of the shadow radius as a function of the charge parameter q in figure 28. We see that the EHT observations set the upper limits  $q \leq 0.8M (1\sigma)$  and  $q \leq M (2\sigma)$ , ruling out the extremal Einstein–Maxwell-dilaton-1 BH ( $q = \sqrt{2}M$ ).

## 3.26. Einstein-æther gravity

Einstein–æther gravity is a theory of gravity that dynamically violates Lorentz symmetry by means of a unit norm time-like vector field (the 'æther'), which defines a preferred time-like direction at each point of space-time. The Einstein–æther gravity Lagrangian is governed by 4 free parameters  $c_i$  with i = 1, 2, 3, 4, and is given by the following [771–773] (see also [774–783] for other important works):

$$\mathcal{L} \supset -h^{\alpha\beta}_{\mu\nu} \left( \nabla_{\alpha} u^{\mu} \right) \left( \nabla_{\beta} u^{\nu} \right) + \lambda \left( u^{2} + 1 \right), \tag{89}$$

where  $u^{\mu}$  is the æther field,  $\lambda$  is a Lagrange multiplier enforcing the condition  $u^{\mu}u_{\mu} = -1$ , and the tensor  $h^{\alpha\beta}_{\mu\nu}$  is given by:

$$h^{\alpha\beta}_{\mu\nu} = c_1 g^{\alpha\beta} g_{\mu\nu} + c_2 \delta^{\alpha}_{\mu} \delta^{\beta}_{\nu} + c_3 \delta^{\alpha}_{\nu} \delta^{\beta}_{\mu} - c_4 u^{\alpha} u^{\beta} g_{\mu\nu} \,. \tag{90}$$

In what follows, we will adopt the widely used shorthand notation  $c_{ij} = c_i + c_j$ ,  $c_{ijk} = c_i + c_j + c_k$ .

Two exact static spherically symmetric BH solutions within Einstein–æther gravity are known [784, 785]. The first solution, commonly referred to as Einstein–æther type 1, holds for  $c_{14} = 0$  and  $c_{123} \neq 0$ . The metric function for this space-time is given by:

$$A(r) = 1 - \frac{2M}{r} - \frac{27c_{13}M^4}{16(1 - c_{13})r^4},$$
(91)

with the requirement  $0 < c_{13} < 1$ , where  $c_{13}$  obviously characterizes an universal hair. We compute the shadow radius numerically, plotting it as a function of the Lagrangian parameter  $c_{13}$  in figure 29. We see yet again an example of shadow radius increasing with increasing



**Figure 29.** Same as in figure 3 for the Einstein–æther-1 BH with metric function given by equation (91), as discussed in section 3.26.

fundamental parameter, which leads to this scenario being tightly constrained by the ETH observations, which set the upper limits  $c_{13} \leq 0.1$  (1 $\sigma$ ) and  $c_{13} \leq 0.75$  (2 $\sigma$ )—note that the  $c_i$ s are dimensionless. While this is an independent constraint on  $c_{13}$  obtained in the strong-field regime, we note that it is still significantly weaker than constraints arising from GWs. For instance, the multi-messenger GW event GW170817 sets the very stringent limit  $|c_{13}| \leq 10^{-15}$  [786].

The second exact BH solution in this theory holds for  $c_{14} \neq 0$  and  $c_{123} = 0$ , and is commonly referred to as Einstein–æther type 2. The metric function for this space-time is given by:

$$A(r) = 1 - \frac{(2 - c_{14})M}{r} - \frac{(2 - c_{13} - c_{14})(2 - c_{14})^2 M^2}{8(1 - c_{13})r^2},$$
(92)

with the requirement  $0 \le c_{14} \le 2c_{13} < 2$ , where  $c_{13}$  and  $c_{14}$  obviously characterize universal hairs. In this case, the space-time is described by 2 free parameters. For illustrative purposes, we fix  $c_{13} = 0.99$ , close to the maximum value allowed, which as argued earlier is completely excluded in the case of the Einstein–æther type 1 BH. The purpose is simply to illustrate how a non-zero value of  $c_{14}$  can in principle help bring a previously excluded range of  $c_{13}$  values into agreement with the EHT observations. We compute the shadow radius numerically (see also [787]), plotting it as a function of the Lagrangian parameter  $c_{13}$  in figure 30. In this case we see that in principle a value of  $1.3 \le c_{14} \le 1.4$  is allowed by the EHT observations, when fixing  $c_{13} = 0.99$ . However, we note that these values of  $c_{14}$  are excluded by GW observations, with the GW170817 event setting the very stringent limit  $c_{14} \le 2.5 \times 10^{-5}$  [786].

In closing, we discuss important relations between Einstein–æther gravity and other wellknown theories of gravity, to which our results could therefore carry over. We first note that Einstein-æther gravity is closely related to Hořava–Lifshitz (HL) gravity [788, 789], a wellknown and well-studied candidate theory of quantum gravity which achieves power-counting



**Figure 30.** Same as in figure 3 for the Einstein–æther-2 BH with metric function given by equation (92), where we have fixed the Lagrangian parameter  $c_{13}-c_{13} = 0$ , as discussed in section 3.26.

renormalizability by adding higher order spatial derivatives but not time derivatives, thus taming the UV behavior of the graviton propagator, at the cost of explicitly breaking diffeomorphism invariance<sup>48</sup>. Einstein–æther gravity is known to appear in the IR limit of HL gravity (when operators higher than second order are neglected), provided one requires the æther field to be hypersurface orthogonal at the level of the action [800]. In fact, the Einstein-æther BH solutions we have studied are also solutions of the IR limit of HL gravity, as discussed in [785] (see also [801] for other BH solutions in HL gravity). Another well-known modified theory of gravity appearing in the IR limit of (projectable) HL gravity is mimetic gravity [802–804]: see [805] for a full proof of this equivalence, and [806, 807] for discussions on relations to Einstein-æther and HL gravity. Mimetic gravity is related to GR by a singular (non-invertible) disformal transformation, and has gained significant attention recently (see e.g. [808–833], and [834] for a review). Spherically symmetric solutions in mimetic gravity have been studied in [835–846], though the issue remains open as to whether mimetic gravity can actually support non-pathological BH solutions. A full investigation of the relation between Einstein-æther and mimetic gravity as far as BH solutions are concerned is therefore warranted, and is left to future work, where we also plan to study the shadows cast by spherically symmetric spacetimes (BHs, WHs, or NSs) in mimetic gravity.

## 3.27. 4D Einstein-Gauss-Bonnet gravity

The Gauss–Bonnet invariant G is a very important quantity, which in more than four dimensions plays a role in determining the allowed Lagrangian terms (Lovelock invariants) of diffeomorphism-invariant, metric theories of gravity with equations of motion of second order [847], and is given by:

<sup>&</sup>lt;sup>48</sup> This explicit diffeomorphism invariance breaking has been argued to lead to several pathologies in the IR, see e.g. [790–799] for discussions thereof, as well as possible ways out.

$$\mathcal{G} = R^{\mu\nu}_{\sigma\sigma} R^{\rho\sigma}_{\mu\nu} - 4R^{\mu}_{\nu} R^{\nu}_{\mu} + R^2 \,. \tag{93}$$

Crucially, in four dimensions G amounts to a total derivative, and therefore cannot contribute to the gravitational dynamics. However, in [848] Glavan and Lin proposed what effectively amounts to a dimensional regularization procedure applied to the following action:

$$S = \int \mathrm{d}^D x \sqrt{-g} \, \alpha \mathcal{G} \,, \tag{94}$$

by rescaling the coupling constant as  $\alpha \rightarrow \alpha/(D-4)$  (see also the generalization of [849]). This procedure is shown to lead to non-trivial contributions even in D=4, and we refer to the resulting theory as 4D Einstein–Gauss–Bonnet gravity (4DEGB). See e.g. [850] for a recent review of the theory, [851–860] for important discussions on possible pathologies associated to the rescaling procedure, and [861–880] for other important works on the theory.

With all the caveats discussed in [851, 853, 855–859] in mind, static spherically symmetric solutions within 4DEGB gravity have been studied, and the metric function is given by the following (see e.g. [112, 115–117, 848, 881–890]):

$$A(r) = 1 + \frac{r^2}{32\pi\alpha} \left( 1 \pm \sqrt{1 + \frac{128\pi\alpha M}{r^3}} \right),$$
(95)

which, although  $\alpha$  characterizes universal hair, requires  $\alpha \leq M^2/16\pi \sim 0.02M^2$ , i.e.  $\alpha \leq O(10^{18}) \,\mathrm{m}^2$  for Sgr A\*. We need to take the negative branch of equation (95), as it is the only one which recovers the Schwarzschild metric function A(r) = 1 - 2M/r in the limit  $r \to \infty$ .

We compute the shadow radius numerically, and show its evolution as a function of  $\alpha$  in figure 31.<sup>49</sup> We see that the shadow radius decreases with increasing  $\alpha$ . However, for all values of  $\alpha \leq 1/16\pi$ , the shadow radius is always consistent with the EHT observations within better than  $1\sigma$ . Therefore, the size of Sgr A\*'s shadow does not place meaningful constraints on the 4DEGB coupling  $\alpha$ , which is instead constrained by alternative astrophysical probes. For instance, the tightest constraint on  $\alpha$  comes from binary BH systems, which limit  $\alpha \leq 10^8 \text{ m}^2$  [891]. For comparison, it is useful to recast the previous limit in units of Sgr A\*'s mass: noting that Sgr A\*'s gravitational radius is of order  $r_{\rm g} \approx \mathcal{O}(10^7)$  km, the previous constraint translates to the extremely stringent limit  $\alpha \leq 10^{-28}M^2$ , much tighter than any constraint realistically achievable from BH shadows.

## 3.28. Asymptotically safe (AS) gravity

The issue of the non-renormalizability of GR, currently one of the biggest obstacles towards constructing a full quantum theory of gravity, has prompted a number of interesting ideas, many of which conjecture the existence of novel symmetries. AS gravity broadly falls within this category, in that it provides a quantum realization of scale symmetry. While, classically, the absence of dimensionful couplings is enough to ensure the realization of scale symmetry, this is no longer true at the quantum level, due to quantum fluctuations in the vacuum. In AS gravity, quantum scale symmetry is realized thanks to the presence of an interacting fixed point of the renormalization group (RG) [892, 893]: beyond a certain energy scale, a transition to a quantum scale-invariant regime occurs, and the strength of the gravitational constant flows to a constant but non-zero value (the case where this constant is zero would essentially correspond to a realization of asymptotic freedom). Therefore, within AS gravity quantum scale symmetry

<sup>&</sup>lt;sup>49</sup> We have found that a closed-form expression for the shadow radius exists, but is too cumbersome to report here. We have nonetheless verified that our numerical results perfectly reproduce those provided by this closed-form expression.



**Figure 31.** Same as in figure 3 for the BH in 4D Einstein–Gauss–Bonnet gravity with metric function given by equation (95), as discussed in section 3.27.

is achieved at fixed points of the RG, enabling an UV extension of GR (interpreted as an effective field theory) and restoring the predictivity of the latter in the UV. Currently, AS gravity is considered a very promising approach towards a theory of quantum gravity (see [894–897] for reviews), and has been the subject of study of an enormous body of literature, see e.g. [898–929] for important examples.

RG-improved BH space-times within the framework of AS gravity were first studied in detail by Bonanno and Reuter in [930], considering the following expression for the running of the gravitational constant as a function of momentum k:

$$G(k) = \frac{G(k_0)}{1 + \omega G(k_0)(k - k_0)^2},$$
(96)

where  $k_0$  is a reference scale later set to  $k_0 \rightarrow 0$ , so that  $G(k_0)$  effectively corresponds to Newton's constant  $G_N$  as measured by low-energy observations. Close to the RG fixed point, i.e. for  $k^2 \gg G_N^{-1}$ , the gravitational constant runs as  $G(k) = 1/\omega k^2$ , where  $\omega$  is the inverse dimensionless fixed point value: thus, the gravitational constant 'forgets' its IR value when entering the quantum scale-invariant regime. To consider the impact of equation (96) on BH space-times, one needs a prescription for converting k to a position-dependent IR cutoff. Various well-motivated prescriptions were studied in [930]. Here, we simply quote the final expression, leaving the reader to [930] for further details:

$$G(r) = \frac{G_N r^3}{r^3 + \widetilde{\omega} G_N (r + \gamma G_N M)},$$
(97)

where *M* is the BH mass, and  $\tilde{\omega}$  is an in principle free parameter, related to the inverse dimensionless fixed point value  $\omega$ , and expected to be of order unity from first principles considerations<sup>50</sup>. To obtain the RG-improved AS-driven corrections to the Schwarzschild space-time, one therefore simply replaces *G* in the latter with equation (97), after identifying *r* with the

<sup>&</sup>lt;sup>50</sup> It is worth noting that the arguments presented in [931] lead to the specific prediction of  $\tilde{\omega} = 118/15\pi \approx 2.5$ .



**Figure 32.** Same as in figure 3 for the BH in asymptotically safe gravity (in the IR limit) with metric function given by equation (98), as discussed in section 3.28.

usual radial coordinate (see also [83, 932–947] for other relevant works on BHs in AS gravity). As argued explicitly in [930],  $\gamma$  plays a very minor role in the phenomenology of the model, especially at the scales relevant for BH observations. Therefore, in the following, we shall set  $\gamma = 0$ , which effectively corresponds to considering the IR limit ( $r \gg \sqrt{\gamma}$ ) of equation (97). Explicitly, the metric function we consider is therefore given by (where we have obviously set  $G_N = 1$  for convenience):

$$A(r) = 1 - \frac{2Mr}{r^2 + \widetilde{\omega}},\tag{98}$$

where  $\tilde{\omega}$  characterizes an universal hair. One can easily show that the Kretschmann and Ricci scalars associated to the above metric are finite throughout the entire space-time, which therefore describes a globally regular BH. Because the curvature scalars are finite everywhere, including r = 0, the metric describes a globally regular BH space-time. The impact of this and related space-times on BH observables such as shadows and gravitational deflection of light have been recently studied in [83, 948].

Here, we shall focus on the region  $\tilde{\omega} < M^2$ , for which the BH possesses two horizons. We compute the shadow size numerically, and show its evolution against  $\tilde{\omega}$  in figure 32. We see that the EHT observations set very weak limits on  $\tilde{\omega}$ , with the upper limit  $\tilde{\omega} \leq 0.9M^2$  at  $1\sigma$ . Converting this to the dimensionless  $\tilde{\omega}$ , we find that this corresponds to a very weak limit, and we are left to conclude that based on this metric test the EHT observations are unable to set strong constraints on AS gravity (though the situation may be different with other types of tests, see e.g. [942, 943, 945–947]). On the other hand, complementary x-ray reflection spectroscopy constraints obtained by one of us set a much tighter limit,  $\tilde{\omega} \leq 0.05$  at  $1\sigma$  [949] from observations of GRS 1915+105, an x-ray binary system with mass of order  $\mathcal{O}(10)M_{\odot}$ .

### 3.29. Rastall gravity

Rastall's theory of gravity is perhaps one of the best known and studied non-conservative theories of gravity, i.e. theories where the stress-energy tensor is not covariantly conserved,

 $\nabla_{\mu}T^{\mu\nu} \neq 0$ . The theory was introduced by Rastall in [950] with the motivation that in the strong curvature regime the covariant derivative of the stress-energy tensor may not vanish but could depend on the space-time curvature. Based on phenomenological arguments, Rastall proposed that  $\nabla_{\mu}T^{\mu\nu} \propto \lambda g^{\mu\nu} \nabla_{\mu}R$ , with Einstein's equations being modified to:

$$R_{\mu\nu} + \left(\kappa - \frac{1}{2}\right)g_{\mu\nu}R = T_{\mu\nu}, \qquad (99)$$

where we refer to  $\kappa$  as the Rastall coupling parameter, and Einstein's equations are recovered in the limit  $\kappa \rightarrow 0$ . Rastall gravity has been at the center of a surge of interest in recent years, with applications to a number of interesting scenarios from thermodynamics to cosmology (see e.g. [170, 951–969]), although we note that concerns have been raised about its possibly being equivalent to GR in the gravity sector, modulo an artificially isolated part of the physical conserved stress-energy tensor in the matter sector [970].

By construction, while all vacuum solutions of GR are also solutions of Rastall gravity, the non-vacuum solutions depend on the Rastall coupling parameter and can be significantly different from their GR counterparts. BH solutions in Rastall gravity have been studied in several works [971–973]. Here, we shall consider the solution obtained by Heydarzade and Darabi [972], describing BHs in Rastall gravity surrounded by a perfect fluid with equation of state  $\omega$ . In this case, the metric function is given by:

$$A(r) = 1 - \frac{2M}{r} - N_{\rm s} r^{\frac{6\kappa(1+\omega)-1-3\omega}{1-3\kappa(1+\omega)}},$$
(100)

where  $N_s$  is the field structure parameter, which is related to the energy density of the surrounding fluid and therefore characterizes a specific hair, as does  $\omega$ , whereas  $\kappa$  characterizes an universal hair. The reason why it is interesting to consider a surrounding fluid is that, as is clear from equation (100), depending on the value of  $\kappa$  the surrounding field acts as if it had a different effective equation of state. In the following, we aim to focus on the Rastall coupling  $\kappa$ , while also exploring the role of the equation of state of the surrounding fluid. In our baseline cases, we shall therefore fix  $N_s = 0.005$  (so that it has little impact on the resulting shadow size), considering the cases where the BH is surrounded by dust ( $\omega = 0$ ), which we later extend to a stiff fluid ( $\omega = 1$ , which can be generated during a period of kination, where the energy is dominated by the kinetic energy of a scalar field), radiation ( $\omega = 1/3$ ), and a fluid with equation of state corresponding to spatial curvature).

We compute the resulting shadow size numerically, and show its evolution as a function of the Rastall coupling  $\kappa$  in figure 33. For the cases where the equation of state of the surrounding fluid is  $\omega \ge 0$ , we set the upper limit  $\kappa \le 1/6$  [212], ensuring that the energy conditions are not violated (by construction this cannot be done for the  $\omega = -1/3$  fluid) and the space-time is asymptotically flat. In these cases, and in particular for our baseline case with  $N_{\rm s} = 0.005$ and  $\omega = 0$ , we see that the shadow size grows extremely slowly with increasing  $\kappa$ , in a way which is hardly visible by eye. We find that the largest correction to the shadow size relative to the Schwarzschild counterpart within the whole  $\kappa \le 1/6$  regime is at most  $\mathcal{O}(10^{-2})$ , and as a result the EHT observations are unable to set meaningful constraints on the Rastall coupling parameter  $\kappa$ . On the other hand, for  $\omega = -1/3$  the shadow size starts increasing quickly once  $\kappa \gtrsim 0.25$ , and we find the upper limits  $\kappa \lesssim 0.01$  ( $1\sigma$ ) and  $\kappa \lesssim 0.25$  ( $2\sigma$ ). To further show the role of  $N_{\rm s}$  instead, we return to our baseline case with  $\omega = 0$ , but setting  $N_{\rm s} = 0.1$ . In this case



**Figure 33.** Same as in figure 3 for the BH in Rastall gravity with metric function given by equation (100), with the field structure parameter  $N_s$  and equation of state of the surrounding fluid  $\omega$  fixed to different values (our baseline case corresponds to  $N_s = 0.005$  and  $\omega = 0$ ), as discussed in section 3.29.

we see that the shadow size increases more markedly, and is excluded by the EHT observations within  $1\sigma$ , while at  $2\sigma$  we find the upper limit  $\kappa \leq 0.07$ .

# 3.30. Loop quantum gravity (LQG)

LQG is arguably one of the leading approaches towards constructing a quantum theory of gravity. One of the main hurdles towards constructing such a theory is the difficulty in constructing a quantum Riemannian geometry, going beyond the classical (fixed) reference space-time. The first steps towards this task were completed by reformulating gravity without reference to any background field (not even a metric), and subsequently introducing non-perturbative techniques from gauge theories, once more without reference to any background (see e.g. [974–978]). A theory with no background fields only has access to an underlying manifold and is therefore inevitably diffeomorphism-covariant: this, together with the non-perturbative methods mentioned previously, leads to a fundamentally discrete geometry, whereas the familiar continuum space-time of GR results as an emergent phenomenon upon a coarse grained description of the fundamental quantum Riemannian discrete structures. Summarizing the main features of LQG in a few lines is close to impossible: loosely speaking, we can say that LQG leads to a granular space-time structure, with a minimum length of order the Planck length, and 'woven' out of Planck-scale loops whose network and evolution thereof are referred to as spin network and spin foam respectively. A significant amount of work has been devoted to LQG in recent years, and we refer the reader to [979–1008] for examples thereof (including cosmological applications), as well as [1009] for a comprehensive review of the status of LOG.

BHs in LQG have been explored in several works [985, 1010–1013]. Here we shall consider the LQG BH constructed in [1014], whose shadow was studied by one of us in [111], and described by the following metric functions:



**Figure 34.** Same as in figure 3 for the loop quantum gravity BH with metric described by equation (101), as discussed in section 3.30.

$$A(r) = \frac{(r - r_{+})(r - r_{-})(r + r_{*})^{2}}{r^{4} + a_{0}^{2}}, \quad C(r) = r^{2} + \frac{a_{0}^{2}}{r^{2}}, \quad (101)$$

where  $r_+ = 2M/(1+P)^2$  and  $r_- = 2MP^2/(1+P)^2$  are the BH's two horizons, while  $r_* = \sqrt{r_+r_-} = 2MP/(1+P)^2$ . On the other hand  $P = (\sqrt{1+\epsilon^2}-1)/(\sqrt{1+\epsilon^2}+1)$  is the polymeric function, where  $\epsilon$  denotes the product of the Immirzi parameter  $\gamma$  and the polymeric parameter  $\delta$  (where typically  $\epsilon = \gamma \delta \ll 1$ ): the former is an area parameter [1015–1017], whereas the latter was introduced to define the Hamiltonian constraint in terms of holonomies [1018], and was conservatively assumed to be constant in the approach of [1014]. Finally,  $a_0 = A_{\min}/8\pi$  is related to the minimum area gap of LQG, i.e. the smallest non-zero eigenvalue of the kinematical area operator. Note that all three  $a_0$ ,  $\gamma$ , and  $\delta$  (and therefore  $\epsilon$  and P) characterize universal hairs.

The above space-time depends upon the two parameters P (or equivalently  $\epsilon$ ) and  $a_0$ , and reduces to the Schwarzschild space-time in the limit  $P \rightarrow 0$ ,  $a_0 \rightarrow 0$ . However, since  $a_0$  is expected to be extremely small in LQG, we do not expect it to play a relevant role as far as astrophysical BHs are concerned. Indeed, we have explicitly verified that  $a_0$  plays a minor role in determining the shadow size. Therefore, for the purposes of this work, we set  $a_0 = 0$ , so that the only free parameter is P. We compute the shadow size numerically, and show its evolution as a function of P in figure 34.<sup>51</sup> We see that increasing the polymeric function decreases the shadow size, and the EHT observations set the upper limits  $P \leq 0.05$  ( $1\sigma$ ) and  $P \leq 0.08$  ( $2\sigma$ ). These can be translated into the upper limits  $\epsilon \leq 0.5$  ( $1\sigma$ ) and  $\epsilon \leq 0.6$  ( $2\sigma$ ), which are rather loose as on theoretical grounds we expect  $\epsilon \ll 1$ . However, this remains a valuable constraint on an interesting theoretical scenario, particularly given that observational constraints on LQG are not easy to obtain.

<sup>&</sup>lt;sup>51</sup> We have found that a closed-form expression for the shadow radius exists, but is far too cumbersome to report here, to the extent that it would easily occupy an entire page, but would lead to little physical insight. We have nonetheless verified that our numerical results perfectly reproduce those provided by this closed-form expression.

## 3.31. Kottler black hole

The Kottler space-time [213], also known as Schwarzschild–de Sitter space-time, is the unique spherically symmetric solution to Einstein's equations sourced by a positive cosmological constant  $\Lambda$ . It is a particular case of the McVittie metric, which describes models interpolating between Schwarzschild and FLRW space-times in a continuous manner, and is part of a larger class of space-times which generically describe BHs embedded in an expanding Universe (see e.g. [1019–1043] for further relevant works). The Kottler BH is described by the following metric function [213]:

$$A(r) = 1 - \frac{2M}{r} - \frac{\Lambda}{3}r^2,$$
(102)

where  $\Lambda$  characterizes an universal hair. The Kottler space-time possesses two horizons for  $0 < \Lambda < \sqrt{3M}$ , with the vector field  $\partial_t$  only being time-like between these two horizons. We therefore restrict our discussion to static observers whose radial coordinate is bounded above by the coordinate of the outer horizon, which is anyhow always guaranteed to be the case since the latter corresponds to our de Sitter cosmological horizon in the physically interesting case where  $\Lambda$  is the cosmological constant responsible for cosmic acceleration.

As discussed at length in [43], the Kottler space-time is an excellent example of a BH whose shadow size depends explicitly on the radial coordinate of the observer, due to the metric described by equation (102) being manifestly non-asymptotically flat. Considering the importance of the Kottler space-time, we shall derive the exact expression for the shadow size, following the discussion of section 2.1. Explicitly, we find:

$$r_{\rm sh} = \frac{3\sqrt{3}M}{\sqrt{1 - 9\Lambda M^2}} \sqrt{1 - \frac{2M}{r_{\rm O}} - \frac{\Lambda}{3}r_{\rm O}^2},$$
(103)

which clearly reduces to  $3\sqrt{3}M$  in the limit  $\Lambda \to 0$  and  $r_0 \gg M$ , the latter being clearly the case for the EHT observations of Sgr A<sup>\*</sup>. We further note that the requirement of the observer lying within the cosmological horizon ensures that the term in the square root to the right of equation (103) does not turn negative. As per our discussion in section 2.1, for the purposes of our discussion it suffices to consider a static observer, whereas considering an observer comoving with the expansion of the Universe would complicate the discussion considerably.

We show the evolution of the shadow size as a function of the cosmological constant  $\Lambda$  in equation (35). As a note of caution, we plot the shadow radius in units of the Newtonian mass *M*: since the Kottler space-time is not asymptotically flat, the ADM mass is not well-defined, however for the purposes of our discussion considering M will be sufficient. Figure 35 displays the features we could have expected from a qualitative assessment. For  $\Lambda \lesssim 10^{-41} \, \text{m}^2$ , the product  $\Lambda r_0^2 \ll 1$  considering  $r_0 \sim 8 \,\mathrm{kpc} \simeq 2.5 \times 10^{20} \,\mathrm{m}$ , and therefore the size of the Kottler BH is essentially that of a Schwarzschild BH of the same mass. Clearly, this includes the value of  $\Lambda$  indicated by cosmological observations,  $\Lambda \sim \mathcal{O}(10^{-52})\,m^2$  (corresponding to the dashed vertical line in figure 35), for which the product  $\Lambda r_0^2 \sim \mathcal{O}(10^{-12})$ , clearly too small to be appreciated by current EHT measurements. Only for larger values of  $\Lambda$ , when  $\Lambda r^2$  becomes of order unity, does the effect of the cosmological constant become visible, acting to quickly decrease the shadow size. We therefore conclude that within a Kottler space-time where  $\Lambda$ is responsible for cosmic acceleration, there is no detectable imprint on the size of the BH shadow, as we could have expected from back-of-the-envelope calculations, but reassuringly confirmed by the full computation (see also [160] by two of us for different perspectives on the subject, as well as [1044, 1045]).



**Figure 35.** Same as in figure 3 for the Kottler BH with metric function given by equation (102), as discussed in section 3.31. The dashed black vertical line corresponds to the value of the cosmological constant indicated by observations,  $\Lambda \sim 1.1 \times 10^{-52} \,\mathrm{m}^2$ .

# 3.32. Electromagnetic-Weyl coupling

One of the simplest extensions of the Einstein–Maxwell equations of GR and electromagnetism involves a direct coupling between the electromagnetic and Weyl tensors. The Weyl tensor  $C_{\mu\nu\rho\sigma}$ , invariant under conformal transformations of the metric, is a measure of the space-time shear, and is given by the following [1046]:

$$C_{\mu\nu\rho\sigma} = R_{\mu\nu\rho\sigma} - \left(g_{\mu[\rho}R_{\sigma]\nu} - g_{\nu[\rho}R_{\sigma]\mu}\right) + \frac{1}{3}Rg_{\mu[\rho}g_{\sigma]\nu}, \qquad (104)$$

where the square brackets indicate the antisymmetric part. A direct coupling between the electromagnetic and Weyl tensors is allowed, in which case the electromagnetic sector of the Lagrangian is given by [1047, 1048]:

$$\mathcal{L} = -\frac{1}{4} F_{\mu\nu} F^{\mu\nu} + \alpha C^{\mu\nu\rho\sigma} F_{\mu\nu} F_{\rho\sigma} , \qquad (105)$$

where  $\alpha$  is a constant with dimensions of length squared. Assuming that the gravitational sector is still described by GR, Maxwell's equations are modified as follows:

$$\nabla_{\mu} \left( F^{\mu\nu} - 4\alpha C^{\mu\nu\rho\sigma} F_{\rho\sigma} \right) = 0. \tag{106}$$

Clearly, the coupling to the Weyl tensor modifies the propagation of photons across the spacetime, leading to a series of interesting phenomena including apparent superluminal propagation and birefringence [1049–1051]. This model is perhaps not a model of modified gravity in a strict sense, but is perhaps best interpreted as a model of modified electrodynamics, with such modifications actually expected when considering the effects of one-loop vacuum polarization on the photon effective action in QED as well as the effects of tidal gravitational forces (see [1047] for early discussions). As in most modified electrodynamics models, here as well photons travel along null geodesics of an effective geometry. We follow [1052] who computed the effective geometry experienced by photons traveling in a Schwarzschild space-time in the presence of electromagnetic-Weyl coupling. As discussed in [1052], only two of the three possible polarizations turn out to be physical (likely as a consequence of gauge invariance and the Ward–Takahashi identities), which the authors refer to as  $\ell$ -polarization and *m*-polarization, or PPL and PPM respectively: we refer the reader to [1052] for further details on the derivation of the modified light-cone conditions and of the physical polarizations. The Weyl-corrected effective geometry corresponding to an underlying Schwarzschild space-time is then described by the following metric functions [1052] (see also [1053–1062] for further relevant works):

$$A(r) = 1 - \frac{2M}{r},$$

$$C(r) = r^2 \left(\frac{r^3 - 8\alpha M}{r^3 + 16\alpha M}\right)^{-1} \quad (PPL \text{ polarization}),$$

$$C(r) = r^2 \left(\frac{r^3 + 16\alpha M}{r^3 - 8\alpha M}\right)^{-1} \quad (PPM \text{ polarization}),$$

$$(107)$$

where  $\alpha$  characterizes an universal hair. The fact that only the angular metric function C(r) is modified is a direct reflection of the modifications to the light-cone structure due to the direct coupling between photons and the Weyl tensor.

We compute the shadow size numerically for both PPL- and PPM-polarized photons, and show its evolution as a function of the Weyl coupling  $\alpha$  in figures 36 and 37 for PPL- and PPM-polarized photons respectively. We see that in the former case increasing  $\alpha$  increases the shadow size, whereas the reverse happens in the latter case: this is not surprising, as the respective angular metric functions are the inverse of one another. For PPL-polarized photons, we find that the EHT observations set the upper limits  $\alpha \leq 0.01M^2$  ( $1\sigma$ ) and  $\alpha \leq 0.2M^2$  ( $2\sigma$ ). For PPM-polarized photons, the limits are instead looser and read  $\alpha \leq 0.25M^2$  ( $1\sigma$ ) and  $\alpha \leq 0.3M^2$  ( $2\sigma$ ). Taking  $\alpha \leq \mathcal{O}(0.1)M^2$  as a representative limit, in SI units this translates to  $\alpha \leq \mathcal{O}(10^{17}) \text{ m}^2$ , significantly weaker than some of the best constraints on this parameter arising from gravitational time delay/advancement, light deflection, and frequency shifts, with Cassini tracking setting the upper limit  $\alpha \leq \mathcal{O}(10^{12}) \text{ m}^2$  [1063, 1064].

## 3.33. DST black hole

In [1065], Deser, Sarioglu, and Tekin (DST hereafter) introduced a class of gravitational models whose modifications to GR forbid Ricci-flat geometries, while preserving as many features of GR as possible, including the derivative order of the Einstein equations in Schwarzschild gauge. This class of models was introduced as a tool to better appreciate the nature of Ricci-flat geometries, and relies on the addition to the Einstein–Hilbert Lagrangian of terms which are non-polynomial in the Weyl tensor, as follows:

$$\Delta \mathcal{L} = \sqrt{-g} \beta_n \left| \operatorname{tr} \mathcal{C}^n \right|^{\frac{1}{n}},\tag{108}$$

where C is the Weyl tensor already encountered earlier in equation (104), the index *n* labels the various models belonging to this class, and the trace operator acts upon the (mixed) Weyl tensor as follows:

$$\operatorname{tr} \mathcal{C}^{n} \equiv C_{ab}^{\ cd} C_{cd}^{\ ef} \dots C_{..}^{\ yz} C_{yz}^{\ ab}, \tag{109}$$



**Figure 36.** Same as in figure 3 for the Weyl-corrected (PPL polarization) BH with metric described by equation (107), as discussed in section 3.32.



**Figure 37.** Same as in figure 3 for the Weyl-corrected (PPM polarization) BH with metric described by equation (107), as discussed in section 3.32.

for *n* copies of the Weyl tensor, where *n* can be any arbitrary positive integer. To the best of our knowledge, [1065] was the first to introduce the non-polynomial gravity approach. Models of these type have later been considered in various works (particularly in the search for regular BHs), see for instance [1066–1075].

Without loss of generality, we shall follow [1065] and take n = 2, while defining  $\sigma = \beta_2/\sqrt{3}$ , so that equation (108) reduces to:

$$\Delta \mathcal{L}_{n=2} = \sqrt{-g} \,\sigma \sqrt{3 \left| C_{ab}^{\ cd} C_{cd}^{\ ef} \right|} \,. \tag{110}$$

This particular Lagrangian admits an explicit static spherically symmetric solution, the so-called DST BH, with metric function given by [1065]:

$$A(r) = \left[\frac{1-\sigma}{1-4\sigma} - \left(\frac{2M}{r}\right)^{\frac{1-4\sigma}{1-\sigma}}\right] \left(\frac{r}{2M}\right)^{\frac{6\sigma}{\sigma-1}},\tag{111}$$

where  $\sigma$  characterizes an universal hair. It is easy to show that the above space-time is asymptotically flat for  $0 < \sigma < 1$ . In our subsequent analysis we have imposed  $\sigma > 0$ , whereas the upper limit guaranteeing asymptotic flatness  $\sigma < 1$  turns out *a posteriori* to be satisfied given the very stringent limits on  $\sigma$  imposed by the EHT observations.

While slightly cumbersome, there exists a closed-form expression for the shadow radius, given by:

$$r_{\rm sh} = \frac{2^{\frac{3\sigma}{\sigma-1}} 3^{\frac{\sigma-1}{4\sigma-1}} \mathcal{X}}{\sqrt{\left(3^{\frac{\sigma-1}{4\sigma-1}} \mathcal{X}\right)^{\frac{6\sigma}{\sigma-1}} \left(\frac{\sigma-1}{4\sigma-1} - 2^{\frac{4\sigma-1}{\sigma-1}} \left(3^{\frac{1-\sigma}{4\sigma-1}} \mathcal{X}\right)^{\frac{4\sigma-1}{\sigma-1}}\right)}},$$
(112)

where the quantity  $\mathcal{X}$  is defined as:

$$\mathcal{X} \equiv \frac{8^{\frac{\sigma}{\sigma-1}}(4\sigma-1)}{(\sigma-1)(2\sigma+1)},$$
(113)

which clearly reduces to  $\mathcal{X} \to 1$  when  $\sigma \to 0$ , from which it is relatively straightforward to see that the limit  $r_{\rm sh} = 3\sqrt{3}$  is indeed recovered when  $\sigma \to 0$  in equation (112). We show the evolution of the shadow size as a function of  $\sigma$  in figure 38, where we see that increasing  $\sigma$  quickly decreases the shadow size. We note that the EHT observations set the very stringent limits  $\sigma \leq 0.04 (1\sigma)$  and  $\sigma \leq 0.05 (2\sigma)$ , which directly translate into upper limits on the Lagrangian coupling  $\beta_2$  in equation (110) of  $\beta_2 \leq 0.07 (1\sigma)$  and  $\beta_2 \leq 0.09 (2\sigma)$ . To the best of our knowledge, we are not aware of any observational constraints on  $\sigma$ : a study on collisions of spinning particles near DST BHs was conducted by one of us in [1075], finding a behavior very close to that of GR, while the same work entertained the possibility that a very tiny discrepancy in the observed value of Mercury's perihelion precession might be explained by a value  $\sigma \sim \mathcal{O}(10^{-9})$ . However, our upper limits of order  $\mathcal{O}(10^{-2})$  on  $\sigma$  represent the first ever robust observational constraints on a non-polynomial gravity model.

## 3.34. Rindler gravity

In [1076] (see also [1077]), Grümiller constructed *ab initio* the most general theory for gravity in the IR (i.e. at large distances) consistent with an assumed set of symmetries and assumptions, including spherical symmetry, power counting renormalizability, analyticity, and so on. While the model was constructed based on well-founded theoretical assumptions, it also enjoys phenomenological motivation in its ability (at the time) to potentially mimic DM and/or dark energy, as well as explain a number of puzzles such as the flyby [1078] and Pioneer anomalies [1079]. While the theory is now understood to be unlikely to successfully mimic DM and dark energy (see e.g. [1080]), and interest in these anomalies has now partially faded [1081], here we shall simply consider it as being an interesting IR model for gravity.

In [1076], Grümiller constructed the Rindler gravity model from the following effective field theory for a scalar field  $\Phi$  in 2 dimensions:

$$S = \int d^2 x \sqrt{-g} \left[ \Phi^2 R + 2(\partial \Phi)^2 - 6\Lambda \Phi^2 + 8a\Phi + 2 \right],$$
(114)



**Figure 38.** Same as in figure 3 for the DST BH with metric function given by equation (111), as discussed in section 3.33.

where  $\Lambda$  will eventually play the role of dark energy, and *a* is a constant with SI units of acceleration. For a test particle located at large distance from a source (where the Newtonian acceleration is negligible), positive/negative *a* provides a constant acceleration towards/away from the source. The new term admits a simple interpretation: a Rindler acceleration, see [1082]. While the above theory is written in 2 dimensions and thus cannot be a realistic description of our world, given the interesting phenomenology associated to the Rindler acceleration outlined in [1076], here we shall adopt an equally phenomenological standpoint where we examine the consequences of a Rindler correction to the Schwarzschild metric (setting  $\Lambda = 0$ ), taking the above model as theoretical motivation. That is, we consider the following metric function [1076] (see also [1083, 1084]):

$$A(r) = 1 - \frac{2M}{r} + 2ar,$$
(115)

where a characterizes an universal hair, and which is manifestly not asymptotically flat<sup>52</sup>.

We compute the shadow size numerically, using the full computation outlined in section 2.1 for non-asymptotically flat metrics, and show its evolution against the Rindler acceleration *a* in figure 39, considering only positive values of *a* as required to explain the observational anomalies the model was constructed to account for [1076]. For convenience, rather than reporting the shadow evolution against *aM*, we report *a* in units of  $pc^{-1}$ . We also note that a closed-form expression for the shadow radius can be found, given by:

$$r_{\rm sh} \simeq \frac{\sqrt{3} \left(\sqrt{1+12a}-1\right) \sqrt{1+2ar_{\rm O}}}{2a \sqrt{2\sqrt{1+12a}-1}} \tag{116}$$

which, although not obvious at first glance, does indeed reduce to  $3\sqrt{3}$  as  $a \rightarrow 0$ , as can be seen by Taylor expanding equation (116) in *a* (we have made the valid approximation that

<sup>&</sup>lt;sup>52</sup> We note that this same space-time can arise in a wide range of theories other than Rindler gravity, see e.g. [1085–1091]. It is also a particular case of the Kiselev BH [1092], with equation of state of the surrounding fluid given by w = -2/3.



**Figure 39.** Same as in figure 3 for the Rindler BH with metric function given by equation (115), as discussed in section 3.34. The dashed black box corresponds to the region possibly indicated by observational hints (explaining DM and/or the Pioneer and flyby anomalies).

whenever the  $2ar_0$  term is dominant over 1, it naturally also dominates over  $2/r_0$ ). We have verified that this closed-form expression is in excellent agreement with our numerical results.

As with the Kottler space-time considered previously in section 3.31, it is clear that the effects of Rindler gravity on the BH shadow only become significant when the term *ar* becomes sufficiently large. For smaller values of *a* required to mimic DM and/or address anomalous acceleration hints of the order of  $a \sim 10^{-10} \text{ m s}^{-2}$ ,<sup>53</sup> corresponding to the dashed black box in figure 39, the effect on the shadow size is too small to be appreciated by the EHT observations. We conclude that, for phenomenologically interesting values of *a*, this model is in excellent agreement with the EHT observations but cannot be distinguished from its Schwarzschild counterpart, at least not based on the metric test we have considered here.

#### 3.35. Black hole surrounded by perfect fluid dark matter

We have spent considerable time discussing BH solutions in modified theories of gravity, broadly defined. We shall now turn our attention to BHs surrounded by various forms of matter. The influence of external matter fields on BHs has been the subject of significant study, at least since the well-known Kiselev space-time was proposed [1092]. While particular emphasis has been placed on the possibility of quintessence surrounding BHs (see e.g. [1098–1117]), here we shall consider a much more strongly motivated scenario: that is, one where the BH is surrounded by DM. In fact, it is now an observationally established fact that DM is present in (almost) all galaxies, and is therefore expected to accumulate around SMBHs inhabiting galactic centers: therefore, it is worthwhile to consider how the surrounding DM back-reacts on the BH metric.

 $<sup>^{53}</sup>$  It is of course no coincidence that this if of the order of the MOND acceleration [1093, 1094], as Rindler gravity produces an interesting MOND-like phenomenology (see [1095–1097] for related work).

The above question has been considered in several works [1118–1120]. Specifically, consider a perfect fluid DM component, described by an equation of state w = 0. The metric function associated to such a configuration is then given by (see [1120]):

$$A(r) = 1 - \frac{2M}{r} + \frac{k}{r} \ln\left(\frac{r}{|k|}\right),\tag{117}$$

where *k* is an integration constant, ultimately physically connected to the amount of DM surrounding the BH, and which characterizes a specific hair. We note that the above metric is *not* the metric one would obtain when setting w = 0 in the well-known Kiselev metric [1118]. The reason, as explained in [1121], is that the Kiselev metric does not describe a space-time surrounded by a perfect fluid in the sense we are using here: in fact, it is easy to show that the energy-momentum tensor sourcing the Kiselev metric inevitably possesses shear stress components, and therefore cannot describe a perfect fluid [1121]. This is not the case, instead, for the space-time described by equation (117), which has been explicitly derived by starting from a perfect fluid source term (see [1122–1141] for other relevant works).

While the size of the shadow cast by the BH described by equation (117) is easy to compute numerically, we also note that a closed-form expression exists and is given by:

$$r_{\rm sh} = \frac{9kW\left(\frac{2}{3}e^{\frac{1}{3} + \frac{2}{k}}\right)^{\frac{3}{2}}}{2\sqrt{2 + 3W\left(\frac{2}{3}e^{\frac{1}{3} + \frac{2}{k}}\right)}}$$
(118)

where once more *W* denotes the principal branch of the Lambert *W* function. As a check, we find equation (118) to be in perfect agreement with our numerical results. We show the evolution of the shadow size as a function of the perfect fluid DM parameter *k* in figure 40, where we see that the shadow size quickly decreases as *k* increases. We find that the EHT observations set the upper limits  $k \leq 0.07M (1\sigma)$  and  $k \leq 0.15M (2\sigma)$ . We are not aware of an explicit discussion of the physical meaning of *k* in the literature. On dimensional grounds, we expect the combination  $c^2k/G$  to be related to the integrated amount of DM present in a shell surrounding the BH. Given the physical situation at play, we would expect the upper limits on k/M to constrain the integrated amount of DM present in a shell size of size given by the photon sphere, although this does not allow us to estimate the (dimensionless) numerical factor relating these two quantities. At any rate, a more complete discussion requires a full physical interpretation of the parameter *k*, which is currently lacking in the literature: we plan to return to this issue in future work, to which we defer a more comprehensive discussion on the topic.

#### 3.36. Black hole with a topological defect

Topological defects play an important role in several different context of physical interest, and likewise can be produced within a variety of different scenarios, including during phase transitions in the early Universe [1142], or associated to the spontaneous breaking of symmetries [1143–1146]. For example, global monopoles can arise in association to the breaking of a global O(3) symmetry down to a U(1) subgroup (this can be realized for instance with a triplet of scalar fields  $\phi^i$ ), and their stability has been studied in various works [1147–1149]. The gravitational field associated to a Schwarzschild BH carrying a global monopole charge was first studied in [1150, 1151] (see also [1152]), and the corresponding BH solution is described by the following metric function:

$$A(r) = 1 - k - \frac{2M}{r},$$
(119)



**Figure 40.** Same as in figure 3 for the BH surrounded by perfect fluid DM with metric function given by equation (117), as discussed in section 3.35.

where the fundamental interpretation of k depends on the model being examined. For instance, we can consider a triplet of scalar fields  $\phi^i$  satisfying a global O(3) symmetry, described by the following Lagrangian:

$$\mathcal{L} = \frac{1}{2} \left( \partial \phi^i \right)^2 - \frac{\lambda}{4} \left( \phi^i \phi^i - \eta^2 \right)^2, \qquad (120)$$

for which the associated global monopole is given by:

$$\phi^{i} = \eta f(r) \frac{x^{a}}{|\mathbf{\overline{x}}|}, \qquad (121)$$

with  $f(r) \to 1$  as  $|\bar{\mathbf{x}}| \to 0$ , and k in equation (119) is given by  $k = 8\pi\eta^2$ . In this case, k is therefore related to the amplitude of the monopole field. Interestingly, the same metric can describe a BH surrounded by a cloud of strings [1153], in which case k is associated to the gauge-invariant density of the cloud of strings  $\sqrt{-\gamma\rho} = k/r^2$ . In both cases, k characterizes a specific hair. See also [1154–1168] for further works on BHs carrying global monopole charges or surrounded by clouds of strings.

The above space-time is manifestly not asymptotically flat, itself a direct consequence of the presence of the global monopole charge: in fact, it is globally conical. We can easily find that the location of the photon sphere is given by:

$$r_{\rm ph} = \frac{3M}{1-k} \,. \tag{122}$$

To obtain the shadow size we can exploit the fact that  $A(r_{\rm O}) \rightarrow \sqrt{1-k}$  as  $r_{\rm O} \rightarrow \infty$  (and more precisely for  $r_{\rm O} \gg M$ ), although we need to pay attention to the mass parameter which sets the scale of the shadow radius. In fact, performing the coordinate transformations  $t \rightarrow (1-k)^{-1/2}t'$ ,  $r \rightarrow (1-k)^{1/2}r'$ ,  $M \rightarrow (1-k)^{3/2}M'$ , it is easy to see that the ADM mass is given by  $\mathcal{M} = (1-k)M'$ . We then find that the shadow size is given by:

$$r_{\rm sh} = \frac{3\sqrt{3}\mathcal{M}}{\sqrt{1-k}}.$$
(123)



**Figure 41.** Same as in figure 3 for the Schwarzschild-like BH with a topological defect with metric function given by equation (119), as discussed in section 3.36.

The difference between whether we choose to report the shadow size in units of the ADM mass  $\mathcal{M}$  or the parameter M appearing in equation (119), then, amounts to a factor of (1 - k). Here, we choose to report the shadow size in units of  $\mathcal{M}$ , and show its evolution against k in figure 41. We find that the EHT observations set the limits  $k \leq 0.005$  (1 $\sigma$ ) and  $k \leq 0.1$  (2 $\sigma$ ), which correspond to limits on the amplitude of the global monopole field of  $\eta \leq 0.014$  and  $\eta \leq 0.06$  respectively. Given the very tight limits on k,  $1 - k \approx 1$  and the ambiguity associated to reporting the shadow radius in units of  $\mathcal{M}$  or M ends up making virtually no difference in our final results.

### 3.37. Hairy black hole from gravitational decoupling

Obtaining exact solutions to Einstein's equations in theories of gravity beyond GR is usually a difficult, if not impossible, task. While one case where this is possible is that where a perfect fluid acts as gravitational source, this ceases to be true as soon as the perfect fluid is coupled to more complex forms of matter-energy. In recent years, a very interesting technique has been proposed by Ovalle to overcome these difficulties: the minimal geometric deformation approach and its latter extension, the extended gravitational decoupling approach. We shall broadly refer to these approaches as 'gravitational decoupling'. This approach was initially proposed in the context of the Randall–Sundrum brane-world model [1169], but its power was quickly realized, to the extent that it has now been applied to an enormous variety of scenarios (see e.g. [1170-1219] for an inevitably incomplete list of examples). The strength of the gravitational decoupling approach is at least three-fold: (i) it allows to construct new viable BH solutions by extending simpler seed solutions; (ii) it can be used to systematically decouple a complex source energy-momentum tensor into more simple tractable components; and (iii) it can be utilized to find exact solutions in theories beyond GR. In the context of the present discussion, our interest in this approach rests upon its potential to describe BHs surrounded by various forms of matter. For complete details on the gravitational decoupling approach, we refer the reader to [1169, 1183, 1192].

Here, we shall base ourselves upon the work of [1220], where the authors applied the gravitational decoupling approach to obtain a minimal deformation of the seed Schwarzschild spacetime describing a hairy BH sourced by spherically symmetric 'tensor-vacuum'  $\theta_{\mu\nu}$  such that the existence of a well-defined event horizon is preserved, while the (primary) hair associated to the additional source does not violate the strong and dominant energy conditions. In short, [1220] obtains a general solution describing a spherically symmetric space-time which violates the NHT in a non-pathological way. In particular, the work finds that the charge associated to the primary hair increases the entropy of the space-time compared to its Schwarzschild value. Specifically, this BH solution is described by the following metric function [1220]:

$$A(r) = 1 - \frac{2M}{r} + \alpha \exp\left(-\frac{r}{M - \alpha\ell/2}\right),$$
(124)

where  $\alpha$  and  $\ell$  are the parameters which quantify deviations from the Schwarzschild metric, both characterizing specific hairs, with  $0 \le \ell \le 2M$  representing the charge associated to the primary hair and the Schwarzschild space-time is recovered for  $\alpha \ell = 2M$  (as well as in the physically less interesting case  $\alpha = 0$ ). In particular, the combination  $\ell_0 = \alpha \ell$  quantifies the increase in the BH entropy from its Schwarzschild value  $S = 4\pi M^2$ . The fluid surrounding the BH which sources this space-time can be associated to DM.

For illustrative purposes, in order to focus our study on the charge parameter  $\ell$ , we fix  $\alpha = 1.0$ , so that  $\ell_0 = \ell$ . We compute the resulting shadow size numerically, and show its evolution against  $\ell$  in figure 42. We see that decreasing  $\ell$  from  $\ell = 2M$  decreases the shadow size, consistently with earlier findings of some of us in [142]. For this particular choice of  $\alpha$ , we see that the EHT observations set the lower limits  $\ell \gtrsim 0.35M (1\sigma)$  and  $\ell \gtrsim 0.15M (2\sigma)$ . These limits can be translated into upper limits on the entropy increase compared to the Schwarzschild value  $S = 4\pi$  (in units where M = 1). In particular, we find that the associated hair cannot increase the entropy by more than 3%. Broadly speaking, this places constraints on the strength of NHT violation, in the presence of hair which does not violate the strong and dominant energy conditions. Finally, we stress that much as with a few cases studied earlier, these results only correspond to a slice of the full  $\alpha$ - $\ell$  parameter space, and fixing  $\alpha$  to smaller (larger) values would lead to weaker (tighter) constraints on  $\ell$  (for more details see [142]).

#### 3.38. Modified uncertainty principles (UPs)

The final scenarios we study can generically be referred to as 'novel fundamental physics frameworks', which do not necessarily modify gravity (at least not directly, although as we shall see they lead to a sort of 'backreaction' on the metrics of BHs in GR). We begin by discussing modified UPs which go beyond the Heisenberg one.

While GR is strictly speaking not incompatible with QM, the introduction of gravity into quantum field theories appears to spoil their renormalizability, which is one of the main reasons why the construction of a full theory of quantum gravity has eluded us so far. Various proposals for rendering gravity UV-complete have been explored, ranging from the addition of higher order curvature terms (it is known that adding terms quadratic in the curvature tensor to the Einstein–Hilbert action renders the theory renormalizable [1221–1223], albeit at the price of a loss of unitarity), to the use of Lorentz violation as a field theory regulator [1224].

However, it has long been suspected that gravity should lead to an effective UV cut-off, in turn leading to a minimum observable length and/or a maximum possible momentum, based on gedanken experiments wherein the high energies used to resolve small distances eventually 'disturb' the space-time structure due to their gravitational effects, to the point that space-time itself cannot be resolved below some distance. Support for these ideas range from various


**Figure 42.** Same as in figure 3 for the hairy BH from gravitational decoupling with metric function given by equation (124), as discussed in section 3.37.

general considerations, including within quantum gravity and QM [1225–1229], as well as string theory [1230–1234] (see also [1235–1244]). In the presence of a minimum length scale, gravity could therefore actually help rather than hamper achieving renormalizability.

These ideas have motivated approaches wherein the Heisenberg position-momentum commutator [x,p] = i, or equivalently the Heisenberg UP  $\Delta x \Delta p \ge 1/2$ , receives (quantum gravitational) non-linear corrections [1245–1247]. Such approaches are typically referred to as generalized uncertainty principle (GUP) or EUP, see e.g. [1248, 1249] for reviews. The notation and language when it comes to GUP and EUP is at times rather loose, but for consistency here we shall use the term 'GUP' to refer to the case where Heisenberg's UP receives corrections non-linear in momentum, typically associated to minimum length scenarios. On the other hand, we shall use the term 'EUP' to denote the case where the corrections are non-linear in position, as occurring in scenarios which introduce a (typically large) new fundamental scale (see e.g. [1250]). Here we shall first discuss GUP scenarios, before turning to EUP scenarios afterwards.

**3.38.1.** Generalized uncertainty principle. Various GUP models non-linearly deforming the Heisenberg UP have been proposed in the literature, in most cases driven by phenomenological considerations themselves based on theoretical expectations. One of the earliest proposals is that of Kempf, Mangano, and Mann, wherein Heisenberg's position-momentum commutator receives corrections non-linear in momentum, as per the following [1251]:

$$[x_i, p_j] = i\delta_{ij} \left( 1 + \beta_1 \ell_{\rm Pl}^2 p^2 \right), \tag{125}$$

where  $\ell_{\text{Pl}}$  is the Planck length and  $\beta_1$  is a dimensionless parameter. In turn, equation (125) implies the following corrected UP:

$$\Delta x \Delta p \ge \frac{1}{2} \left[ 1 + \beta_1 \ell_{\rm Pl}^2 (\Delta p)^2 \right], \tag{126}$$

with minimum measurable length given by  $\Delta x \approx \ell_{\text{Pl}} \sqrt{\beta_1}$ .

A different GUP scenario which admits both a minimum length and a maximum momentum was instead presented in [1252, 1253] (see also [1254–1256]), and modifies Heisenberg's position-momentum commutator as follows:

$$[x_i, p_j] = i \left[ \delta_{ij} - \beta \left( \delta_{ij} p + \frac{p_i p_j}{p} \right) + \beta^2 \left( \delta_{ij} p^2 + 3p_i p_j \right) \right],$$
(127)

where  $\beta = \beta_0 \ell_{\text{Pl}}$  and which in turn implies the following corrected UP:

$$\Delta x \Delta p \ge \frac{1}{2} \left( 1 - 2\beta \langle p \rangle + 4\beta^2 \langle p^2 \rangle \right) \,. \tag{128}$$

The minimum measurable length implied by equation (128) is given by  $\Delta x \approx \ell_{\text{Pl}}\beta_0$ , whereas the maximum momentum is given by  $\Delta p \approx \ell_{\text{Pl}}/\beta_0$ . Notice that the deformed commutators of equations (125) and (127) have both been introduced on phenomenological grounds: however, in both cases the expected value of  $\beta$  is of order unity, and can be obtained using low-energy formulations of string theory [1257, 1258], by studying corrections to Hawking's theorem for the BH temperature obtained from two different approaches, or by computing the leading quantum corrections to Newton's gravitational potential [1259].

In the following, we shall consider the GUP characterized by equations (127) and (128). GUP-deformed BH solutions have been studied in several papers, and here we shall build upon the results of [1260–1262], where it has been shown that GUP effects physically lead to what one could consider quantum corrections to the BH ADM mass, which is modified to  $\mathcal{M} = M + \beta M_{\text{Pl}}^2/M^2$ , where *M* can be interpreted as the 'bare' mass. Thus, the metric function for a GUP-corrected Schwarzschild BH reads [1260–1262]:

$$A(r) = 1 - \frac{2M\left(1 + \frac{\beta}{2M^2}\right)}{r},$$
(129)

where  $\beta$  characterizes an universal hair.

Despite the fact that various GUP tests have been performed in the literature using BHs and their environments (including their shadows), the form of the metric function given by equation (129) raises the question of whether one can actually discriminate a GUP-corrected BH from a 'genuine' Schwarzschild BH. In our case, we believe this might indeed be possible for Sgr A\*, for the following reason which echoes the argument presented previously for MOG in section 3.20. We interpret M as being the 'bare' mass of the system. Quantum effects which correct  $M \to \mathcal{M}$  effectively change the total mass/energy of the system, as measured by an observer at infinity (given the definition of ADM mass). However, a measurement of the BH mass through dynamical probes sufficiently close to the event horizon will effectively probe M and not  $\mathcal{M}$ : this is indeed the case for the S-stars orbiting Sgr A<sup>\*</sup> (and in particular the S2-star), whose motion has been used to obtain the exquisite mass-to-distance ratio measurements we adopt as priors. Moreover, the uncertainty on the mass (which does not disappear when we later fix M = 1, but is simply transferred to the mass-to-distance ratio) is part of what allows us to constrain the hair parameter  $\beta$ : we fix the bare mass, and ask how much of a change in the total mass/energy of the system we can tolerate when varying the hair parameter. While this physical interpretation has, to the best of our knowledge, never been explicitly discussed in the literature, it is the assumption which de facto underlies the many BH metric-based GUP tests (see e.g. [97, 1263-1266]).



**Figure 43.** Same as in figure 3 for GUP-corrected BH with metric function given by equation (129), as discussed in section 3.38.1.

Keeping in mind the assumptions and caveats discussed above, it is straightforward to show that the shadow radius associated to the GUP-corrected BH with metric function given by equation (129) is:

$$r_{\rm sh} = 3\sqrt{3}\mathcal{M} = 3\sqrt{3}M\left(1 + \frac{\beta}{2}\right). \tag{130}$$

We show the evolution of the shadow size as a function of  $\beta$  in figure 43. We see that the EHT observations set the upper limits  $\beta \leq 0.01M^2 (1\sigma)$  and  $\beta \leq 0.14M^2 (2\sigma)$ . Restoring the Planck mass to obtain the dimensionless GUP parameter  $\beta$ , we require  $\beta \leq 0.14(M/M_{\rm Pl})^2 \sim 2 \times 10^{88}$ . Unsurprisingly this is approximately six orders of magnitude stronger than the limit obtained from the shadow of M87\* in [97, 1264], as Sgr A\*'s mass is about three orders of magnitude smaller. However, this same limit is still weaker by several orders of magnitude compared to the limits obtained from independent tests such as perihelion precession, pulsar periastron shift, light deflection, gravitational waves, gravitational redshift, Shapiro time delay, geodetic precession, and quasiperiodic oscillations (see e.g. [1264, 1267, 1268]). Nonetheless, our limit remains a valuable complementary constraint on an fundamental physics scenario of great theoretical interest.

3.38.2. Extended uncertainty principle. In the GUP scenarios discussed so far, momentum appears to play a preferred role. However, other possibilities discussed in the literature envisage either a 'democratic' extension where both position and momentum play an equally important role (see e.g. [1250] by one of us), or scenarios where the corrections to Heisenberg's UP depend only on the position operator itself. It is worth reminding the reader once more that these deformations, while inspired by theoretical scenarios, are nonetheless introduced at the phenomenological level. With these caveats in mind, in the following we shall consider an EUP scenario, where Heisenberg's UP is corrected as follows:

$$\Delta x \Delta p \ge \frac{1}{2} \left[ 1 + \alpha \frac{(\Delta x)^2}{L_\star^2} \right], \tag{131}$$



**Figure 44.** Same as in figure 3 for the EUP-inspired BH with metric function given by equation (132), as discussed in section 3.38.2. The black shaded band denotes the range corresponding to galactic and sub-galactic scales  $L_{\star} \sim \mathcal{O}(0.1-10)$  kpc (with  $\alpha = 1$ ), where EUP-related effects could potentially mimic dark matter.

where  $L_{\star}$  is a new large fundamental distance scale, and  $\alpha$  is generically expected to be of order unity. It is fair to say that EUP scenarios have received much less attention than their GUP counterparts, the reason possibly being that the need for new macroscopic gravitational physics has only recently started to become appreciated. This is not unrelated to the recent realization that when dealing with BHs, quantum gravity effects may be relevant not only on microscopic scales, but perhaps more importantly on macroscopic horizon scales, and may well be tied to puzzling theoretical aspects such as the information paradox or singularity removal: see e.g. [1269–1271] for various proposals to address the information paradox which envisage new physics appearing at the gravitational radius of a system rather than at the Planck scale. Related recent works have put forward the possibility that the horizon itself may have a quantum origin, for example within the corpuscular framework, wherein BHs are described as Bose–Einstein condensates of N soft gravitons weakly confined to a region whose size is comparable to the BH gravitational radius [1272–1276]. See [182, 1277–1280] for further EUP-related works.

The effect of an EUP of the form given by equation (131) on BHs was studied in [1281]. The effect of the EUP is analogous to the GUP effect discussed in section 3.38.1, and amounts to (quantum) corrections to the BH ADM mass. As a result, the metric function of an EUP-corrected BH is given by [1281] (see also [1282, 1283]):

$$A(r) = 1 - \frac{2M}{r} \left( 1 + \frac{4\alpha M^2}{L_{\star}^2} \right),$$
(132)

where  $\alpha$  characterizes an universal hair. As is clear from equation (132), the EUP-inspired corrections are large only when  $L_{\star}$  is sufficiently small (this makes sense if we think of  $L_{\star}$  as corresponding to an inverse energy scale). We show the evolution of the shadow size as a function of  $\log_{10}(L_{\star}/\sqrt{\alpha})$ , with  $L_{\star}$  measured in pc, in figure 44. We find it convenient to express  $L_{\star}$  in these units since the possibility of partially mimicking the galactic effects of DM has been discussed in the literature [1281], and relies upon  $L_{\star}/\sqrt{\alpha}$  lying on  $\mathcal{O}(0.1-10)$  kpc

scales, i.e. those relevant to dwarf and spiral galaxies. From figure 44 we see that EUP effects do not lead to visible changes in the shadow radius as long as  $L/\sqrt{\alpha} \gtrsim 10^{-5}$  pc. For smaller values of  $L/\sqrt{\alpha}$ , the shadow radius increases drastically. Values of  $L/\sqrt{\alpha} \simeq 10^{-6}$  pc are completely excluded by the EHT observations, as the resulting shadow is much larger than what these observations allow. In addition, the black shaded band in figure 44 denotes the range corresponding to galactic and sub-galactic scales  $L_{\star} \sim \mathcal{O}(0.1-10)$  kpc (with  $\alpha = 1$ ) discussed previously, where EUP-related effects have been argued to potentially mimic the effects of DM. On these scales, EUP effects leave no visible imprint on the size of Sgr A\*'s shadow: this is to be expected, given that such scales are several orders of magnitude above the scale of Sgr A\*'s gravitational radius, which is in the  $\sim \mathcal{O}(10^{-7})$  pc range<sup>54</sup>.

### 3.39. Non-commutative geometry

Another approach (arguably enjoying stronger theoretical motivation compared to the ones discussed previously) which naturally introduces a minimum length scale occurs within NCGs. The term NCG encompasses a rather large ensemble of approaches: while we shall here shall use it in the sense of a space-time with non-zero commutation relations among the position coordinates [1286], of enormous importance is the spectral approach to NCG pioneered by Connes [1287]. The non-commutativity of space-time coordinates naturally encapsulates the discrete nature of space-time expected within several high-energy frameworks, including attempts towards constructing a theory of quantum gravity. For instance, this is a natural outcome of string theory, where target space-time coordinates become non-commuting operators on a *D*-brane [1288]. It is worth noting, however, that motivation for NCG covers a wide range of frameworks going well beyond string theory.

One of the earliest approaches towards constructing a non-commutative field theory was based on the Weyl–Wigner–Moyal \*-product [1289–1293]. However, this approach left open several important questions, concerning among others the issues of Lorentz invariance and UV finiteness. This prompted the development of a different approach, the so-called coherent state formalism, wherein physical coordinates are commuting c-numbers, built as expectation values on coherent states [1294, 1295]: within this approach, one can construct an explicitly UV-finite non-commutative field theory (see also [1296]). The literature on NCG is extremely vast and it is not possible to do it justice here: for representative examples of important works on NCG and its applications to various scenarios of interest, as well as related ideas, we refer the reader to [1297–1346].

In the approach we shall follow, the non-commutativity of space-time will be encoded in the commutation relation between space-time coordinates:

$$[x^{\mu}, x^{\nu}] = i\vartheta^{\mu\nu}, \tag{133}$$

where  $\vartheta^{\mu\nu}$  is an anti-symmetric real matrix. Within the coherent state formalism, the issues of covariance and unitarity are solved if one assumes the following block-diagonal form for  $\vartheta^{\mu\nu}$ :

$$\vartheta^{\mu\nu} = \vartheta \operatorname{diag}\left(\epsilon_{ij}, \epsilon_{ij}, \dots \epsilon_{ij}\right), \tag{134}$$

<sup>&</sup>lt;sup>54</sup> Another modified uncertainty relation scenario with a minimal length and maximal momentum it could be interesting to consider has been proposed in [1284, 1285]. To the best of our knowledge, BH, WH, and naked singularity solutions within this model have not been derived. We defer a full study thereof to future work.

where  $\epsilon_{ii}$  is the two-dimensional Levi-Civita tensor, and  $\vartheta$  is a fundamental constant with dimensions of length squared, and reflects the fact that within NCG the best possible localization of a particle is within a fundamental cell of area  $\vartheta$ . It has been shown in [1294, 1295] that the main outcome of this approach is to replace point-like objects with smeared structures, a fact which can be seen explicitly by computing the NCG-induced modifications to the Feynman path integral. This basically amounts to a substitution rule, where Dirac delta functions are replaced by a distribution of 'typical' (in a loose sense) width  $\sqrt{\vartheta}$ . The same of course can and will occur for finite mass distributions, including the mass density of a particle-like gravitational source: rather than being localized at the origin, this will be smeared and diffuse throughout a region of typical width  $\sqrt{\vartheta}$ , as a result of the uncertainty encoded in the non-commutativity of the space-time coordinates in equations (133) and (134). Applying this 'substitution rule', [1347] was the first to study NCG-inspired BHs: we use the term 'inspired' since, in this approach, the choice of the function with which to replace the Dirac delta essentially amounts to a phenomenological, but theory-inspired, choice. In the following, we shall consider two different choices for such a function describing the smearing of the mass distribution of a point-like gravitational source: a Gaussian in section 3.39.1 and a Lorentzian in section 3.39.2.

3.39.1. Gaussian mass distribution. We begin by choosing a Gaussian mass distribution, as first considered in [1347]. We note that this is the choice which enjoys strongest theoretical motivation. In this case, the point-like mass distribution  $\rho(r) = M\delta(r)$  is replaced by the following:

$$\rho_{\vartheta}(r) = \frac{M}{(4\pi\vartheta)^{\frac{3}{2}}} \exp\left(-\frac{r^2}{4\vartheta}\right),\tag{135}$$

so that the point-like gravitational source diffuses throughout a region of size  $\sqrt{\vartheta}$ . Given this choice of mass distribution, the corresponding BH solution was studied in [1347] by constructing the (covariantly conserved) energy-momentum tensor corresponding to the energy density of equation (135). The final result is that the metric function describing NCG-inspired BHs within this scenario is given by [1347] (see also [1348–1362] for further works on NCG-inspired BHs):

$$A(r) = 1 - \frac{4M}{r\sqrt{\pi}}\gamma\left(\frac{3}{2}, \frac{r^2}{4\vartheta}\right),\tag{136}$$

where  $\vartheta$  characterizes an universal hair,  $\gamma$  denotes the lower incomplete gamma function, and the Schwarzschild metric is recovered in the limit  $r/\sqrt{\vartheta} \to \infty$ . For  $\sqrt{\vartheta} < 0.5252$  the above space-time possesses two horizons, whereas an extremal BH with degenerate horizons at r = 1.5873 is recovered for  $\sqrt{\vartheta} = 0.5252$ . In the following, we shall focus on the  $\sqrt{\vartheta} \leq 0.5252$  regime.

We compute the shadow size numerically, and show its evolution as a function of the NCG parameter  $\vartheta$  in figure 45. We find that the shadow size depends extremely weakly on  $\vartheta$ , so much so that the evolution cannot be appreciated by the naked eye. This is consistent with the earlier findings of [1363]. We find that for the extremal value  $\sqrt{\vartheta} = 0.5252$ , the relative change in the shadow size compared to that of a Schwarzschild BH with the same mass is of order  $\mathcal{O}(10^{-3})$ . We therefore conclude that this particular NCG-inspired BH is in perfect agreement with the EHT observations, but distinguishing it from its Schwarzschild counterpart appears extremely unlikely, unless unrealistic angular resolutions of  $\mathcal{O}(nas)$  can be reached (which is extremely far from feasible).



**Figure 45.** Same as in figure 3 for the NCG-inspired BH (with Gaussian mass distribution) with metric function given by equation (136), as discussed in section 3.39.1.

3.39.2. Lorentzian mass distribution. Remaining in the realm of NCG-inspired phenomenology, another possibility which has been considered in the literature is that where a point-like gravitational source is smeared across a non-Gaussian distribution. An example which has been studied in this sense envisages a Lorentzian distribution (see e.g. [1364–1366]), so that equation (135) is modified to the following:

$$\rho_{\vartheta}(r) = \frac{M\sqrt{\vartheta}}{\pi^{\frac{3}{2}}(r^2 + \pi\vartheta)^2},\tag{137}$$

where again the source diffuses across a region of linear size  $\sqrt{\vartheta}$ . The resulting NCG-inspired BHs were studied in [1365], and shown to be described by the following metric function (to lowest order in  $\sqrt{\vartheta}$ ):

$$A(r) = 1 - \frac{2M}{r} + \frac{8M\sqrt{\vartheta}}{\sqrt{\pi}r^2},$$
(138)

where  $\vartheta$  characterizes an universal hair, and which is clearly of the RN-like form, so we can write down the following closed-form expression for the shadow radius:

$$r_{\rm sh} = \frac{\sqrt{2\sqrt{\pi}} \left(3 + \sqrt{9 - 64\sqrt{\frac{\theta}{\pi}}}\right)}{\sqrt{\frac{32\sqrt{\frac{\theta}{\pi}} + \sqrt{9 - 64\sqrt{\frac{\theta}{\pi}} - 3}}{8\sqrt{\theta}}}}.$$
(139)

We show the evolution of the shadow size as a function of the NCG parameter  $\vartheta$  in figure 46. Compared to the previous Gaussian mass distribution case, we now find a stronger effect on the BH shadow size, which decreases with increasing  $\vartheta$ . This is likely due to the fact that a Lorentzian distribution has heavier tails compared to a Gaussian one with the same value of  $\vartheta$ . We find that the EHT observations set the upper limits  $\vartheta \leq 0.02M^2$  ( $1\sigma$ ) and  $\vartheta \leq 0.04M^2$ ( $2\sigma$ ), both comparatively weak as one would typically expect NCG effects to show up at much smaller scales (see e.g. [1367]).



**Figure 46.** Same as in figure 3 for the NCG-inspired BH (with Lorentzian mass distribution) with metric function given by equation (138), as discussed in section 3.39.2.

### 3.40. Barrow entropy

Since the discovery of Hawking evaporation and the fact that a temperature can be associated to BHs [219, 1368], the intriguing connections between thermodynamics and gravity have flourished and given birth to the vibrant field of BH thermodynamics [1369–1371]. The Bekenstein–Hawking entropy relation envisages a simple proportionality relation between the entropy and area of a BH [1368, 1372], and is unquestionably the most studied entropy relation in the BH thermodynamics field. However, over the past decades various alternative entropy relations have been explored, with a wide range of different theoretical motivations, including the fact that the non-extensive nature of gravitational systems such as BHs calls for a non-extensive entropy-area relation (unlike the Bekenstein–Hawking entropy): see e.g. [1373–1382] for important examples thereof.

As first explicitly noted by one of us in [155], in the context of a BH the presence of a modified entropy-area relation effectively *back-reacts* on other thermodynamical quantities, including the BH's mass. If *M* is the BH 'bare' mass and S(A) = S(M) is the entropy-area/mass relation in question, the 'dressed' mass  $\widetilde{M}$  is then given by  $\widetilde{M} = (\partial S/\partial M)/8\pi$ . The necessity of constructing a consistent thermodynamical system in the presence of a modified entropy relation is indeed an issue which has started receiving significant attention recently, see e.g. discussions in [1383–1386], as well as [1387].

A particularly interesting modified entropy relation has been considered by Barrow [1388], motivated by possible space-time foam effects. Barrow entropy leads to a fractal structure for the horizon with a volume which is finite (but larger than in the standard case), and area which can be either finite or infinite. In the toy model considered by Barrow, a three-dimensional fractal is built around the BH event horizon, modifying its total surface and volume<sup>55</sup>. In this picture, the entropy-area law of a BH is modified to:

<sup>&</sup>lt;sup>55</sup> As Barrow himself stated, this modification was inspired by animations of the Covid-19 virus as a sphere with several attachments leaving off its surface to increase its surface area in order to increase the total number of possible latches to other cells.

$$S = \left(\frac{A}{4}\right)^{1+\frac{\Delta}{2}},\tag{140}$$

where  $\Delta$  characterizes an universal hair,  $0 \leq \Delta \leq 1$  is related to the dimension of the fractal, with  $\Delta = 0$  resulting in the usual Bekenstein–Hawking entropy S = A/4, and  $\Delta = 1$  leading the largest deviation of the BH entropy from the Bekenstein–Hawking value. Since its proposal, Barrow entropy has been at the center of several interesting applications, among which it is worth mentioning its use in driving a model of holographic dark energy [1389–1391] (see also [1392–1415]).

As shown in [155] by one of us, the back-reaction of Barrow entropy on the Schwarzschild metric modifies the 'bare' mass to  $\mathcal{M} = (\Delta/2 + 1)(4\pi)^{\Delta/2}M^{\Delta+1}$ , so that the metric function of the Barrow entropy-corrected BH is given by [155]:

$$A(r) = 1 - \frac{(\Delta + 2)M^{\Delta + 1}(4\pi)^{\Delta/2}}{r}.$$
(141)

It is therefore trivial to show that the shadow size is given by 56:

$$r_{\rm sh} = 3\sqrt{3}\mathcal{M} = 3\sqrt{3}\left(\frac{\Delta}{2} + 1\right)M^{1+\Delta}\pi^{\Delta/2}.$$
(142)

The shadow size increases with increasing  $\Delta$ , reflecting the fact that the BH surface area increases due to the quantum gravity effects envisaged in [1388].

We show the evolution of the shadow size as a function of  $\Delta$  in figure 47. With the same caveats already discussed previously for MOG in section 3.20, as well as GUP and EUP scenarios in section 3.38 (see also [1385] for a different point of view), we clearly see that the EHT observations set extremely strong constraints on the parameter  $\Delta$ . In particular, we find the upper limit  $\Delta \leq 0.001$  at  $1\sigma$ , tighter than cosmological constraints requiring  $\Delta \leq 0.1888$  [1416, 1417], but weaker than Big Bang Nucleosynthesis constraints requiring  $\Delta \leq 1.4 \times 10^{-4}$  [1418]. At  $2\sigma$ , we have the weaker constraint  $\Delta \leq 0.035$ , which is nonetheless still stronger than the cosmological one. One particularly interesting consequence of these limits is that the behavior of a holographic DE component based on Barrow entropy is prevented from deviating significantly from the cosmological constant  $\Lambda$ , particularly in the phantom regime, and for this reason we expect that Barrow entropy cannot play a significant role in the context of the Hubble tension [1419–1423]: see e.g. [1424–1438] for relevant discussions in relation to phantom dark energy. This highlights an interesting complementarity between BH shadows and cosmology, already hinted to in some of our earlier results, and which we plan to explore more in future works.

## 3.41. Other models and space-times

Although in this work we have covered a large number of scenarios, our limited resources meant that we were inevitably forced to make a choice as to which models to study, leaving out an even larger number of potentially interesting ones. Here, we very briefly comment on a few interesting models which have been left out.

3.41.1. Vector-tensor Horndeski gravity. While Horndeski's scalar-tensor theory [610] is probably among the best known modified gravity frameworks, much less known is the fact

<sup>&</sup>lt;sup>56</sup> Note that equation (27) in [155] by one of us features a spurious factor of  $2^{\Delta}$ , which we have corrected here.



**Figure 47.** Same as in figure 3 for the Barrow entropy-corrected BH with metric function given by equation (141), as discussed in section 3.40.

that Horndeski also wrote down the most general 4D vector-tensor theory with second order equations of motion [1439]. While not sharing the same success as its scalar-tensor counterpart, the theory has started re-gaining interest recently (see e.g. [1440–1459]). The theory allows for the coupling of vector fields to gravity through their field-strength tensors, and this can be done for *any* vector field, including the gauge fields of the Standard Model. With  $F^{\mu\nu}$  (possibly carrying additional gauge and/or family indices) denoting the field-strength tensor of a gauge field, the gauge-gravity coupling is of the form:

$$\mathcal{L} \supset \frac{\lambda}{4M_{\rm Pl}^2} \left( RF_{\mu\nu}F^{\mu\nu} - 4R_{\mu\nu}F^{\mu\alpha}F^{\nu}_{\alpha} + R_{\mu\nu\alpha\beta}F^{\mu\nu}F^{\alpha\beta} \right) \,, \tag{143}$$

with  $\lambda$  the dimensionless gauge-gravity coupling, which of course can take different values for each gauge field.

In [1460], one of us derived upper limits on  $\lambda$  by estimating the order-of-magnitude of the gauge-gravity coupling as  $\sim \lambda \mathcal{RF}^2/M_{\text{Pl}}^2$ , with  $\mathcal{R}$  and  $\mathcal{F}$  the typical curvature and field-strength tensor strengths respectively, and requiring that the Horndeski interaction term be weaker than both the Einstein–Hilbert and Maxwell Lagrangian terms, in order for the associated effects to be subdominant. This leads to upper limits of the order of:

$$|\lambda| \lesssim \left(\frac{\ell_{\mathcal{R}}}{\ell_{\text{Pl}}}\right)^2,\tag{144}$$

where  $\ell_{\mathcal{R}}$  is the length scale of the system under consideration. For instance, in the case of BH shadows one expects the correction to the shadow size due to the gauge-gravity interaction to be of the order of:

$$\frac{\delta r_{\rm sh}}{r_{\rm sh}} \sim \lambda \left(\frac{\ell_{\rm Pl}}{\ell_{\mathcal{R}}}\right)^2,\tag{145}$$

where  $\ell_R \sim M_{\rm bh}/M_{\rm Pl}^2$ , with  $M_{\rm bh}$  the BH mass. Requiring that the fractional correction to the shadow size be much smaller than unity results in the upper limit:

$$|\lambda| \lesssim \left(\frac{M_{\rm bh}}{M_{\rm Pl}}\right)^2. \tag{146}$$

For M87<sup>\*</sup>, equation (146) gives the limit  $|\lambda| \lesssim 10^{88}$ . Of course, with Sgr A<sup>\*</sup> we can improve this limit by about 6 orders of magnitude as a direct consequence of the fact that we are probing a regime of much higher curvature (see figure 1), obtaining  $|\lambda| \lesssim 10^{82}$ . This is still 7 order of magnitude short of the strongest constraint obtained in [1460] from neutron stars, and is overall weak in a broad sense. It is plausible that cosmological or GW constraints on this theory may yield much stronger bounds but such constraints, to the best of our knowledge, are not yet available.

3.41.2. Bumblebee gravity. Bumblebee gravity, first proposed by Kostelecký and Samuel in [761] (see also [1461, 1462]), is a class of effective field theories of gravity exhibiting spontaneous Lorentz symmetry breaking driven by the dynamics of a vector field acquiring a VEV. As such, the model constitutes an interesting playground for studying effects associated to spontaneous breaking of Lorentz symmetry, and has been the subject of a large number of studies (see e.g. [395, 1058, 1162, 1463–1487]). An exact non-rotating BH solution in bumblebee gravity was constructed in [1488], and is described by a single additional parameter  $\ell$ , which is related to the VEV of the bumblebee field, with the full metric given by:

$$ds^{2} = -\left(1 - \frac{2M}{r}\right)dt^{2} + (1+\ell)\frac{dr^{2}}{1 - \frac{2M}{r}} + r^{2}d\Omega^{2},$$
(147)

However, in this metric the only modifications to the Schwarzschild space-time appear in the  $g_{rr}$  term, and do not therefore change the size of the shadow, which remains  $3\sqrt{3}M$ . Therefore, for the purposes of the metric test presented in this work, BHs in bumblebee gravity are not distinguishable from a Schwarzschild BH with the same mass. A rotating version of this solution was derived in [1489] (see also [1490]): for what concerns the shadow size computation, this BH has the same form of the Kerr metric, with an effective spin given by  $a_*\sqrt{1+\ell}$ , and therefore can only be sensibly constrained if one has an independent spin measurement. We note, however, that the correctness of this BH solution was criticized in [1491] (see also [1492, 1493]).

3.41.3. Dynamical Chern–Simons gravity. Dynamical Chern–Simons gravity modifies the Einstein–Hilbert action by adding a scalar field  $\vartheta$  with a standard kinetic term and a potential, and coupling this field to the Pontryagin density obtained by contracting the Riemann tensor with its dual, as follows [1494, 1495]:

$$\mathcal{L} \supset \frac{\alpha}{4} \vartheta R_{\mu\nu\rho\sigma}^{*} R^{\mu\nu\rho\sigma}, \qquad (148)$$

where the dual of the Riemann tensor is given by:

$${}^{*}R^{\mu\nu\rho\sigma} = \frac{1}{2}\epsilon^{\rho\sigma\alpha\beta}R^{\mu\nu}_{\alpha\beta}.$$
(149)

Such a coupling enjoys strong motivation from various theoretical scenarios, including string theory, LQG, effective field theories of inflation, and so on, and has been extensively studied in the literature [1496–1510].

As shown in [1494], non-rotating BHs in dynamical Chern–Simons gravity are still described by the Schwarzschild space-time. Physically speaking, this is a consequence of the fact that the stationary Cotton tensor vanishes on a spherically symmetric background. Therefore, at the level of non-rotating solutions as considered in the present work, the EHT observations cannot be used to test dynamical Chern–Simons gravity. Rotating BH solutions in this theory have been considered in various works. While exact solutions valid for all values of the spin are not known, a number of slowly-rotating solutions have been derived in the literature [1511–1515].

3.41.4. Casimir wormhole. The Casimir effect, i.e. the force exerted between two plane parallel, close, uncharged, metallic plates in vacuum, was predicted theoretically almost a century ago [1516], and later verified experimentally. The associated Casimir energy has long been known to be a potential candidate source to generate and support a traversable WH (see e.g. [369, 1517], as well as [1518–1527] for later works). In [1528], Garattini explicitly constructed various interesting traversable WH solutions supported by Casimir energy. One example is described by the following metric:

$$ds^{2} = -\left(\frac{3r}{3r+a}\right)^{2} dt^{2} + \frac{dr^{2}}{1-\frac{2a}{3r}+\frac{a^{2}}{3r^{2}}} + r^{2} d\Omega^{2},$$
(150)

where *a* is the throat radius. From the above we can immediately read off that the WH's ADM mass is M = 3a, so that the metric function can be rewritten as:

$$A(r) = \left(\frac{r}{r+M}\right)^2.$$
(151)

It is easy to see that the above metric function is identical to the metric function of the null NS considered in section 3.15. The key difference between the two, rendering one a WH and the other a NS, lies in the  $g_{rr}$  metric element. However, this does not impact the computation of the photon sphere and associated shadow size. Just as for the null NS, we find that the Casimir WH does not have a photon sphere (formally the solution to the equation determining the photon sphere is  $r_{ph} = 0$ ). We would nonetheless expect it to cast a shadow of size  $r_{sh} = M$ , which is immediately ruled out at very high significance by the EHT observations. However, a definitive answer as to whether or not the Casimir WH casts a shadow would require a light trajectory analysis similar to the one performed by some of us in [74] for the null NS (see figure 1(f) thereof), given the presence of the WH throat at r = M/3, which may prevent the results of [74] from fully carrying over here. We defer this analysis to a future work.

3.41.5. *T*-duality. In string theory, T-duality is an important symmetry relating the physical properties of two otherwise unrelated backgrounds (loosely speaking, one at large radii and the other one at small radii), which however behave identically as far as physical observables are concerned [1529]. However, T-duality is not an exclusive property of string theory, and has been observed to emerge quite generally in a number of other systems [1530], wherein the path integral is invariant under exchange of the infinitesimal path length  $Dx \rightarrow 1/Dx$ . Recently, an important application of T-duality consisted in the formulation of a finite electrodynamics theory driven by a T-dual photon propagator [1531, 1532], which naturally incorporates a minimum length  $\ell_0$ . String-inspired T-duality corrections to BH space-times have been studied in [1533], and subsequently in [1534] by one of us. There, it was found that the resulting BH solution is identical to the Bardeen regular BH considered in section 3.2, with the minimum

length  $\ell_0$  playing the role of Bardeen charge  $q_m$  in equation (20). Therefore, we can automatically translate the results obtained in section 3.2 to the T-duality scenario, finding that the EHT observations are unable to place meaningful constraints on the minimum length  $\ell_0$ , at least based on the metric test considered here.

# 4. Brief discussion

Having tested a wide variety of models and fundamental physics scenarios, our goal here is to provide a very brief bird's-eye overall discussion of our results, while also providing an outlook on complementary constraints and future prospects. We note that a much more complete discussion, at least as far as complementary constraints from additional probes is concerned, is provided in section 6 of [20], to which we refer the reader who wishes to examine the topic more in depth. Several of the scenarios we have considered featured dimensionful parameters, either tied to fundamental Lagrangian parameters, appearing as integration constants (possibly tied to conserved charges/hair parameters), or introduced on phenomenological grounds. For these parameters, the constraints we have obtained (upper or lower limits) are typically of order  $\mathcal{O}(M)$ , with M Sgr A\*'s mass. This is not unexpected, given the fact that BH shadow imaging probes horizon scales of order M, that the EHT resolution is comparable to Sgr A\*'s gravitational radius, and also given the results of related earlier works based on M87\*'s shadow (whose constraints we have nonetheless improved typically by at least 3 orders of magnitude). Where available, we have reported constraints on the same scenarios based on complementary probes such as cosmology, GWs, and x-ray reflection spectroscopy to mention a few: typically, these constraints are much stronger than the ones we have obtained from the shadow of Sgr A<sup>\*</sup>.

The above points clearly raise the question of whether there is value in performing shadowbased tests of fundamental physics, if the resulting limits are so weak. We believe the answer is a resounding yes! Although in figure 1 we have only shown a small region of the Baker–Psaltis– Skordis potential-curvature plot, it is clear that BH shadows probe a regime which cannot otherwise be tested by other observations: therefore, regardless of the weakness of the resulting limits, verifying that GR holds (or at the very least is not in strong tension with observations) in this region of parameter space *as well as within other widely distant regimes* provides a very non-trivial consistency check/null test of the theory. This point was strongly emphasized in [20], to which we refer the reader for further discussions (see in particular figure 23 thereof). Nonetheless, we have also noted that for the case of dimensionless parameters we have been able to obtain comparatively strong constraints, in some cases competitive with or even stronger than cosmological ones, particularly within scenarios where the extra parameter makes the BH shadow larger.

The above discussion, however, raises another question: can one truly straightforwardly compare constraints based on BH shadows to those obtained from these other probes? This is not a simple question. If one has a specific model in mind, then this is in principle possible, provided one is able to exactly compute the signature expected in these complementary probes, within a given model, from first principles. However, this is not always simple: in fact, in most cases, this is not even possible. To model cosmological observations, leaving aside the spiny issue of non-linearities, one needs to solve the full Einstein–Boltzmann system, whose derivation is in general not simple for an arbitrary fundamental model<sup>57</sup>. For GWs, if the field

<sup>&</sup>lt;sup>57</sup> In fact, one often works with parametrizations which capture the effect of new physics in a model-agnostic way, but are not necessarily tied to any fundamental model.

**Table 2.** Summary of our results. The first four columns indicate respectively the model's name ('BH', 'WH', and 'NS' stand for 'black hole', 'wormhole', and 'naked singularity'), the corresponding section, the main references for the corresponding solution (these are not necessarily the references where the underlying model was first proposed or discussed), and the equation(s) where the metric function(s) are presented, alongside further restrictions on the values of the hair parameter(s). The final column summarizes the  $1\sigma$  and  $2\sigma$  constraints on the hair parameters: 'unconstrained (consistent)' indicates that the parameter is unconstrained because for all parameter values the shadow size is consistent with the EHT observations (for AS gravity this occurs only for the  $2\sigma$  constraints), whereas 'excluded' indicates that the model is excluded for all parameter values. The final five rows are separated from the previous ones and formatted differently to indicate that these models were only briefly discussed in section 3.41.

Model	Section	References	A(r), C(r)	Constraints $(1\sigma, 2\sigma)$
Reissner–Nordström BH	Section 3.1	[319–322]	Equation (18), $O/M \leq 1$	$Q \lesssim 0.8M, 0.95M$
Reissner–Nordström NS	Section 3.1	[319–322]	Equation (18), $1 < Q/M \leq \sqrt{Q/R}$	Excluded
Bardeen BH	Section 3.2	[330]	Equation (20)	$q_{\rm m}$ unconstrained
Hayward BH	Section 3.3	[335]	Equation (21)	$\ell$ unconstrained (consistent)
Frolov BH	Section 3.4	[341]	Equation (22), $\ell = 0.3$	$q \lesssim 0.8M, 0.9M$
Einstein–Bronnikov BH	Section 3.5	[349]	Equation (26)	$q_{ m m} \lesssim 0.8 M, M$
GCSV BH	Section 3.6	[351-353]	Equation $(27)$	$g \leq 0.8M, M$
Kazakov–Solodukhin BH	Section 3.7	[365]	Equation (28)	$\ell \lesssim 0.2M, M$
Ghosh–Kumar BH	Section 3.8	[367]	Equation (30)	$k \lesssim 1.4M, 1.6M$
Simpson–Visser BH	Section 3.9	[417]	Equation (33), $a/M < 2$	<i>a</i> unconstrained (consistent)
Simpson–Visser WH	Section 3.9	[417]	Equation (33), $2 \leq a/M < 3$	<i>a</i> unconstrained (consistent)
Morris–Thorne WH	Section 3.10	[454]	Equations (35) and (36)	Excluded
Damour–Solodukhin WH	Section 3.11	[461]	Equation (37)	$\lambda \lesssim 0.4, 0.5$
Janis–Newman– Winicour NS	Section 3.12	[506]	Equation (39)	$\overline{\nu} \lesssim 0.4M, 0.45M$
Joshi–Malafarina– Narayan-1 NS	Section 3.13	[520]	Equation (41), $2/3 \leq \chi < 1$	$\chi$ unconstrained (consistent)
Thin shell matter NS	Section 3.14	[526]	Equations (43) and (44), $\chi_e = 0.6$ , $R_{b2} \approx 0.33$	$R_{b1}\gtrsim 1.5M, 2M$
Null NS	Section 3.15	[74]	Equation (46)	Excluded
Minimally coupled scalar with potential	Section 3.16	[128, 536, 537]	Equation (50)	$\nu \lesssim 0.01 M, 0.4 M$
BH with conformal scalar hair	Section 3.17	[128, 538]	Equation (53), $0 \leq S < M^2$	$S \lesssim 0.65M^2,$ $0.9M^2$
WH with conformal scalar hair	Section 3.17	[128, 538]	Equation (53), $S < 0$	$S \gtrsim -0.04M^2, \\ -0.4M^2$
Clifton–Barrow $f(R)$ gravity	Section 3.18	[609]	Equation (56)	$\delta \gtrsim -0.15, -0.20$
BH in Horndeski gravity (case 1)	Section 3.19.1	[644]	Equation (64), $-M^2/8\pi < p_{\text{eff}} \leq 0$	$p_{\rm eff} \gtrsim -0.005 M^2, -0.01 M^2$

(Continued.)

Model	Section	References	A(r), C(r)	Constraints $(1\sigma, 2\sigma)$
WH in Horndeski	Section 3.19.1	[644]	Equation (64), $n \approx 20$	$p_{\rm eff} \lesssim 0.002 M^2 \cdot 0.01 M^2$
BH in Horndeski gravity (case 2)	Section 3.19.2	[645]	Equation (68)	$-0.01, -0.15 \leq h$ $\leq 0.40, 0.60$
Modified Gravity	Section 3 20	[682]	Equation $(72)$	$\alpha \leq 0.01, 0.1$
Brane-world BH	Section 3.21	[728]	Equation (72) Equation (75), $a \ge 0$	$q \lesssim 0.15, 0.2$
Brane-world WH	Section 3.21	[728]	Equation (75), $q < 0$	$q\gtrsim -0.01, -0.1$
Einstein–Euler– Heisenberg BH	Section 3.22	[104]	Equation (78), $\mu = 0.3, q_{\rm m} < 1$	$q_{ m m} \lesssim 0.7 M, 0.8 M$
Sen BH	Section 3.23	[744]	Equation (82), $a_{\rm m} \leq \sqrt{2}$	$q_{\rm m} \lesssim 0.6M, 0.75M$
Kalb–Ramond BH	Section 3.24	[760]	Equation (84), $\Gamma = 0.5$	$k \lesssim 1.2, 1.5$
Einstein–Maxwell- dilaton-1 BH	Section 3.25	[745, 746, 770]	Equation (86), $q \leq \sqrt{2}M$	$q \lesssim 0.8M, M$
Einstein–æther type 1 BH	Section 3.26	[784, 785]	Equation (91)	$c_{13} \lesssim 0.1, 0.75$
Einstein-æther type 2 BH	Section 3.26	[784, 785]	Equation (92), $c_{13} = 0.99$	$1.3 \lesssim c_{14} \lesssim 1.4$
4D Einstein–Gauss– Bonnet gravity	Section 3.27	[848]	Equation (95), $\alpha \leq M^2/16\pi$	$\alpha$ unconstrained (consistent)
Asymptotically safe gravity (IR limit)	Section 3.28	[930]	Equation (98), $\widetilde{\omega} < M^2$	$\widetilde{\omega} \lesssim 0.9 M^2$ , unconstrained (consistent)
Rastall gravity	Section 3.29	[972]	Equation (100), $N_{\rm s} = 0.005$ ,	$\kappa$ unconstrained (consistent)
Loop quantum	Section 3.30	[1014]	Equation (101), $a_0 = 0$	$P \lesssim 0.05, 0.08$
Kottler BH	Section 3.31	[213]	Equation (102), $0 < \Lambda < \sqrt{3M}$	$\log_{10}(\Lambda[m^{-2}] \lesssim -41.0, -40.5$
Electromagnetic- Weyl	Section 3.32	[1052]	Equation (107)	$lpha \lesssim 0.01 M^2, \ 0.2 M^2$
coupling (PPL)				
Electromagnetic- Weyl	Section 3.32	[1052]	Equation (107)	$lpha \lesssim 0.25 M^2, \ 0.3 M^2$
coupling (PPM)				
DST BH Rindler gravity	Section 3.33 Section 3.34	[1065] [1076]	Equation (111) Equation (115)	$\sigma \lesssim 0.04, 0.05$ $\log_{10}(a[\mathrm{pc}^{-1}]) \lesssim$
BH surrounded by	Section 3.35	[1120]	Equation (117)	$k \lesssim 0.07M, 0.15M$
BH with a topological defect	Section 3.36	[1150, 1151]	Equation (119)	$k \lesssim 0.005, 0.1$
Hairy BH from gravitational	Section 3.37	[1220]	Equation (124), $\alpha = 1.0$	$\ell \gtrsim 0.35M, 0.15M$
GUP BH	Section 3.38.1	[1260–1262]	Equation (129)	$\beta \gtrsim 0.01 M^2 = 0.14 M^2$
EUP BH	Section 3.38.2	[1281]	Equation (132)	$\log_{10}(L_{\star}[\text{pc}]/\sqrt{\alpha}) \\ \geq -5, -4$
Non-commutative geometry (Gaussian)	Section 3.39.1	[1347]	Equation (136), $\vartheta \leq 0.5252$	$\vartheta$ unconstrained (consistent)
Non-commutative geometry	Section 3.39.2	[1365]	Equation (138)	$\vartheta \lesssim 0.02 M^2,$ $0.04 M^2$
Barrow entropy	Section 3.40	[155]	Equation (141)	$\Delta \lesssim 0.001, 0.035$

(Continued.)

Model	Section	References	A(r), C(r)	Constraints $(1\sigma, 2\sigma)$
Vector-tensor Horndeski gravity	Section 3.41.1	[1439, 1460]	_	$ \lambda  \lesssim 10^{82}$ ?
Bumblebee gravity	Section 3.41.2	[1489–1491]	_	$\ell$ unconstrained (consistent)?
Dynamical Chern–Simons gravity	Section 3.41.3	[1494]	—	$\alpha$ unconstrained (consistent)?
Casimir WH	Section 3.41.4	[1528]	Equation (151)	Excluded?
String-inspired T-duality-corrected BH	Section 3.41.5	[1534]	_	$\ell_0$ unconstrained (consistent)?

Table 2.	(Continued.)
----------	--------------

equations cannot be solved exactly, in general one can only obtain conservative limits on new physics provided the BH metric is known. Finally, for x-ray observations, one needs to known the corresponding rotating BH solution, which again is not always known or even easy to obtain.

Even if we leave aside the above problems, there remains the issue that several tests of gravity are not based on specific models, but rather on model-agnostic parametrized tests, each based on different frameworks and optimal only for the particular systems studied: among the many examples, we mention the PPN expansion [1535], post-Keplerian parametrizations [1536], parametrized post-Einstein frameworks [1537], effective-one-body frameworks [1538], post-Newtonian frameworks [1539], and so on. The differences between these types of tests do not allow for a simple cross-comparison (if such a comparison is possible at all) with the results obtained from BH shadows, particularly without reference to any specific fundamental model. However, as emphasized in section 6 of [144] (to which we once more encourage the reader to refer to), the wide range of assumptions and conditions involved allow at the very least for important null tests/consistency checks of GR. In fact, if no deviation from GR is observed in *any* of these probes, barring extremely unlikely fine-tuning or fortuitous cancellations, large deviations from the Kerr-Newman family of metrics are very hard to accommodate. Of course, going from such a consistency check to constraints on specific theories remains difficult, but there is no doubt that such consistency checks are highly valuable in their own right.

A related concern was raised in [58], where it was pointed out that the  $\mathcal{O}(M)$  upper limits on fundamental physics obtained from BH shadows, when extrapolated to the potential-curvature regime probed by GW observations, would allow for extremely large deviations in the latter, which are not observed. This is simply another way of stating the fact that GW constraints are in principle much tighter than BH shadows-based ones. However, most viable models beyond GR (including many of the ones we have considered) are equipped with screening mechanisms which allow them to pass Solar System tests of gravity [254, 1540, 1541]. The computation of the expected GW signal in the presence of such screening mechanisms can be significantly more complicated, and is an issue which has not yet been addressed in full generality (see e.g. [1542–1546] for important works).

Still along the above line, it is also worth noting that GWs are sensitive to the dynamics of the theory, and in particular to the propagating gravitational modes, whereas BH shadows provide a static test of *photons* propagating along the BH background (see e.g. [1547–1549] for important discussions in this sense). This distinction can become particularly important in the presence of screening mechanisms, and further hinders a direct, straightforward comparison

between constraints obtained from BH shadows and GWs or, for that matter, other dynamical probes of gravity in the strong-field regime, particularly if based on model-agnostic parametrizations which do not necessarily (if at all) take the effect of screening into account. These difficulties notwithstanding, a more detailed exploration of these points, and more generally of how to cross-compare and combine BH shadow tests with complementary tests, is clearly warranted, and we certainly plan to return to these issues in future works.

In closing, we briefly comment on how the constraints we have obtained can be improved by future probes. We expect improvements at the very least on two fronts: (*i*) improvements in the precision of the (already exquisitely precise) mass-to-distance ratio priors for Sgr A<sup>\*</sup>, and (*ii*) improvements in the angular resolution achievable by VLBI arrays. Regarding the first point, significant improvements can be expected from GRAVITY+, the upgrades to GRAVITY and the VLTI instrument, featuring wide-separation fringe tracking, new adaptive optics, and laser guide stars. These will enable all-sky, high contrast,  $\mu$ as interferometry, which can significantly improve the precision to which sources orbiting around Sgr A<sup>\*</sup> can be tracked, thereby improving the determination of Sgr A<sup>\*</sup>'s mass. The first phase of GRAVITY+, named GRAVITY Wide, has just begun, and we can therefore expect more precise measurements in the near future [1550]. Further improvements can also be expected from MATISSE, a second generation mid-infrared imaging spectro-interferometer for VLTI, which has recently seen first light [1551].

On the VLBI side, significant improvements can be expected from the next-generation EHT (ngEHT), the planned upgrade to the EHT array, which will double the number of array antennas and incorporate dual-frequency receivers, while more than doubling the sensitivity and dynamical range compared to the current EHT capabilities [1552]. The ngEHT is expected to begin the commissioning phase no earlier than 2027, so improvements in this sense may be expected in the next decade. Even more futuristic upgrades to the EHT involving space-based telescopes can further improve the achievable angular resolution by more than an order of magnitude, potentially reaching  $O(\mu as)$  resolutions [1553]. Looking even further into the future (most likely no earlier than 2060), x-ray interferometry (XRI) through constellations of satellites is an interesting possibility which may achieve sub- $\mu as$  angular resolution [1554]. However, XRI is best suited for imaging SMBHs with optically thick disks, and the corresponding images would therefore not probe the photon sphere, but the inner edge of the accretion disk, whose location however is expected to depend more strongly on the BH spin, making it absolutely necessary to move beyond the spherically symmetric space-times considered here.

We expect that the improvements on the mass measurements and VLBI sides discussed above will allow for substantial improvements in the limits we have obtained, which should be strengthened by at least a couple of orders of magnitude. Of course, by the time these limits are obtained, we expect complementary limits (e.g. from GWs and cosmology) to have become tighter as well. Additional improvements on the BH shadow side can be achieved if a robust measurement of the spin of Sgr A\* (or, for that matter, of other BHs being imaged) can be obtained. In this case, the use of techniques based on orbital angular momentum and twisted light in the vicinity of the BH appears promising, although for Sgr A\* the use of these techniques may be challenging due to its very short dynamical timescale [85, 1555, 1556]. In closing, we further note that VLBI images allow for a wide variety of tests of fundamental physics beyond those we have considered here, including but not limited to tests of the equivalence principle based on multi-frequency images [105], to probes of ultra-light particles and other aspects of fundamental physics based on shadow evolution [92, 150, 159, 169], spin measurements [1557], and polarimetric measurements [1558, 1559], and various tests of fundamental physics based on the BH photon ring, its auto-correlation, and other probes [40, 169,

1560–1564]. Studies on these and related possibilities are underway, and we expect to report on these in future works.

#### 5. Conclusions

Horizon-scale images of SMBHs and their shadows have opened a new unparalleled window onto tests of gravity and fundamental physics in the very strong-field regime, including the possibility that astrophysical BHs may be described by alternatives to the Kerr metric. In this work, we have used the horizon-scale images of Sgr A\* provided by the EHT [15–24] to test some of the most popular and well-motivated scenarios deviating from the Kerr metric<sup>58</sup>. Compared to horizon-scale images of M87\*, there are significant advantages in the use of images of Sgr A\*, as we discussed towards the end of section 1, with the most notable advantage being that Sgr A\*'s proximity to us results in an exquisite calibration of its mass-to-distance ratio, essential for connecting the angular size of its shadow to theoretical predictions within a given model. Our tests have been performed by connecting the angular size of the bright ring of emission to that of the underlying BH shadow, and utilizing prior information on Sgr A\*'s exquisitely calibrated mass-to-distance ratio: this is a robust and well-tested methodology, adopted for instance by the EHT collaboration themselves [20, 144]. We also note that our tests are parametric, and are complementary to theory-agnostic tests which can simultaneously constrain several theories, and which have been considered elsewhere (see e.g. [52, 194, 195]).

We have studied over 50 well-motivated scenarios, including both fundamental theoretical scenarios and more phenomenological ones, considering in particular (see table 2 for a summary of the models considered as well as the corresponding results): various regular BH space-times; string-inspired space-times; metrics arising within NLED theories; space-times violating the NHT due to the presence of additional fields (including but not limited to scalar fields); alternative gravity theories; fundamental frameworks leading to new ingredients such as modified commutation or entropy relations; and finally BH mimickers including various WH and NS solutions. For almost all of the space-times considered, we have found that the EHT observations set limits on fundamental dimensional parameters of order  $\mathcal{O}(0.1)$  times some power of mass M, which is not surprising considering that the EHT angular resolution is comparable to Sgr A\*'s gravitational radius, so constraints of this order should be expected. For dimensionless parameters, we have found constraints of the order or  $\mathcal{O}(0.1)$  or stronger (in some cases surpassing complementary constraints from cosmology). Overall, we have shown that the EHT horizon-scale images of Sgr A\* place particularly tight constraints on fundamental physics scenarios which predict shadows *larger* than that of a Schwarzschild BH of the same mass, as the EHT observations on average prefer a shadow size which is slightly smaller than the latter, given the Keck and VLTI priors on the mass-to-distance ratio (this had also been noted in [20]). Finally, we have shown that the metric tests considered in this work do not exclude the extremely intriguing possibility of Sgr A\* being a BH mimicker, in the context of various WH and NS space-times we have studied.

There is certainly ample opportunity for further work in this very promising direction. First of all, for obvious reasons we were forced to make a selection when considering which fundamental scenarios to study, and many interesting models were inevitably left out: we plan to return to these in future works. These scenarios could include viable models for the DM

<sup>&</sup>lt;sup>58</sup> See [1565] for a similar set of tests making use of flare data from Sgr A\*.

and dark energy permeating the Universe, as well as well-motivated alternatives to  $GR^{59}$ : in this sense, a detailed investigation of the synergy between BH shadows, GWs, cosmological and astrophysical probes, and laboratory tests of gravity and fundamental physics is certainly a direction worth pursuing (see e.g. [191]), as one can expect a very strong complementarity across these probes, although with all the caveats discussed in section 4. A future study could also include the effects of rotation on all the space-times considered. In closing, BH shadows have opened an unprecedented window onto gravity and fundamental physics in the era of multi-messenger astrophysics, and we have only just begun exploring their potential (see e.g. [1581–1630] for other works along similar lines, first appearing or updated around the same time or after our preprint had been posted to arXiv).

# Data availability statement

No new data were created or analyzed in this study.

### Acknowledgments

We thank Richard Brito, Yifan Chen, and Naoki Tsukamoto for useful discussions. S V was partially supported by the Isaac Newton Trust and the Kavli Foundation through a Newton-Kavli Fellowship, and by a grant from the Foundation Blanceflor Boncompagni Ludovisi, née Bildt. R R is supported by the Shanghai Government Scholarship (SGS). Y-D T is supported by U.S. National Science Foundation (NSF) Theoretical Physics Program, Grant PHY-1915005, and also in part by the NSF under Grant No. NSF PHY-1748958. M A is supported by a DST-INSPIRE Fellowship, Department of Science and Technology, Government of India. S G G is supported by SERB-DST through Project No. CRG/2021/005771. R K W acknowledges support from the United States-India Educational Foundation (USIEF) through a Fulbright-Nehru Postdoctoral Research Fellowship (Award 2847/FNPDR/2022), and from the University of KwaZulu-Natal and Prof. Sunil Maharaj through a postdoctoral fellowship of the National Research Foundation of South Africa (NRF). A Ö thanks Prof. Durmus Ali Demir for hospitality at Sabancı University while this work was being conducted. A Ö acknowledges the contribution of the COST Action CA18108—Quantum gravity phenomenology in the multimessenger approach (QG-MM). C B is supported by the National Natural Science Foundation of China (NSFC) through Grant No. 11973019, the Natural Science Foundation of Shanghai through Grant No. 22ZR1403400, the Shanghai Municipal Education Commission through Grant No. 2019-01-07-00-07-E00035, and Fudan University through Grant No. JIH1512604.

### Note added

We had independently started this work before the EHT press conference announcing the new observations of Sgr A\* on 12 May 2022. Some of the models we tested (specifically the Reissner–Nordström, Bardeen, Hayward, Frolov, Janis–Newman–Winicour, Joshi– Malafarina–Narayan, Kazakov–Solodukhin, and Einstein–Maxwell-dilaton space-times) have

<sup>&</sup>lt;sup>59</sup> Concerning alternatives to GR, it could be particularly interesting to examine possible constraints on torsion and non-metricity, especially in light of the so-called 'geometrical trinity of gravity' [1566], which offers three complementary descriptions of gravity in terms of the effects of curvature, torsion, and non-metricity. Static spherically symmetric within teleparallel f(T) gravity and non-metricity (symmetric teleparallel) f(Q) gravity have been obtained in recent years, see e.g. [1567–1576]: we defer a dedicated study of torsion and non-metricity constraints from BH shadows to a follow-up work. Other solutions it would be interesting to study include those of [1577–1580].

also inevitably been tested by the EHT collaboration in [20]. However, in this work we have considered other models and scenarios beyond those considered by the EHT collaboration. We have verified that, for the overlapping models, both results agree, therefore providing a valuable and independent cross-check.

# **ORCID** iDs

Sunny Vagnozzi b https://orcid.org/0000-0002-7614-6677 Luca Visinelli b https://orcid.org/0000-0001-7958-8940 Parth Bambhaniya b https://orcid.org/0000-0001-8424-3357 Sushant G Ghosh b https://orcid.org/0000-0002-0835-3690 Kimet Jusufi b https://orcid.org/0000-0003-0527-4177 Rahul Kumar Walia b https://orcid.org/0000-0003-2471-1360 Cosimo Bambi b https://orcid.org/0000-0002-3180-9502

## References

- [1] Schwarzschild K 1916 Sitzungsber. Preuss. Akad. Wiss. Berlin (Math. Phys.) 1916 189
- [2] Penrose R 1965 Phys. Rev. Lett. 14 57
- [3] Bambi C 2018 Ann. Phys., Lpz. 530 1700430
- [4] Hawking S W 1976 Phys. Rev. D 14 2460
- [5] Barack L et al 2019 Class. Quantum Grav. 36 143001
- [6] Fish V L, Akiyama K, Bouman K L, Chael A A, Johnson M D, Doeleman S S, Blackburn L, Wardle J F C and Freeman W T (The Event Horizon Telescope Collaboration) 2016 Galaxies 4 54
- [7] Akiyama K et al (The Event Horizon Telescope Collaboration) 2019 Astrophys. J. Lett. 875 L1
- [8] Akiyama K et al (The Event Horizon Telescope Collaboration) 2019 Astrophys. J. Lett. 875 L2
- [9] Akiyama K et al (The Event Horizon Telescope Collaboration) 2019 Astrophys. J. Lett. 875 L3
- [10] Akiyama K et al (The Event Horizon Telescope Collaboration) 2019 Astrophys. J. Lett. 875 L4
- [11] Akiyama K et al (The Event Horizon Telescope Collaboration) 2019 Astrophys. J. Lett. 875 L5
- [12] Akiyama K et al (The Event Horizon Telescope Collaboration) 2019 Astrophys. J. Lett. 875 L6
- [13] Akiyama K *et al* (The Event Horizon Telescope Collaboration) 2021 Astrophys. J. Lett. **910** L12
  [14] Akiyama K *et al* (The Event Horizon Telescope Collaboration) 2021 Astrophys. J. Lett. **910** L13
- [14] Akiyama K et al (The Event Horizon Telescope Collaboration) 2021 Astrophys. J. Lett. 910 E13
   [15] Akiyama K et al (The Event Horizon Telescope Collaboration) 2022 Astrophys. J. Lett. 930 L12
- [16] Akiyama K *et al* (The Event Horizon Telescope Collaboration) 2022 Astrophys. J. Lett. **930** L12
- [17] Akiyama K *et al* (The Event Horizon Telescope Collaboration) 2022 Astrophys. J. Lett. **930** L14
- [18] Akiyama K *et al* (The Event Horizon Telescope Collaboration) 2022 Astrophys. J. Lett. **930** L15
- [19] Akiyama K *et al* (The Event Horizon Telescope Collaboration) 2022 Astrophys. J. Lett. **930** L16
- [20] Akiyama K *et al* (The Event Horizon Telescope Collaboration) 2022 Astrophys. J. Lett. **930** L17
- [21] Farah J et al (The Event Horizon Telescope Collaboration) 2022 Astrophys. J. Lett. 930 L18
- [22] Wielgus M et al (The Event Horizon Telescope Collaboration) 2022 Astrophys. J. Lett. 930 L19
- [23] Georgiev B *et al* (The Event Horizon Telescope Collaboration) 2022 Astrophys. J. Lett. **930** L20
- [24] Broderick A E *et al* (The Event Horizon Telescope Collaboration) 2022 Astrophys. J. Lett.
   930 L21
- [25] Synge J L 1966 Mon. Not. R. Astron. Soc. 131 463
- [26] Luminet J P 1979 Astron. Astrophys. 75 228
- [27] Virbhadra K S and Ellis G F R 2000 *Phys. Rev.* D **62** 084003
- [28] Falcke H, Melia F and Agol E 2000 Astrophys. J. Lett. 528 L13
- [29] Claudel C-M, Virbhadra K S and Ellis G F R 2001 J. Math. Phys. 42 818
- [30] Bozza V, Capozziello S, Iovane G and Scarpetta G 2001 Gen. Relativ. Gravit. 33 1535
- [31] Virbhadra K S and Ellis G F R 2002 Phys. Rev. D 65 103004
- [32] Bozza V 2002 Phys. Rev. D 66 103001
- [33] Takahashi R 2004 Astrophys. J. 611 996
- [34] Virbhadra K S 2009 Phys. Rev. D 79 083004

- [35] Abdujabbarov A A, Rezzolla L and Ahmedov B J 2015 Mon. Not. R. Astron. Soc. 454 2423
- [36] Tsupko O Y 2017 Phys. Rev. D 95 104058
- [37] Medeiros L, Psaltis D and Özel F 2020 Astrophys. J. 896 7
- [38] Farah J R, Pesce D W, Johnson M D and Blackburn L L 2020 Astrophys. J. 900 77
- [39] Wang M, Chen S and Jing J 2022 Commun. Theor. Phys. 74 097401
- [40] Tsupko O Y 2022 Phys. Rev. D 106 064033
- [41] Cunha P V P and Herdeiro C A R 2018 Gen. Relativ. Gravit. 50 42
- [42] Dokuchaev V I and Nazarova N O 2020 Usp. Fiz. Nauk 190 627
- [43] Perlick V and Tsupko O Y 2022 Phys. Rep. 947 1
- [44] Chen S, Jing J, Qian W-L and Wang B 2023 Sci. China Phys. Mech. Astron. 66 260401
- [45] Chowdhuri A, Ghosh S and Bhattacharyya A 2023 Front. Phys. 11 1113909
- [46] Narayan R, Johnson M D and Gammie C F 2019 Astrophys. J. Lett. 885 L33
- [47] Völkel S H, Barausse E, Franchini N and Broderick A E 2021 Class. Quantum Grav. 38 21LT01
- [48] Bronzwaer T and Falcke H 2021 Astrophys. J. 920 155
- [49] Lara G, Völkel S H and Barausse E 2021 Phys. Rev. D 104 124041
- [50] Ozel F, Psaltis D and Younsi Z 2022 Astrophys. J. 941 88
- [51] Younsi Z, Psaltis D and Özel F 2023 Astrophys. J. 942 47
- [52] Kocherlakota P and Rezzolla L 2022 Mon. Not. R. Astron. Soc. 513 1229
- [53] Gralla S E, Holz D E and Wald R M 2019 *Phys. Rev.* D 100 024018
- [54] Gralla S E and Lupsasca A 2020 Phys. Rev. D 101 044031
- [55] Gralla S E and Lupsasca A 2020 Phys. Rev. D 102 124003
- [56] Gralla S E, Lupsasca A and Marrone D P 2020 Phys. Rev. D 102 124004
- [57] Gralla S E 2021 Phys. Rev. D 103 024023
- [58] Glampedakis K and Pappas G 2021 Phys. Rev. D 104 L081503
- [59] Atamurotov F, Ahmedov B and Abdujabbarov A 2015 Phys. Rev. D 92 084005
- [60] Babar G Z, Babar A Z and Atamurotov F 2020 Eur. Phys. J. C 80 761
- [61] Atamurotov F, Jusufi K, Jamil M, Abdujabbarov A and Azreg-Aïnou M 2021 Phys. Rev. D 104 064053
- [62] Johannsen T, Broderick A E, Plewa P M, Chatzopoulos S, Doeleman S S, Eisenhauer F, Fish V L, Genzel R, Gerhard O and Johnson M D 2016 Phys. Rev. Lett. 116 031101
- [63] Psaltis D 2019 Gen. Relativ. Gravit. 51 137
- [64] Israel W 1967 Phys. Rev. 164 1776
- [65] Israel W 1968 Commun. Math. Phys. 8 245
- [66] Carter B 1968 Phys. Rev. 174 1559
- [67] Carter B 1971 Phys. Rev. Lett. 26 331
- [68] Robinson D C 1975 Phys. Rev. Lett. 34 905
- [69] Bambi C and Freese K 2009 Phys. Rev. D 79 043002
- [70] Ohgami T and Sakai N 2015 *Phys. Rev.* D **91** 124020
- [71] Shaikh R 2018 Phys. Rev. D 98 024044
- [72] Cardoso V and Pani P 2019 Living Rev. Relativ. 22 4
- [73] Bambi C, Freese K, Vagnozzi S and Visinelli L 2019 Phys. Rev. D 100 044057
- [74] Joshi A B, Dey D, Joshi P S and Bambhaniya P 2020 Phys. Rev. D 102 024022
- [75] Guo J-Q, Joshi P S, Narayan R and Zhang L 2021 Class. Quantum Grav. 38 035012
- [76] Visinelli L 2021 Int. J. Mod. Phys. D 30 2130006
- [77] Herdeiro C A R, Pombo A M, Radu E, Cunha P V P and Sanchis-Gual N 2021 J. Cosmol. Astropart. Phys. JCAP04(2021)051
- [78] Kumar S, Bambhaniya P and Joshi P S 2022 arXiv:2202.00588 [gr-qc]
- [79] Cardoso V and Duque F 2022 Phys. Rev. D 105 104023
- [80] Tahelyani D, Joshi A B, Dey D and Joshi P S 2022 Phys. Rev. D 106 044036
- [81] Atamurotov F, Abdujabbarov A and Ahmedov B 2013 Phys. Rev. D 88 064004
- [82] Giddings S B 2019 Universe 5 201
- [83] Held A, Gold R and Eichhorn A 2019 J. Cosmol. Astropart. Phys. JCAP06(2019)029
- [84] Wei S-W, Zou Y-C, Liu Y-X and Mann R B 2019 J. Cosmol. Astropart. Phys. JCAP08(2019)030
- [85] Tamburini F, Thidé B and Della Valle M 2020 Mon. Not. R. Astron. Soc. 492 L22
- [86] Konoplya R A, Posada C, Stuchlík Z and Zhidenko A 2019 Phys. Rev. D 100 044027
- [87] Shaikh R 2019 Phys. Rev. D 100 024028
- [88] Contreras E, Ramirez-Velasquez J M, Rincón A, Panotopoulos G and Bargueño P 2019 Eur. Phys. J. C 79 802

- [89] Bar N, Blum K, Lacroix T and Panci P 2019 J. Cosmol. Astropart. Phys. JCAP07(2019)045
- [90] Jusufi K, Jamil M, Salucci P, Zhu T and Haroon S 2019 Phys. Rev. D 100 044012
- [91] Vagnozzi S and Visinelli L 2019 Phys. Rev. D 100 024020
- [92] Roy R and Yajnik U A 2020 Phys. Lett. B 803 135284
- [93] Long F, Wang J, Chen S and Jing J 2019 J. High Energy Phys. JHEP10(2019)269
- [94] Zhu T, Wu Q, Jamil M and Jusufi K 2019 Phys. Rev. D 100 044055
- [95] Contreras E, Rincón A, Panotopoulos G, Bargueño P and Koch B 2020 Phys. Rev. D 101 064053
- [96] Qi J-Z and Zhang X 2020 Chin. Phys. C 44 055101
- [97] Neves J C S 2020 Eur. Phys. J. C 80 343
- [98] Javed W, Abbas J and Övgün A 2019 *Phys. Rev.* D **100** 044052
- [99] Tian S X and Zhu Z-H 2019 Phys. Rev. D 100 064011
- [100] Cunha P V P, Herdeiro C A R and Radu E 2019 Universe 5 220
- [101] Banerjee I, Chakraborty S and SenGupta S 2020 Phys. Rev. D 101 041301
- [102] Shaikh R and Joshi P S 2019 J. Cosmol. Astropart. Phys. JCAP10(2019)064
- [103] Kumar R, Ghosh S G and Wang A 2019 Phys. Rev. D 100 124024
- [104] Allahyari A, Khodadi M, Vagnozzi S and Mota D F 2020 J. Cosmol. Astropart. Phys. JCAP02(2020)003
- [105] Yan S-F, Li C, Xue L, Ren X, Cai Y-F, Easson D A, Yuan Y-F and Zhao H 2020 Phys. Rev. Res. 2 023164
- [106] Jusufi K 2020 Phys. Rev. D 101 084055
- [107] Rummel M and Burgess C P 2020 J. Cosmol. Astropart. Phys. JCAP05(2020)051
- [108] Kumar R, Ghosh S G and Wang A 2020 *Phys. Rev.* D **101** 104001
- [109] Li P-C, Guo M and Chen B 2020 Phys. Rev. D 101 084041
- [110] Narang A, Mohanty S and Kumar A 2020 arXiv:2002.12786 [gr-qc]
- [111] Liu C, Zhu T, Wu Q, Jusufi K, Jamil M, Azreg-Aïnou M and Wang A 2020 Phys. Rev. D 101 084001
  - Liu C, Zhu T, Wu Q, Jusufi K, Jamil M, Azreg-Aïnou M and Wang A 2021 *Phys. Rev.* D **103** 089902 (erratum)
- [112] Konoplya R A and Zinhailo A F 2020 Eur. Phys. J. C 80 1049
- [113] Guo M and Li P-C 2020 Eur. Phys. J. C 80 588
- [114] Pantig R C and Rodulfo E T 2020 Chin. J. Phys. 68 236
- [115] Wei S-W and Liu Y-X 2021 Eur. Phys. J. Plus 136 436
- [116] Kumar R and Ghosh S G 2020 J. Cosmol. Astropart. Phys. JCAP07(2020)053
- [117] Islam S U, Kumar R and Ghosh S G 2020 J. Cosmol. Astropart. Phys. JCAP09(2020)030
- [118] Chen C-Y 2020 J. Cosmol. Astropart. Phys. JCAP05(2020)040
- [119] Sau S, Banerjee I and SenGupta S 2020 Phys. Rev. D 102 064027
- [120] Jusufi K 2020 Phys. Rev. D 101 124063
- [121] Kumar R and Ghosh S G 2021 Class. Quantum Grav. 38 8
- [122] Chen S, Wang M and Jing J 2020 J. High Energy Phys. JHEP07(2020)054
- [123] Zeng X-X, Zhang H-Q and Zhang H 2020 Eur. Phys. J. C 80 872
- [124] Neves J C S 2020 Eur. Phys. J. C 80 717
- [125] Övgün A and Sakallı I 2020 Class. Quantum Grav. 37 225003
- [126] Badía J and Eiroa E F 2020 *Phys. Rev.* D **102** 024066
- [127] Jusufi K, Jamil M and Zhu T 2020 Eur. Phys. J. C 80 354
- [128] Khodadi M, Allahyari A, Vagnozzi S and Mota D F 2020 J. Cosmol. Astropart. Phys. JCAP09(2020)026
- [129] Belhaj A, Chakhchi L, El Moumni H, Khalloufi J and Masmar K 2020 Int. J. Mod. Phys. A 35 2050170
- [130] Jusufi K, Amir M, Ali M S and Maharaj S D 2020 Phys. Rev. D 102 064020
- [131] Kumar R, Kumar A and Ghosh S G 2020 Astrophys. J. 896 89
- [132] Jusufi K, Azreg-Aïnou M, Jamil M, Wei S-W, Wu Q and Wang A 2021 Phys. Rev. D 103 024013
- [133] Belhaj A, Benali M, El Balali A, El Hadri W, El Moumni H and Torrente-Lujan E 2021 Int. J. Mod. Phys. D 30 2150026
- [134] Kruglov S I 2020 Mod. Phys. Lett. A 35 2050291
- [135] Psaltis D et al (The Event Horizon Telescope Collaboration) 2020 Phys. Rev. Lett. 125 141104
- [136] Ghasemi-Nodehi M, Azreg-Ainou M, Jusufi K and Jamil M 2020 Phys. Rev. D 102 104032
- [137] Ghosh S G, Kumar R and Islam S U 2021 J. Cosmol. Astropart. Phys. JCAP03(2021)056
- [138] Khodadi M and Saridakis E N 2021 Phys. Dark Universe 32 100835

- [139] Lee B-H, Lee W and Myung Y S 2021 *Phys. Rev.* D 103 064026
- [140] Contreras E, Ovalle J and Casadio R 2021 Phys. Rev. D 103 044020
- [141] Shaikh R, Pal K, Pal K and Sarkar T 2021 Mon. Not. R. Astron. Soc. 506 1229
- [142] Afrin M, Kumar R and Ghosh S G 2021 Mon. Not. R. Astron. Soc. 504 5927
- [143] Addazi A, Capozziello S and Odintsov S 2021 Phys. Lett. B 816 136257
- [144] Kocherlakota P *et al* (The Event Horizon Telescope Collaboration) 2021 *Phys. Rev.* D **103** 104047
- [145] Shaikh R, Paul S, Banerjee P and Sarkar T 2022 Eur. Phys. J. C 82 696
- [146] Badía J and Eiroa E F 2021 Phys. Rev. D 104 084055
- [147] Wang H-M and Wei S-W 2022 Eur. Phys. J. Plus 137 571
- [148] Khodadi M, Lambiase G and Mota D F 2021 J. Cosmol. Astropart. Phys. JCAP09(2021)028
- [149] Cai X-C and Miao Y-G 2021 arXiv:2107.08352 [gr-qc]
- [150] Frion E, Giani L and Miranda T 2021 Open J. Astrophys. 4
- [151] Wei S-W and Zou Y-C 2021 arXiv:2108.02415 [gr-qc]
- [152] Rahaman F, Singh K N, Shaikh R, Manna T and Aktar S 2021 Class. Quantum Grav. 38 215007
- [153] Walia R K, Maharaj S D and Ghosh S G 2022 Eur. Phys. J. C 82 547
- [154] Afrin M and Ghosh S G 2022 Astrophys. J. 932 51
- [155] Jusufi K, Azreg-Aïnou M, Jamil M and Saridakis E N 2022 Universe 8 102
- [156] Çimdiker I, Demir D and Övgün A 2021 Phys. Dark Universe 34 100900
- [157] Pal K, Pal K, Shaikh R and Sarkar T 2022 Phys. Lett. B 829 137109
- [158] Li G-P and He K-J 2021 Eur. Phys. J. C 81 1018
- [159] Roy R, Vagnozzi S and Visinelli L 2022 Phys. Rev. D 105 083002
- [160] Afrin M and Ghosh S G 2022 Universe 8 52
- [161] Pantig R C and Övgün A 2022 Eur. Phys. J. C 82 391
- [162] He K-J, Tan S-C and Li G-P 2022 Eur. Phys. J. C 82 81
- [163] Jha S K and Rahaman A 2022 Eur. Phys. J. C 82 411
- [164] Meng Y, Kuang X-M and Tang Z-Y 2022 *Phys. Rev.* D **106** 064006
- [165] Daas J, Kuijpers K, Saueressig F, Wondrak M F and Falcke H 2022 arXiv:2204.08480 [gr-qc]
- [166] Rosa J a L and Rubiera-Garcia D 2022 Phys. Rev. D 106 084004
- [167] Bogush I, Gal'tsov D, Gyulchev G, Kobialko K, Nedkova P and Vetsov T 2022 Phys. Rev. D 106 024034
- [168] Chen C-Y, Chiang H-W and Tsao J-S 2022 Phys. Rev. D 106 044068
- [169] Chen Y, Roy R, Vagnozzi S and Visinelli L 2022 Phys. Rev. D 106 043021
- [170] Narzilloev B, Hussain I, Abdujabbarov A and Ahmedov B 2022 Eur. Phys. J. Plus 137 645
- [171] He A, Tao J, Wang P, Xue Y and Zhang L 2022 Eur. Phys. J. C 82 683
- [172] Zhu Q-H, Han Y-X and Huang Q-G 2023 Eur. Phys. J. C 83 88
- [173] Belhaj A, Belmahi H, Benali M, Oualaid M and Sedra M B 2022 arXiv:2206.00615 [gr-qc]
- [174] Zhang Z, Yan H, Guo M and Chen B 2023 Phys. Rev. D 107 024027
- [175] Hendi S H, Jafarzade K and Eslam Panah B 2023 J. Cosmol. Astropart. Phys. JCAP02(2023)022
- [176] Rayimbaev J, Pantig R C, Övgün A, Abdujabbarov A and Demir D 2022 arXiv:2206.06599 [gr-qc]
- [177] Tang Z-Y, Kuang X-M, Wang B and Qian W-L 2022 Sci. Bull. 67 2272
- [178] Karmakar R, Gogoi D J and Goswami U D 2022 Int. J. Mod. Phys. A 37 2250180
- [179] Hou Y, Zhang Z, Yan H, Guo M and Chen B 2022 Phys. Rev. D 106 064058
- [180] Yu C, Chen D and Gao C 2022 Nucl. Phys. B 983 115925
- [181] Li S, Mirzaev T, Abdujabbarov A A, Malafarina D, Ahmedov B and Han W-B 2022 Phys. Rev. D 106 084041
- [182] Lobos N J L S and Pantig R C 2022 Physics 4 1318
- [183] Psaltis D and Johannsen T 2011 J. Phys.: Conf. Ser. 283 012030
- [184] Johannsen T 2012 Adv. Astron. 2012 486750
- [185] Psaltis D, Wex N and Kramer M 2016 Astrophys. J. 818 121
- [186] Johannsen T 2016 Class. Quantum Grav. 33 113001
- [187] Goddi C et al 2016 Int. J. Mod. Phys. D 26 1730001
- [188] Gebhardt K, Adams J, Richstone D, Lauer T R, Faber S M, Gultekin K, Murphy J and Tremaine S 2011 Astrophys. J. 729 119
- [189] Walsh J L, Barth A J, Ho L C and Sarzi M 2013 Astrophys. J. 770 86
- [190] Vagnozzi S, Bambi C and Visinelli L 2020 Class. Quantum Grav. 37 087001
- [191] Baker T, Psaltis D and Skordis C 2015 Astrophys. J. 802 63

- [192] Johannsen T, Psaltis D, Gillessen S, Marrone D P, Ozel F, Doeleman S S and Fish V L 2012 Astrophys. J. 758 30
- [193] Gillessen S, Eisenhauer F, Trippe S, Alexander T, Genzel R, Martins F and Ott T 2009 Astrophys. J. 692 1075
- [194] Nampalliwar S and Saurabh K 2021 arXiv:2108.01190 [gr-qc]
- [195] Nampalliwar S 2021 55th Rencontres de Moriond on Gravitation (arXiv:2108.03573 [gr-qc])
- [196] de Martino I, della Monica R and de Laurentis M 2021 Phys. Rev. D 104 L101502
- [197] Jusufi K, Kumar S, Azreg-Aïnou M, Jamil M, Wu Q and Bambi C 2022 *Eur. Phys. J.* C 82 633
   [198] Benisty D and Davis A-C 2022 *Phys. Rev.* D 105 024052
- [199] Borka D, Jovanović V B, Capozziello S, Zakharov A F and Jovanović P 2021 Universe 7 407
- [200] Yuan G-W, Shen Z-Q, Tsai Y-L S, Yuan Q and Fan Y-Z 2022 Phys. Rev. D 106 103024
- [201] Do T et al 2019 Science 365 664
- [202] Abuter R et al GRAVITY 2020 Astron. Astrophys. 636 L5
- [203] Broderick A E, Fish V L, Doeleman S S and Loeb A 2009 Astrophys. J. 697 45
- [204] Kato Y, Miyoshi M, Takahashi R, Negoro H and Matsumoto R 2010 Mon. Not. R. Astron. Soc. **403** L74
- [205] Broderick A E, Fish V L, Doeleman S S and Loeb A 2011 Astrophys. J. 738 38
- [206] Huang L, Takahashi R and Shen Z-Q 2009 Astrophys. J. 706 960
- [207] Shcherbakov R V, Penna R F and McKinney J C 2012 Astrophys. J. 755 133
- [208] Fragione G and Loeb A 2020 Astrophys. J. Lett. 901 L32
- [209] Fragione G and Loeb A 2022 Astrophys. J. Lett. 932 L17
- [210] Vagnozzi S and Visinelli L 2022 Res. Notes AAS 6 106
- [211] Psaltis D 2008 Phys. Rev. D 77 064006
- [212] Kumar R, Singh B P, Ali M S and Ghosh S G 2021 Phys. Dark Universe 34 100881
- [213] Kottler F 1918 Ann. Phys., Lpz. 361 401
- [214] Novello M, De Lorenci V A, Salim J M and Klippert R 2000 Phys. Rev. D 61 045001
- [215] De Lorenci V A, Klippert R, Novello M and Salim J M 2000 Phys. Lett. B 482 134
- [216] Jacobson T 2007 Class. Quantum Grav. 24 5717
- [217] Newman E T and Janis A I 1965 J. Math. Phys. 6 915
- [218] Newman E T, Couch R, Chinnapared K, Exton A, Prakash A and Torrence R 1965 J. Math. Phys. 6918
- [219] Hawking S W 1975 Commun. Math. Phys. 43 199 Hawking S W 1976 Commun. Math. Phys. 46 206 (erratum)
- [220] Guth A H 1981 Phys. Rev. D 23 347
- [221] Linde A D 1982 Phys. Lett. B 108 389
- [222] Wetterich C 1988 Nucl. Phys. B 302 668
- [223] Ratra B and Peebles P J E 1988 Phys. Rev. D 37 3406
- [224] Caldwell R R, Dave R and Steinhardt P J 1998 Phys. Rev. Lett. 80 1582
- [225] Benaoum H B 2022 Universe 8 340
- [226] Arkani-Hamed N, Finkbeiner D P, Slatyer T R and Weiner N 2009 Phys. Rev. D 79 015014
- [227] Cai Y-F, Saridakis E N, Setare M R and Xia J-Q 2010 Phys. Rep. 493 1
- [228] Cyr-Racine F-Y and Sigurdson K 2013 Phys. Rev. D 87 103515
- [229] Cline J M, Scott P, Kainulainen K and Weniger C 2013 Phys. Rev. D 88 055025 Cline J M, Kainulainen K, Scott P and Weniger C 2015 Phys. Rev. D 92 039906 (erratum)
- [230] Petraki K, Pearce L and Kusenko A 2014 J. Cosmol. Astropart. Phys. JCAP07(2014)039
- [231] Rinaldi M 2015 Class. Quantum Grav. 32 045002
- [232] Foot R and Vagnozzi S 2015 Phys. Rev. D 91 023512
- [233] Foot R and Vagnozzi S 2015 Phys. Lett. B 748 61
- [234] Rinaldi M 2015 J. Cosmol. Astropart. Phys. JCAP10(2015)023
- [235] Cai Y-F, Capozziello S, De Laurentis M and Saridakis E N 2016 Rep. Prog. Phys. 79 106901
- [236] Solà J, Gómez-Valent A and de Cruz Pérez J 2017 Astrophys. J. 836 43
- [237] Nojiri S, Odintsov S D and Oikonomou V K 2017 Phys. Rep. 692 1
- [238] Capozziello S, D'Agostino R and Luongo O 2018 Phys. Dark Universe 20 1
- [239] Benisty D and Guendelman E I 2018 Phys. Rev. D 98 023506
- [240] Poulin V, Smith T L, Karwal T and Kamionkowski M 2019 Phys. Rev. Lett. 122 221301
- [241] Heckman J J, Lawrie C, Lin L, Sakstein J and Zoccarato G 2019 Fortschr. Phys. 67 1900071
- [242] Niedermann F and Sloth M S 2021 Phys. Rev. D 103 L041303
- [243] Sakstein J and Trodden M 2020 Phys. Rev. Lett. 124 161301

- [244] Vagnozzi S, Visinelli L, Mena O and Mota D F 2020 Mon. Not. R. Astron. Soc. 493 1139
- [245] Di Luzio L, Giannotti M, Nardi E and Visinelli L 2020 Phys. Rep. 870 1
- [246] Vagnozzi S 2021 Mon. Not. R. Astron. Soc. 502 L11
- [247] Vagnozzi S 2021 Phys. Rev. D 104 063524
- [248] Pati L, Mishra B and Tripathy S K 2021 Phys. Scr. 96 105003
- [249] Benetti M, Graef L L and Vagnozzi S 2022 Phys. Rev. D 105 043520
- [250] Benevento G, Kable J A, Addison G E and Bennett C L 2022 Astrophys. J. 935 156
- [251] Narawade S A, Pati L, Mishra B and Tripathy S K 2022 Phys. Dark Universe 36 101020
- [252] Reeves A, Herold L, Vagnozzi S, Sherwin B D and Ferreira E G M 2023 Mon. Not. R. Astron. Soc. 520 3688
- [253] Gadbail G N, Arora S and Sahoo P K 2023 Ann. Phys., NY 451 169244
- [254] Brax P, Casas S, Desmond H and Elder B 2021 Universe 8 11
- [255] Peebles P J E and Vilenkin A 1999 Phys. Rev. D 59 063505
- [256] de Haro J and Saló L A 2021 Galaxies 9 73
- [257] Maldacena J M 1998 Adv. Theor. Math. Phys. 2 231
- [258] Gubser S S 2005 Class. Quantum Grav. 22 5121
- [259] Horowitz G T and Roberts M M 2009 J. High Energy Phys. JHEP11(2009)015
- [260] Liu H, McGreevy J and Vegh D 2011 Phys. Rev. D 83 065029
- [261] Hawking S W and Penrose R 1970 Proc. R. Soc. A 314 529
- [262] Penrose R 1969 Riv. Nuovo Cimento 1 252
- [263] Wald R M 1997 arXiv:gr-qc/9710068
- [264] Roman T A and Bergmann P G 1983 Phys. Rev. D 28 1265
- [265] Borde A 1997 Phys. Rev. D 55 7615
- [266] Ayon-Beato E and Garcia A 1998 Phys. Rev. Lett. 80 5056
- [267] Ayon-Beato E and Garcia A 1999 Phys. Lett. B 464 25
- [268] Bronnikov K A and Fabris J C 2006 Phys. Rev. Lett. 96 251101
- [269] Iso S, Umetsu H and Wilczek F 2006 Phys. Rev. D 74 044017
- [270] Berej W, Matyjasek J, Tryniecki D and Woronowicz M 2006 Gen. Relativ. Gravit. 38 885
- [271] Bronnikov K A, Konoplya R A and Zhidenko A 2012 Phys. Rev. D 86 024028
- [272] Rinaldi M 2012 Phys. Rev. D 86 084048
- [273] Bambi C and Modesto L 2013 Phys. Lett. B 721 329
- [274] Sebastiani L, Momeni D, Myrzakulov R and Odintsov S D 2013 Phys. Rev. D 88 104022
- [275] Frolov V P 2014 J. High Energy Phys. JHEP05(2014)049
- [276] Toshmatov B, Ahmedov B, Abdujabbarov A and Stuchlik Z 2014 Phys. Rev. D 89 104017
- [277] Bardeen J M 2014 arXiv:1406.4098 [gr-qc]
- [278] Stuchlík Z and Schee J 2014 Int. J. Mod. Phys. D 24 1550020
- [279] Schee J and Stuchlik Z 2015 J. Cosmol. Astropart. Phys. JCAP06(2015)048
- [280] Johannsen T 2013 *Phys. Rev.* D 88 044002
- [281] Junior E L B, Rodrigues M E and Houndjo M J S 2015 J. Cosmol. Astropart. Phys. JCAP10(2015)060
- [282] Myrzakulov R, Sebastiani L and Vagnozzi S 2015 Eur. Phys. J. C 75 444
- [283] Schee J and Stuchlík Z 2016 Class. Quantum Grav. 33 085004
- [284] Abdujabbarov A, Amir M, Ahmedov B and Ghosh S G 2016 Phys. Rev. D 93 104004
- [285] Fan Z-Y and Wang X 2016 Phys. Rev. D 94 124027
- [286] Frolov V P and Zelnikov A 2017 Phys. Rev. D 95 124028
- [287] Toshmatov B, Stuchlík Z and Ahmedov B 2017 Phys. Rev. D 95 084037
- [288] Chinaglia S and Zerbini S 2017 Gen. Relativ. Gravit. 49 75
- [289] Chinaglia S 2017 arXiv:1707.02795 [gr-qc]
- [290] Frolov V P 2018 EPJ Web Conf. 168 01001
- [291] Jusufi K, Övgün A, Saavedra J, Vásquez Y and González P A 2018 Phys. Rev. D 97 124024
- [292] Carballo-Rubio R, Di Filippo F, Liberati S, Pacilio C and Visser M 2018 J. High Energy Phys. JHEP07(2018)023
- [293] Cano P A, Chimento S, Ortín T and Ruipérez A 2019 Phys. Rev. D 99 046014
- [294] Bardeen J M 2018 arXiv:1811.06683 [gr-qc]
- [295] Stuchlík Z and Schee J 2019 Eur. Phys. J. C 79 44
- [296] Övgün A 2019 Phys. Rev. D 99 104075
- [297] Han Y-W, Hu X-Y and Lan M-J 2020 Eur. Phys. J. Plus 135 172
- [298] Rodrigues M E and Silva M V D S 2019 Phys. Rev. D 99 124010

- [299] Panotopoulos G and Rincón A 2019 Eur. Phys. J. Plus 134 300
- [300] Schee J and Stuchlik Z 2019 Eur. Phys. J. C 79 988
- [301] Jusufi K, Jamil M, Chakrabarty H, Wu Q, Bambi C and Wang A 2020 Phys. Rev. D 101 044035
- [302] Ndongmo R, Mahamat S, Bouetou T B and Kofane T C 2021 Phys. Scr. 96 095001
- [303] Bertipagani M, Rinaldi M, Sebastiani L and Zerbini S 2021 Phys. Dark Universe 33 100853
- [304] Nashed G G L and Saridakis E N 2022 J. Cosmol. Astropart. Phys. JCAP05(2022)017
- [305] Simpson A and Visser M 2022 J. Cosmol. Astropart. Phys. JCAP03(2022)011
- [306] Zhao Y, Ren X, Ilyas A, Saridakis E N and Cai Y-F 2022 J. Cosmol. Astropart. Phys. JCAP10(2022)087
- [307] Javed W, Riaz S and Övgün A 2022 Universe 8 262
- [308] Riaz S, Shashank S, Roy R, Abdikamalov A B, Ayzenberg D, Bambi C, Zhang Z and Zhou M 2022 J. Cosmol. Astropart. Phys. JCAP10(2022)040
- [309] Sebastiani L and Zerbini S 2022 Astronomy 1 99
- [310] Franzin E, Liberati S, Mazza J and Vellucci V 2022 Phys. Rev. D 106 104060
- [311] Singh B P, Ali M S and Ghosh S G 2022 arXiv:2207.11907 [gr-qc]
- [312] Chataignier L, Kamenshchik A Y, Tronconi A and Venturi G 2023 Phys. Rev. D 107 023508
- [313] Zhou T and Modesto L 2023 Phys. Rev. D 107 044016
- [314] Olmo G J and Rubiera-Garcia D 2022 arXiv:2209.05061 [gr-qc]
- [315] Bonanno A, Khosravi A-P and Saueressig F 2023 Phys. Rev. D 107 024005
- [316] Carballo-Rubio R, Di Filippo F, Liberati S and Visser M 2022 arXiv:2211.05817 [gr-qc]
- [317] Carballo-Rubio R, Di Filippo F, Liberati S and Visser M 2023 arXiv:2302.00028 [gr-qc]
- [318] Olmo G J, Rosa J a L, Rubiera-Garcia D and Sáez-Chillón Gómez D 2023 arXiv:2302.12064 [gr-qc]
- [319] Reissner H 1916 Ann. Phys., Lpz. 355 106
- [320] Weyl H 1917 Ann. Phys., Lpz. 359 117
- [321] Nordström G 1918 Proc. K. Ned. Akad. Wet. B 20 1238
- [322] Jeffery G B 1921 Proc. R. Soc. A 99 123
- [323] Zajaček M, Tursunov A, Eckart A and Britzen S 2018 Mon. Not. R. Astron. Soc. 480 4408
- [324] Zajaček M, Tursunov A, Eckart A, Britzen S, Hackmann E, Karas V, Stuchlík Z, Czerny B and Zensus J A 2019 J. Phys.: Conf. Ser. 1258 012031
- [325] Wald R M 1974 *Phys. Rev.* D **10** 1680
- [326] Ray S, Espindola A L, Malheiro M, Lemos J P S and Zanchin V T 2003 Phys. Rev. D 68 084004
- [327] Iorio L 2012 Gen. Relativ. Gravit. 44 1753
- [328] Zakharov A F 2014 Phys. Rev. D 90 062007
- [329] Zajacek M and Tursunov A 2019 Observatory 139 231
- [330] Bardeen J 1968 Proc., 5th Int. Conf. on Gravitation and the Theory of Relativity (Tbilisi, USSR, 9–13 December 1968) p 174
- [331] Sakharov A D 1966 Sov. Phys.—JETP 22 241
- [332] Ayon-Beato E and Garcia A 2000 Phys. Lett. B 493 149
- [333] Ayon-Beato E and Garcia A 2005 Gen. Relativ. Gravit. 37 635
- [334] Maluf R V and Neves J C S 2018 Int. J. Mod. Phys. D 28 1950048
- [335] Hayward S A 2006 Phys. Rev. Lett. 96 031103
- [336] Gliner E B 1966 Sov. J. Exp. Theor. Phys. 22 378
- [337] Markov M A 1982 Sov. J. Exp. Theor. Phys. Lett. 36 265
- [338] Markov M A 1987 Sov. J. Exp. Theor. Phys. Lett. 46 431
- [339] Mukhanov V F and Brandenberger R H 1992 Phys. Rev. Lett. 68 1969
- [340] Addazi A et al 2022 Prog. Part. Nucl. Phys. 125 103948
- [341] Frolov V P 2016 Phys. Rev. D 94 104056
- [342] Stehle P and DeBaryshe P G 1966 Phys. Rev. 152 1135
- [343] Born M and Infeld L 1934 Proc. R. Soc. A 144 425
- [344] Fradkin E S and Tseytlin A A 1985 Phys. Lett. B 163 123
- [345] Tseytlin A A 1986 Nucl. Phys. B 276 391
- Tseytlin A A 1987 Nucl. Phys. B 291 876 (erratum)
- [346] Bern Z and Morgan A G 1994 Phys. Rev. D 49 6155
- [347] Dunne G V 2004 Heisenberg-Euler effective Lagrangians: basics and extensions From Fields to Strings: Circumnavigating Theoretical Physics (Ian Kogan Memorial Collection (3 Volume Set) ed M Shifman, A Vainshtein and J Wheater (Singapore: World Scientific) pp 445–522
- [348] Jacobson T N and Ter Veldhuis T 2018 Int. J. Mod. Phys. A 33 1850157

- [349] Bronnikov K A 2001 Phys. Rev. D 63 044005
- [350] Atamurotov F, Ghosh S G and Ahmedov B 2016 Eur. Phys. J. C 76 273
- [351] Ghosh S G 2015 Eur. Phys. J. C 75 532
- [352] Culetu H 2015 Int. J. Theor. Phys. 54 2855
- [353] Simpson A and Visser M 2019 Universe 6 8
- [354] Banerjee I 2022 J. Cosmol. Astropart. Phys. JCAP08(2022)034
- [355] Berry T, Lobo F S N, Simpson A and Visser M 2020 Phys. Rev. D 102 064054
- [356] Berry T, Simpson A and Visser M 2020 Universe 7 2
- [357] Ling Y and Wu M-H 2021 arXiv:2109.05974 [gr-qc]
- [358] Simpson A M 2021 Universe 7 418
- [359] Simpson A and Visser M 2022 Phys. Rev. D 105 064065
- [360] Ling Y and Wu M-H 2022 Symmetry 14 2415
- [361] Guerrero M, Olmo G J, Rubiera-Garcia D and Sáez-Chillón Gómez D 2022 Phys. Rev. D 106 044070
- [362] Banerjee I 2023 Eur. Phys. J. C 83 171
- [363] Kumar R and Ghosh S G 2021 Class. Quantum Grav. 38 8
- [364] Singh D V, Ghosh S G and Maharaj S D 2022 *Nucl. Phys.* B 981 115854
- [365] Kazakov D I and Solodukhin S N 1994 Nucl. Phys. B 429 153
- [366] Kocherlakota P and Rezzolla L 2020 Phys. Rev. D 102 064058
- [367] Ghosh S G and Walia R K 2021 Ann. Phys., NY 434 168619
- [368] Bambi C and Stojkovic D 2021 Universe 7 136
- [369] Morris M S, Thorne K S and Yurtsever U 1988 Phys. Rev. Lett. 61 1446
- [370] Visser M, Kar S and Dadhich N 2003 Phys. Rev. Lett. 90 201102
- [371] Richarte M G and Simeone C 2007 Phys. Rev. D 76 087502
   Richarte M G and Simeone C 2008 Phys. Rev. D 77 089903 (erratum)
- [372] Eiroa E F, Richarte M G and Simeone C 2008 Phys. Lett. A 373 1 Eiroa E F, Richarte M G and Simeone C 2009 Phys. Lett. A 373 2399–400 (erratum)
- [373] Lobo F S N and Oliveira M A 2009 Phys. Rev. D 80 104012
- [374] Richarte M G 2010 Phys. Rev. D 82 044021
- [375] Kanti P, Kleihaus B and Kunz J 2011 Phys. Rev. Lett. 107 271101
- [376] Boehmer C G, Harko T and Lobo F S N 2012 Phys. Rev. D 85 044033
- [377] Bolokhov S V, Bronnikov K A and Skvortsova M V 2012 Class. Quantum Grav. 29 245006
- [378] Capozziello S, Harko T, Koivisto T S, Lobo F S N and Olmo G J 2012 Phys. Rev. D 86 127504
- [379] Harko T, Lobo F S N, Mak M K and Sushkov S V 2013 Phys. Rev. D 87 067504
- [380] Richarte M G 2013 Phys. Rev. D 87 067503
- [381] Di Criscienzo R, Myrzakulov R and Sebastiani L 2013 Class. Quantum Grav. 30 235013
- [382] Shaikh R 2015 Phys. Rev. D 92 024015
- [383] Bahamonde S, Camci U, Capozziello S and Jamil M 2016 Phys. Rev. D 94 084042
- [384] Övgün A and Jusufi K 2017 Adv. High Energy Phys. 2017 1215254
- [385] Moraes P H R S, Correa R A C and Lobato R V 2017 J. Cosmol. Astropart. Phys. JCAP07(2017)029
- [386] Övgün A and Salako I G 2017 Mod. Phys. Lett. A 32 1750119
- [387] Shaikh R and Kar S 2017 *Phys. Rev.* D 96 044037
- [388] Ovgün A and Jusufi K 2017 Eur. Phys. J. Plus 132 543
- [389] Jusufi K, Ovgün A and Banerjee A 2017 Phys. Rev. D 96 084036 Jusufi K, Ovgün A and Banerjee A 2017 Phys. Rev. D 96 089904 (addendum)
- [390] Moraes P H R S and Sahoo P K 2017 Phys. Rev. D 96 044038
- [391] Jusufi K and Övgün A 2018 Phys. Rev. D 97 024042
- [392] Moraes P H R S and Sahoo P K 2018 Phys. Rev. D 97 024007
- [393] Calzà M, Rinaldi M and Sebastiani L 2018 Eur. Phys. J. C 78 178
- [394] Jusufi K, Övgün A, Banerjee A and Sakallı t I 2019 Eur. Phys. J. Plus 134 428
- [395] Övgün A, Jusufi K and Sakallı I 2019 Phys. Rev. D 99 024042
- [396] Amir M, Jusufi K, Banerjee A and Hansraj S 2019 Class. Quantum Grav. 36 215007
- [397] Shaikh R 2018 Phys. Rev. D 98 064033
- [398] Javed W, Babar R and Övgün A 2019 Phys. Rev. D 99 084012
- [399] Antoniou G, Bakopoulos A, Kanti P, Kleihaus B and Kunz J 2020 Phys. Rev. D 101 024033
- [400] Singh K N, Banerjee A, Rahaman F and Jasim M K 2020 *Phys. Rev.* D 101 084012
- [401] Mishra B, Agrawal A S, Tripathy S K and Ray S K T S 2021 Int. J. Mod. Phys. D 30 2150061

- [402] De Falco V, De Laurentis M and Capozziello S 2021 Phys. Rev. D 104 024053
- [403] Benavides-Gallego C A, Han W-B, Malafarina D, Ahmedov B and Abdujabbarov A 2021 Phys. Rev. D 104 084024
- [404] Karakasis T, Papantonopoulos E and Vlachos C 2022 Phys. Rev. D 105 024006
- [405] Mustafa G, Hassan Z, Moraes P H R S and Sahoo P K 2021 Phys. Lett. B 821 136612
- [406] Sadeghi J, Shokri M, Gashti S N, Pourhassan B and Rudra P 2022 Int. J. Mod. Phys. D 31 2250019
- [407] Agrawal A S, Mishra B, Tello-Ortiz F and Alvarez A 2022 Fortschr. Phys. 70 2100177
- [408] Mishra B, Agrawal A S, Tripathy S K and Ray S 2022 Int. J. Mod. Phys. A 37 2250010
- [409] Moraes P H R S, Agrawal A S and Mishra B 2022 arXiv:2201.08392 [gr-qc]
- [410] Kar S 2022 Gen. Relativ. Gravit. 54 66
- [411] Abdulxamidov F, Benavides-Gallego C A, Han W-B, Rayimbaev J and Abdujabbarov A 2022 Phys. Rev. D 106 024012
- [412] Hassan Z, Ghosh S, Sahoo P K and Rao V S H 2022 arXiv:2209.02704 [gr-qc]
- [413] Rosa J a L and Kull P M 2022 Eur. Phys. J. C 82 1154
- [414] Capozziello S and Godani N 2022 Phys. Lett. B 835 137572
- [415] Gao K, Liu L-H and Zhu M 2022 arXiv:2211.17065 [gr-qc]
- [416] Cai T, Huang H, Wang Z and Zhu M 2023 arXiv:2302.13704 [gr-qc]
- [417] Simpson A and Visser M 2019 J. Cosmol. Astropart. Phys. JCAP02(2019)042
- [418] Ellis H G 1973 J. Math. Phys. 14 104
- [419] Bronnikov K A 1973 Acta Phys. Polon. B 4 251
- [420] Carballo-Rubio R, Di Filippo F, Liberati S and Visser M 2020 Class. Quantum Grav. 37 14
- [421] Huang H and Yang J 2019 Phys. Rev. D 100 124063
- [422] Carballo-Rubio R, Di Filippo F, Liberati S and Visser M 2020 Phys. Rev. D 101 084047
- [423] Churilova M S and Stuchlik Z 2020 Class. Quantum Grav. 37 075014
- [424] Bronnikov K A and Konoplya R A 2020 Phys. Rev. D 101 064004
- [425] Lobo F S N, Simpson A and Visser M 2020 Phys. Rev. D 101 124035
- [426] Nascimento J R, Petrov A Y, Porfirio P J and Soares A R 2020 Phys. Rev. D 102 044021
- [427] Lima H C D, Benone C L and Crispino L C B 2020 Phys. Rev. D 101 124009
- [428] Lobo F S N, Rodrigues M E, Silva M V d S, Simpson A and Visser M 2021 Phys. Rev. D 103 084052
- [429] Zhou T-Y and Xie Y 2020 Eur. Phys. J. C 80 1070
- [430] Ayuso I, Lobo F S N and Mimoso J P 2021 Phys. Rev. D 103 044018
- [431] Tsukamoto N 2021 Phys. Rev. D 103 064031
- [432] Mazza J, Franzin E and Liberati S 2021 J. Cosmol. Astropart. Phys. JCAP04(2021)082
- [433] Bronnikov K A, Konoplya R A and Pappas T D 2021 Phys. Rev. D 103 124062
- [434] Cheng X-T and Xie Y 2021 Phys. Rev. D 103 064040
- [435] Islam S U, Kumar J and Ghosh S G 2021 J. Cosmol. Astropart. Phys. JCAP10(2021)013
- [436] Franzin E, Liberati S, Mazza J, Simpson A and Visser M 2021 J. Cosmol. Astropart. Phys. JCAP07(2021)036
- [437] Yang J and Huang H 2021 Phys. Rev. D 104 084005
- [438] Guerrero M, Olmo G J, Rubiera-Garcia D and Gómez D S-C 2021 J. Cosmol. Astropart. Phys. JCAP08(2021)036
- [439] Chakrabarti S and Kar S 2021 Phys. Rev. D 104 024071
- [440] Yang Y, Liu D, Xu Z, Xing Y, Wu S and Long Z-W 2021 Phys. Rev. D 104 104021
- [441] Jiang X, Wang P, Wu H and Yang H 2021 Eur. Phys. J. C 81 1043
- Jiang X, Wang P, Wu H and Yang H 2022 Eur. Phys. J. C 82 5 (erratum)
- [442] Stuchlík Z and Vrba J 2021 Universe 7 279
- [443] Xu Z and Tang M 2021 *Eur. Phys. J.* C **81** 863
- [444] Bambhaniya P, Saurabh K, Jusufi K and Joshi P S 2022 Phys. Rev. D 105 023021
- [445] Ou M-Y, Lai M-Y and Huang H 2022 Eur. Phys. J. C 82 452
- [446] Guo Y and Miao Y-G 2022 *Nucl. Phys.* B **983** 115938
- [447] Tsukamoto N 2022 Phys. Rev. D 105 084036
- [448] Yang Y, Liu D, Övgün A, Long Z-W and Xu Z 2022 arXiv:2205.07530 [gr-qc]
- [449] Calzá M 2023 Phys. Rev. D 107 044067
- [450] Bronnikov K A and Walia R K 2022 Phys. Rev. D 105 044039
- [451] Reynolds C S 2014 Space Sci. Rev. 183 277
- [452] Lima H C D Jr, Crispino L C B, Cunha P V P and Herdeiro C A R 2021 Phys. Rev. D 103 084040

- [453] Tsukamoto N 2021 Phys. Rev. D 103 024033
- [454] Morris M S and Thorne K S 1988 Am. J. Phys. 56 395
- [455] Lemos J P S, Lobo F S N and Quinet de Oliveira S 2003 Phys. Rev. D 68 064004
- [456] Harko T, Kovacs Z and Lobo F S N 2008 *Phys. Rev.* D 78 084005
- [457] Bambi C 2013 Phys. Rev. D 87 084039
- [458] Bambi C 2013 Phys. Rev. D 87 107501
- [459] Arnowitt R L, Deser S and Misner C W 1959 Phys. Rev. 116 1322
- [460] Arnowitt R L, Deser S and Misner C W 2008 Gen. Relativ. Gravit. 40 1997
- [461] Damour T and Solodukhin S N 2007 Phys. Rev. D 76 024016
- [462] Nandi K K, Izmailov R N, Zhdanov E R and Bhattacharya A 2018 J. Cosmol. Astropart. Phys. JCAP07(2018)027
- [463] Övgün A 2018 Phys. Rev. D 98 044033
- [464] Karimov R K, Izmailov R N and Nandi K K 2019 Eur. Phys. J. C 79 952
- [465] Tsukamoto N 2020 Phys. Rev. D 101 104021
- [466] Matyjasek J 2020 Phys. Rev. D 102 024082
- [467] Kodama H 1980 Prog. Theor. Phys. 63 1217
- [468] Hayward S A, Di Criscienzo R, Vanzo L, Nadalini M and Zerbini S 2009 Class. Quantum Grav. 26 062001
- [469] Di Criscienzo R, Hayward S A, Nadalini M, Vanzo L and Zerbini S 2010 Class. Quantum Grav. 27 015006
- [470] Vanzo L, Acquaviva G and Di Criscienzo R 2011 Class. Quantum Grav. 28 183001
- [471] Solodukhin S N 2005 Phys. Rev. D 71 064006
- [472] Joshi P S and Malafarina D 2011 Int. J. Mod. Phys. D 20 2641
- [473] Birkhoff G D and Langer R E 1923 Relativity and Modern Physics (Cambridge, MA: Harvard University Press)
- [474] Oppenheimer J R and Snyder H 1939 Phys. Rev. 56 455
- [475] Datt B 1938 Z. Phys. 108 314
- [476] Ori A and Piran T 1987 Phys. Rev. Lett. 59 2137
- [477] Lake K 1991 Phys. Rev. D 43 1416
- [478] Joshi P S and Dwivedi I H 1993 Phys. Rev. D 47 5357
- [479] Harada T, Iguchi H and Nakao K-I 1998 *Phys. Rev.* D 58 041502
- [480] Joshi P S, Dadhich N and Maartens R 2002 Phys. Rev. D 65 101501
- [481] Giambo R, Giannoni F, Magli G and Piccione P 2004 Gen. Relativ. Gravit. 36 1279
- [482] Goswami R and Joshi P S 2007 Phys. Rev. D 76 084026
- [483] Patil M and Joshi P S 2011 Phys. Rev. D 84 104001
- [484] Chowdhury A N, Patil M, Malafarina D and Joshi P S 2012 Phys. Rev. D 85 104031
- [485] Shaikh R, Kocherlakota P, Narayan R and Joshi P S 2019 Mon. Not. R. Astron. Soc. 482 52
- [486] Virbhadra K S 1999 Phys. Rev. D 60 104041
- [487] Ghosh S G and Dadhich N 2001 Phys. Rev. D 64 047501
- [488] Casadio R and Harms B 2002 Int. J. Mod. Phys. A 17 4635
- [489] Banerjee A, Debnath U and Chakraborty S 2003 Int. J. Mod. Phys. D 12 1255
- [490] Debnath U and Chakraborty S 2004 J. Cosmol. Astropart. Phys. JCAP05(2004)001
- [491] Miyamoto U, Maeda H and Harada T 2005 Prog. Theor. Phys. 113 513
- [492] Torok G and Stuchlik Z 2005 Astron. Astrophys. 437 775
- [493] Maeda H, Harada T and Carr B J 2008 Phys. Rev. D 77 024023
- [494] Virbhadra K S and Keeton C R 2008 Phys. Rev. D 77 124014
- [495] Barausse E, Cardoso V and Khanna G 2010 Phys. Rev. Lett. 105 261102
- [496] Patil M and Joshi P S 2011 Class. Quantum Grav. 28 235012
- [497] Sahu S, Patil M, Narasimha D and Joshi P S 2012 Phys. Rev. D 86 063010
- [498] Stuchlík Z, Schee J and Abdujabbarov A 2014 Phys. Rev. D 89 104048
- [499] Liu H, Zhou M and Bambi C 2018 J. Cosmol. Astropart. Phys. JCAP08(2018)044
- [500] Jusufi K, Banerjee A, Gyulchev G and Amir M 2019 Eur. Phys. J. C 79 28
- [501] Hioki K and Harada T 2020 Int. J. Mod. Phys. A 35 2050201
- [502] Paul S 2020 Phys. Rev. D 102 064045
- [503] Tsukamoto N 2021 Phys. Rev. D 104 124016
- [504] Tsukamoto N 2022 Phys. Rev. D 105 024009
- [505] Atamurotov F and Ghosh S G 2022 Eur. Phys. J. Plus 137 662
- [506] Janis A I, Newman E T and Winicour J 1968 Phys. Rev. Lett. 20 878

- [507] Fisher I Z 1948 Zh. Eksp. Teor. Fiz. 18 636
- [508] Bronnikov K A and Khodunov A V 1979 Gen. Relativ. Gravit. 11 13
- [509] Wyman M 1981 Phys. Rev. D 24 839
- [510] Virbhadra K S 1997 Int. J. Mod. Phys. A 12 4831
- [511] Gyulchev G N and Yazadjiev S S 2008 Phys. Rev. D 78 083004
- [512] Kovacs Z and Harko T 2010 Phys. Rev. D 82 124047
- [513] Chakraborty C, Kocherlakota P and Joshi P S 2017 Phys. Rev. D 95 044006
- [514] Chakraborty C, Patil M, Kocherlakota P, Bhattacharyya S, Joshi P S and Królak A 2017 Phys. Rev. D 95 084024
- [515] Rizwan M, Jamil M and Wang A 2018 Phys. Rev. D 98 024015 Rizwan M, Jamil M and Wang A 2019 Phys. Rev. D 100 029902 (erratum)
- [516] Gyulchev G, Nedkova P, Vetsov T and Yazadjiev S 2019 Phys. Rev. D 100 024055
- [517] Capozziello S, D'Agostino R and Gregoris D 2020 Phys. Dark Universe 28 100513
- [518] Gyulchev G, Kunz J, Nedkova P, Vetsov T and Yazadjiev S 2020 Eur. Phys. J. C 80 1017
- [519] Chowdhury A and Banerjee N 2020 Phys. Rev. D 102 124051
- [520] Joshi P S, Malafarina D and Narayan R 2011 Class. Quantum Grav. 28 235018
- [521] Joshi P S, Malafarina D and Narayan R 2014 Class. Quantum Grav. 31 015002
- [522] Dey D, Kocherlakota P and Joshi P S 2019 arXiv:1907.12792 [gr-qc]
- [523] Pal K, Pal K, Roy P and Sarkar T 2022 arXiv:2206.11764 [gr-qc]
- [524] Bambhaniya P, Joshi A B, Dey D and Joshi P S 2019 Phys. Rev. D 100 124020
- [525] Kaur K P, Joshi P S, Dey D, Joshi A B and Desai R P 2021 arXiv:2106.13175 [gr-qc]
- [526] Dey D, Shaikh R and Joshi P S 2020 Phys. Rev. D 102 044042
- [527] Israel W 1966 Nuovo Cimento B 44 1
- Israel W 1967 Nuovo Cimento B 48 463 (erratum)
- [528] Ori A and Piran T 1990 Phys. Rev. D 42 1068
- [529] Jhingan S and Magli G 2000 Phys. Rev. D 61 124006
- [530] Kudoh H, Harada T and Iguchi H 2000 Phys. Rev. D 62 104016
- [531] Goswami R, Joshi P S, Vaz C and Witten L 2004 Phys. Rev. D 70 084038
- [532] Banik U, Dey D, Bhattacharya K and Sarkar T 2017 Gen. Relativ. Gravit. 49 116
- [533] Bhattacharya K, Dey D, Mazumdar A and Sarkar T 2020 Phys. Rev. D 101 043005
- [534] Mosani K, Dey D and Joshi P S 2020 Phys. Rev. D 101 044052
- [535] Dey D, Joshi P S and Shaikh R 2021 Phys. Rev. D 103 024015
- [536] González P A, Papantonopoulos E, Saavedra J and Vásquez Y 2013 J. High Energy Phys. JHEP12(2013)021
- [537] González P A, Papantonopoulos E, Saavedra J and Vásquez Y 2014 J. High Energy Phys. JHEP11(2014)011
- [538] Astorino M 2013 Phys. Rev. D 88 104027
- [539] Martinez C, Troncoso R and Staforelli J P 2006 Phys. Rev. D 74 044028
- [540] Cisterna A, Erices C, Kuang X-M and Rinaldi M 2018 Phys. Rev. D 97 124052
- [541] Myung Y S and Zou D-C 2019 *Phys. Lett.* B **790** 400 [542] Myung Y S and Zou D-C 2019 *Eur. Phys. J.* C **79** 641
- [543] Myung Y S and Zou D-C 2019 Phys. Rev. D 100 064057
- [544] Zou D-C and Myung Y S 2020 Phys. Lett. B 803 135332
- [545] Zou D-C and Myung Y S 2020 Phys. Rev. D 101 084021
- [546] Caceres N, Figueroa J, Oliva J, Oyarzo M and Stuardo R 2020 J. High Energy Phys. JHEP04(2020)157
- [547] Bocharova N M, Bronnikov K A and Melnikov V N 1970 Vestn. Mosk. Univ. III 6 706
- [548] Bekenstein J D 1974 Ann. Phys., NY 82 535
- [549] Chowdhury A and Banerjee N 2018 Eur. Phys. J. C 78 594
- [550] Castro A, Maloney A and Strominger A 2010 Phys. Rev. D 82 024008
- [551] Bebronne M V and Tinyakov P G 2009 J. High Energy Phys. JHEP04(2009)100
- Bebronne M V and Tinyakov P G 2011 J. High Energy Phys. JHEP06(2011)018 (erratum)
- [552] Jawad A, Jusufi K and Shahzad M U 2021 Phys. Rev. D 104 084045
- [553] Nojiri S and Odintsov S D 2006 Phys. Rev. D 74 086005
- [554] Amendola L, Gannouji R, Polarski D and Tsujikawa S 2007 Phys. Rev. D 75 083504
- [555] Fay S, Nesseris S and Perivolaropoulos L 2007 Phys. Rev. D 76 063504
- [556] Hu W and Sawicki I 2007 Phys. Rev. D 76 064004
- [557] Carloni S, Troisi A and Dunsby P K S 2009 Gen. Relativ. Gravit. 41 1757

- [558] Starobinsky A A 2007 JETP Lett. 86 157
- [559] Cognola G, Elizalde E, Nojiri S, Odintsov S D, Sebastiani L and Zerbini S 2008 Phys. Rev. D 77 046009
- [560] Brax P, van de Bruck C, Davis A-C and Shaw D J 2008 Phys. Rev. D 78 104021
- [561] de la Cruz-Dombriz A, Dobado A and Maroto A L 2009 Phys. Rev. D 80 124011 de la Cruz-Dombriz A, Dobado A and Maroto A L 2011 Phys. Rev. D 83 029903 (erratum)
- [562] Schmidt F, Vikhlinin A and Hu W 2009 Phys. Rev. D 80 083505
- [563] Tsujikawa S, Gannouji R, Moraes B and Polarski D 2009 Phys. Rev. D 80 084044
- [564] Faraoni V 2010 Entropy 12 1246
- [565] Sebastiani L and Zerbini S 2011 Eur. Phys. J. C 71 1591
- [566] Moon T, Myung Y S and Son E J 2011 Gen. Relativ. Gravit. 43 3079
- [567] Olmo G J 2011 Int. J. Mod. Phys. D 20 413
- [568] Elizalde E, Odintsov S D, Sebastiani L and Zerbini S 2012 Eur. Phys. J. C 72 1843
- [569] Cembranos J A R, de la Cruz-Dombriz A and Jimeno Romero P 2014 Int. J. Geom. Methods Mod. Phys. 11 1450001
- [570] Jennings E, Baugh C M, Li B, Zhao G-B and Koyama K 2012 Mon. Not. R. Astron. Soc. 425 2128
- [571] Li B, Hellwing W A, Koyama K, Zhao G-B, Jennings E and Baugh C M 2013 Mon. Not. R. Astron. Soc. 428 743
- [572] de la Cruz-Dombriz A and Saez-Gomez D 2012 Entropy 14 1717
- [573] Basilakos S, Nesseris S and Perivolaropoulos L 2013 Phys. Rev. D 87 123529
- [574] Cai Y-F and Saridakis E N 2014 Phys. Rev. D 90 063528
- [575] Rinaldi M, Cognola G, Vanzo L and Zerbini S 2014 J. Cosmol. Astropart. Phys. JCAP08(2014)015
- [576] Rinaldi M, Cognola G, Vanzo L and Zerbini S 2015 Phys. Rev. D 91 123527
- [577] Sebastiani L and Myrzakulov R 2015 Int. J. Geom. Methods Mod. Phys. 12 1530003
- [578] Odintsov S D and Oikonomou V K 2016 Phys. Rev. D 93 023517
- [579] Khodadi M, Heydarzade Y, Darabi F and Saridakis E N 2016 Phys. Rev. D 93 124019
- [580] Hu B, Raveri M, Rizzato M and Silvestri A 2016 Mon. Not. R. Astron. Soc. 459 3880
- [581] Oikonomou V K 2016 Int. J. Mod. Phys. D 25 1650078
- [582] De Martino I 2016 Phys. Rev. D 93 124043
- [583] Peirone S, Raveri M, Viel M, Borgani S and Ansoldi S 2017 Phys. Rev. D 95 023521
- [584] Burrage C and Sakstein J 2016 J. Cosmol. Astropart. Phys. JCAP11(2016)045
- [585] Odintsov S D, Oikonomou V K and Sebastiani L 2017 Nucl. Phys. B 923 608
- [586] Naik A P, Puchwein E, Davis A-C and Arnold C 2018 Mon. Not. R. Astron. Soc. 480 5211
- [587] Nashed G G L 2018 Eur. Phys. J. Plus 133 18
- [588] Kleidis K and Oikonomou V K 2018 Int. J. Geom. Methods Mod. Phys. 15 1850212
- [589] Banerjee I, Mandal B and SenGupta S 2020 Phys. Rev. D 101 024013
- [590] Odintsov S D, Oikonomou V K and Paul T 2020 Nucl. Phys. B 959 115159
- [591] Nashed G G L and Nojiri S 2021 Phys. Lett. B 820 136475
- [592] Nashed G G L 2021 Phys. Lett. B 812 136012
- [593] Nashed G G L and Nojiri S 2021 Phys. Rev. D 104 124054
- [594] Vagnozzi S, Visinelli L, Brax P, Davis A-C and Sakstein J 2021 Phys. Rev. D 104 063023
- [595] Farrugia C R, Sultana J and Mifsud J 2021 Phys. Rev. D 104 123503
- [596] Oikonomou V K 2021 Ann. Phys., NY 432 168576
- [597] Odintsov S D, Oikonomou V K and Fronimos F P 2022 Phys. Dark Universe 35 100950
- [598] Astashenok A V, Capozziello S, Odintsov S D and Oikonomou V K 2021 *Europhys. Lett.* 136 59001
- [599] Odintsov S D and Oikonomou V K 2022 Phys. Lett. B 824 136817
- [600] Geng C-Q, Hsu Y-T and Lu J-R 2022 Astrophys. J. 926 74
- [601] Oikonomou V K and Giannakoudi I 2022 Int. J. Mod. Phys. D 31 2250075
- [602] Nojiri S, Odintsov S D and Oikonomou V K 2022 Nucl. Phys. B 980 115850
- [603] Oikonomou V K 2022 Ann. Phys., Lpz. 534 2200134
- [604] Oikonomou V K 2022 Phys. Rev. D 106 044041
- [605] Oikonomou V K 2023 Astropart. Phys. 144 102777
- [606] Astashenok A V, Odintsov S D and Oikonomou V K 2022 Phys. Rev. D 106 124010
- [607] Oikonomou V K 2022 Nucl. Phys. B 984 115985
- [608] Oikonomou V K 2023 Mon. Not. R. Astron. Soc. 520 2934

- [609] Clifton T and Barrow J D 2005 *Phys. Rev.* D **72** 103005
- Clifton T and Barrow J D 2014 Phys. Rev. D 90 029902 (erratum)
- [610] Horndeski G W 1974 Int. J. Theor. Phys. 10 363
- [611] Ostrogradsky M 1850 Mem. Acad. St. Petersbourg 6 385
- [612] Gleyzes J, Langlois D, Piazza F and Vernizzi F 2015 Phys. Rev. Lett. 114 211101
- [613] Gleyzes J, Langlois D, Piazza F and Vernizzi F 2015 J. Cosmol. Astropart. Phys. JCAP02(2015)018
- [614] Zumalacárregui M and García-Bellido J 2014 Phys. Rev. D 89 064046
- [615] Deffayet C, Esposito-Farese G and Steer D A 2015 Phys. Rev. D 92 084013
- [616] Langlois D and Noui K 2016 J. Cosmol. Astropart. Phys. JCAP07(2016)016
- [617] Crisostomi M, Hull M, Koyama K and Tasinato G 2016 J. Cosmol. Astropart. Phys. JCAP03(2016)038
- [618] Ben Achour J, Langlois D and Noui K 2016 Phys. Rev. D 93 124005
- [619] Babichev E, Charmousis C and Lehébel A 2016 Class. Quantum Grav. 33 154002
- [620] Ben Achour J, Crisostomi M, Koyama K, Langlois D, Noui K and Tasinato G 2016 J. High Energy Phys. JHEP12(2016)100
- [621] Langlois D, Mancarella M, Noui K and Vernizzi F 2019 J. Cosmol. Astropart. Phys. JCAP02(2019)036
- [622] Nicolis A, Rattazzi R and Trincherini E 2009 Phys. Rev. D 79 064036
- [623] Deffayet C, Esposito-Farese G and Vikman A 2009 Phys. Rev. D 79 084003
- [624] Bettoni D and Liberati S 2013 Phys. Rev. D 88 084020
- [625] Kase R and Tsujikawa S 2014 Phys. Rev. D 90 044073
- [626] Kase R and Tsujikawa S 2015 Int. J. Mod. Phys. D 23 1443008
- [627] Tsujikawa S 2015 J. Cosmol. Astropart. Phys. JCAP04(2015)043
- [628] Cisterna A, Delsate T, Ducobu L and Rinaldi M 2016 Phys. Rev. D 93 084046
- [629] Kobayashi T 2016 Phys. Rev. D 94 043511
- [630] Rinaldi M 2017 Phys. Dark Universe 16 14
- [631] Banerjee S, Cai Y-F and Saridakis E N 2019 Class. Quantum Grav. 36 135009
- [632] Bahamonde S, Dialektopoulos K F and Levi Said J 2019 Phys. Rev. D 100 064018
- [633] Oikonomou V K and Fronimos F P 2021 Class. Quantum Grav. 38 035013
- [634] Bernardo R C and Levi Said J 2021 J. Cosmol. Astropart. Phys. JCAP09(2021)014
- [635] Bahamonde S, Hohmann M, Caruana M, Dialektopoulos K F, Gakis V, Levi Said J, Saridakis E N and Sultana J 2021 Phys. Rev. D 104 084082
- [636] Bernardo R C, Said J L, Caruana M and Appleby S 2021 J. Cosmol. Astropart. Phys. JCAP10(2021)078
- [637] Bernardo R C, Said J L, Caruana M and Appleby S 2022 Class. Quantum Grav. 39 015013
- [638] Oikonomou V K and Fronimos F P 2021 Nucl. Phys. B 971 115522
- [639] Hui L and Nicolis A 2013 Phys. Rev. Lett. 110 241104
- [640] Anabalon A, Cisterna A and Oliva J 2014 Phys. Rev. D 89 084050
- [641] Minamitsuji M 2014 Phys. Rev. D 89 064017
- [642] Sotiriou T P and Zhou S-Y 2014 Phys. Rev. D 90 124063
- [643] Maselli A, Silva H O, Minamitsuji M and Berti E 2015 Phys. Rev. D 92 104049
- [644] Babichev E, Charmousis C and Lehébel A 2017 J. Cosmol. Astropart. Phys. JCAP04(2017)027
- [645] Bergliaffa S E P, Maier R and Silvano N D O 2021 arXiv:2107.07839 [gr-qc]
- [646] Deffayet C, Pujolas O, Sawicki I and Vikman A 2010 J. Cosmol. Astropart. Phys. JCAP10(2010)026
- [647] Babichev E and Esposito-Farèse G 2013 Phys. Rev. D 87 044032
- [648] Couch W E and Torrence R J 1984 Gen. Relativ. Gravit. 16 789
- [649] Bizon P and Friedrich H 2013 Class. Quantum Grav. 30 065001
- [650] Fernandes K, Ghosh D and Virmani A 2020 Class. Quantum Grav. 38 055006
- [651] Bellini E, Jimenez R and Verde L 2015 J. Cosmol. Astropart. Phys. JCAP05(2015)057
- [652] Bellini E, Cuesta A J, Jimenez R and Verde L 2016 J. Cosmol. Astropart. Phys. JCAP02(2016)053
  - Bellini E, Cuesta A J, Jimenez R and Verde L 2016 J. Cosmol. Astropart. Phys. JCAP06(2016)E01 (erratum)
- [653] Myrzakulov R and Sebastiani L 2016 Astrophys. Space Sci. 361 62
- [654] Sakstein J, Wilcox H, Bacon D, Koyama K and Nichol R C 2016 J. Cosmol. Astropart. Phys. JCAP07(2016)019

- [655] Zumalacárregui M, Bellini E, Sawicki I, Lesgourgues J and Ferreira P G 2017 J. Cosmol. Astropart. Phys. JCAP08(2017)019
- [656] Pogosian L and Silvestri A 2016 Phys. Rev. D 94 104014
- [657] Perenon L, Marinoni C and Piazza F 2017 J. Cosmol. Astropart. Phys. JCAP01(2017)035
- [658] Renk J, Zumalacárregui M, Montanari F and Barreira A 2017 J. Cosmol. Astropart. Phys. JCAP10(2017)020
- [659] Arroja F, Okumura T, Bartolo N, Karmakar P and Matarrese S 2018 J. Cosmol. Astropart. Phys. JCAP05(2018)050
- [660] Creminelli P and Vernizzi F 2017 Phys. Rev. Lett. 119 251302
- [661] Sakstein J and Jain B 2017 Phys. Rev. Lett. 119 251303
- [662] Ezquiaga J M and Zumalacárregui M 2017 Phys. Rev. Lett. 119 251304
- [663] Baker T, Bellini E, Ferreira P G, Lagos M, Noller J and Sawicki I 2017 Phys. Rev. Lett. 119 251301
- [664] Kreisch C D and Komatsu E 2018 J. Cosmol. Astropart. Phys. JCAP12(2018)030
- [665] Dima A and Vernizzi F 2018 Phys. Rev. D 97 101302
- [666] Kase R and Tsujikawa S 2018 Phys. Rev. D 97 103501
- [667] Casalino A, Rinaldi M, Sebastiani L and Vagnozzi S 2018 Phys. Dark Universe 22 108
- [668] Kase R and Tsujikawa S 2019 Int. J. Mod. Phys. D 28 1942005
- [669] Frusciante N, Peirone S, Casas S and Lima N A 2019 Phys. Rev. D 99 063538
- [670] Casalino A, Rinaldi M, Sebastiani L and Vagnozzi S 2019 Class. Quantum Grav. 36 017001
- [671] Noller J and Nicola A 2019 Phys. Rev. D 99 103502
- [672] Duniya D, Moloi T, Clarkson C, Larena J, Maartens R, Mongwane B and Weltman A 2020 J. Cosmol. Astropart. Phys. JCAP01(2020)033
- [673] Traykova D, Bellini E and Ferreira P G 2019 J. Cosmol. Astropart. Phys. JCAP08(2019)035
- [674] Perenon L and Velten H 2019 Universe 5 138
- [675] Pace F, Battye R A, Bolliet B and Trinh D 2019 J. Cosmol. Astropart. Phys. JCAP09(2019)018
- [676] Peirone S, Benevento G, Frusciante N and Tsujikawa S 2019 *Phys. Rev.* D 100 063509
- [677] Bahamonde S, Dialektopoulos K F, Gakis V and Levi Said J 2020 Phys. Rev. D 101 084060
- [678] Traykova D, Bellini E, Ferreira P G, García-García C, Noller J and Zumalacárregui M 2021 Phys. Rev. D 104 083502
- [679] Reyes M and Escamilla-Rivera C 2021 J. Cosmol. Astropart. Phys. JCAP07(2021)048
- [680] Petronikolou M, Basilakos S and Saridakis E N 2022 Phys. Rev. D 106 124051
- [681] Papallo G 2017 Phys. Rev. D 96 124036
- [682] Moffat J W 2006 J. Cosmol. Astropart. Phys. JCAP03(2006)004
- [683] Brownstein J R and Moffat J W 2006 Astrophys. J. 636 721
- [684] Brownstein J R and Moffat J W 2006 Mon. Not. R. Astron. Soc. 367 527
- [685] Brownstein J R and Moffat J W 2007 Mon. Not. R. Astron. Soc. 382 29
- [686] Moffat J W and Toth V T 2008 Astrophys. J. 680 1158
- [687] Moffat J W and Rahvar S 2013 Mon. Not. R. Astron. Soc. 436 1439
- [688] Moffat J W and Rahvar S 2014 Mon. Not. R. Astron. Soc. 441 3724
- [689] Moffat J W and Toth V T 2015 Phys. Rev. D 91 043004
- [690] Moffat J W 2015 Eur. Phys. J. C 75 130
- [691] Frusciante N, Raveri M, Vernieri D, Hu B and Silvestri A 2016 Phys. Dark Universe 13 7
- [692] Hussain S and Jamil M 2015 Phys. Rev. D 92 043008
- [693] Lopez Armengol F G and Romero G E 2017 Gen. Relativ. Gravit. 49 27
- [694] De Martino I and De Laurentis M 2017 Phys. Lett. B 770 440
- [695] Pérez D, Lopez Armengol F G and Romero G E 2017 Phys. Rev. D 95 104047
- [696] Oman K A, Marasco A, Navarro J F, Frenk C S, Schaye J and Benítez-Llambay A 2019 Mon. Not. R. Astron. Soc. 482 821
- [697] Moffat J W and Toth V T 2020 Phys. Rev. D 101 024014
- [698] Haghi H, Amiri V, Zonoozi A H, Banik I, Kroupa P and Haslbauer M 2019 Astrophys. J. Lett. 884 L25
- [699] Liu Y, Qian W-L, Gong Y and Wang B 2021 Universe 7 9
- [700] Moffat J W and Toth V T 2021 Phys. Rev. D 103 044045
- [701] Moffat J W 2020 arXiv:2008.04404 [gr-qc]
- [702] Cai X-C and Miao Y-G 2020 Phys. Rev. D 102 084061
- [703] Cai X-C and Miao Y-G 2021 Eur. Phys. J. C 81 559
- [704] Moffat J W and Toth V 2021 Universe 7 358

- [705] Della Monica R, de Martino I and de Laurentis M 2022 Mon. Not. R. Astron. Soc. 510 4757
- [706] Davari Z and Rahvar S 2021 Mon. Not. R. Astron. Soc. 507 3387
- [707] Moffat J W and Toth V T 2021 Eur. Phys. J. C 81 836
- [708] Moffat J W 2015 Eur. Phys. J. C 75 175
- [709] Guo M, Obers N A and Yan H 2018 Phys. Rev. D 98 084063
- [710] Wang H-M, Xu Y-M and Wei S-W 2019 J. Cosmol. Astropart. Phys. JCAP03(2019)046
- [711] Qin X, Chen S, Zhang Z and Jing J 2022 Astrophys. J. 938 2
- [712] Antoniadis I, Arkani-Hamed N, Dimopoulos S and Dvali G R 1998 Phys. Lett. B 436 257
- [713] Johannsen T 2009 Astron. Astrophys. 507 617
- [714] Salumbides E J, Schellekens A N, Gato-Rivera B and Ubachs W 2015 New J. Phys. 17 033015
- [715] Visinelli L, Bolis N and Vagnozzi S 2018 Phys. Rev. D 97 064039
- [716] Pardo K, Fishbach M, Holz D E and Spergel D N 2018 J. Cosmol. Astropart. Phys. JCAP07(2018)048
- [717] Chakraborty S and SenGupta S 2017 J. Cosmol. Astropart. Phys. JCAP07(2017)045
- [718] Chakravarti K, Chakraborty S, Phukon K S, Bose S and SenGupta S 2020 Class. Quantum Grav. 37 105004
- [719] Banerjee I, Chakraborty S and SenGupta S 2019 Phys. Rev. D 100 044045
- [720] Yu H, Lin Z-C and Liu Y-X 2019 Commun. Theor. Phys. 71 991
- [721] Dey R, Chakraborty S and Afshordi N 2020 Phys. Rev. D 101 104014
- [722] Trivedi O 2023 Int. J. Mod. Phys. D 32 2250130
- [723] Trivedi O 2022 Mod. Phys. Lett. A 37 2250162
- [724] Chakraborty S, Datta S and Sau S 2021 Phys. Rev. D 104 104001
- [725] Banerjee I, Chakraborty S and SenGupta S 2021 J. Cosmol. Astropart. Phys. JCAP09(2021)037
- [726] Mishra A K, Ghosh A and Chakraborty S 2022 Eur. Phys. J. C 82 820
- [727] Randall L and Sundrum R 1999 Phys. Rev. Lett. 83 4690
- [728] Dadhich N, Maartens R, Papadopoulos P and Rezania V 2000 Phys. Lett. B 487 1
- [729] Modgil M S, Panda S and Sengupta G 2002 Mod. Phys. Lett. A 17 1479
- [730] Aliev A N and Gumrukcuoglu A E 2005 Phys. Rev. D 71 104027
- [731] Schee J and Stuchlik Z 2009 Int. J. Mod. Phys. D 18 983
- [732] Aliev A N and Talazan P 2009 Phys. Rev. D 80 044023
- [733] Amarilla L and Eiroa E F 2012 Phys. Rev. D 85 064019
- [734] Aliev A N, Esmer G D and Talazan P 2013 Class. Quantum Grav. 30 045010
- [735] Neves J C S and Molina C 2012 Phys. Rev. D 86 124047
- [736] Eiroa E F and Sendra C M 2018 Eur. Phys. J. C 78 91
- [737] Abdujabbarov A, Ahmedov B, Dadhich N and Atamurotov F 2017 Phys. Rev. D 96 084017
- [738] Stuchlík Z, Charbulák D and Schee J 2018 Eur. Phys. J. C 78 180
- [739] Papnoi U, Atamurotov F, Ghosh S G and Ahmedov B 2014 Phys. Rev. D 90 024073
- [740] Abdujabbarov A, Atamurotov F, Dadhich N, Ahmedov B and Stuchlík Z 2015 Eur. Phys. J. C 75 399
- [741] Heisenberg W and Euler H 1936 Z. Phys. 98 714
- [742] Gross D J, Harvey J A, Martinec E J and Rohm R 1985 Phys. Rev. Lett. 54 502
- [743] Kalb M and Ramond P 1974 Phys. Rev. D 9 2273
- [744] Sen A 1992 Phys. Rev. Lett. 69 1006
- [745] Gibbons G W and Maeda K-I 1988 Nucl. Phys. B 298 741
- [746] Garfinkle D, Horowitz G T and Strominger A 1991 Phys. Rev. D 43 3140 Garfinkle D, Horowitz G T and Strominger A 1992 Phys. Rev. D 45 3888 (erratum)
- [747] Horowitz G T and Strominger A 1991 Nucl. Phys. B 360 197
- [748] Shapere A D, Trivedi S and Wilczek F 1991 Mod. Phys. Lett. A 6 2677
- [749] Gyulchev G N and Yazadjiev S S 2010 Phys. Rev. D 81 023005
- [750] Dastan S, Saffari R and Soroushfar S 2016 arXiv:1610.09477 [gr-qc]
- [751] Lin K, Qian W-L, Pavan A B and Abdalla E 2017 Mod. Phys. Lett. A 32 1750134
- [752] Guo M, Song S and Yan H 2020 *Phys. Rev.* D **101** 024055
- [753] Xavier S V M C B, Cunha P V P, Crispino L C B and Herdeiro C A R 2020 Int. J. Mod. Phys. D 29 2041005
- [754] Zhang M and Jiang J 2021 Nucl. Phys. B 964 115313
- [755] Zhang M and Jiang J 2020 Phys. Rev. D 102 124012
- [756] Belhaj A, Balali A E, El Hadri W, El Moumni H, Essebani M A and Sedra M B 2020 Int. J. Geom. Methods Mod. Phys. 17 2050169

- [757] Wu X and Zhang X 2022 Universe 8 604
- [758] Carleo A, Lambiase G and Mastrototaro L 2022 Eur. Phys. J. C 82 776
- [759] Tripathi A, Zhou B, Abdikamalov A B, Ayzenberg D and Bambi C 2021 J. Cosmol. Astropart. Phys. JCAP07(2021)002
- [760] Lessa L A, Silva J E G, Maluf R V and Almeida C A S 2020 Eur. Phys. J. C 80 335
- [761] Kostelecky V A and Samuel S 1989 Phys. Rev. D 39 683
- [762] Colladay D and Kostelecky V A 1998 Phys. Rev. D 58 116002
- [763] Kostelecky V A 2004 Phys. Rev. D 69 105009
- [764] Cruz W T, Maluf R V and Almeida C A S 2013 Eur. Phys. J. C 73 2523
- [765] Maluf R V, Araújo Filho A A, Cruz W T and Almeida C A S 2018 Europhys. Lett. 124 61001
- [766] Maluf R V and Neves J C S 2022 Eur. Phys. J. C 82 135
- [767] Atamurotov F, Ortiqboev D, Abdujabbarov A and Mustafa G 2022 Eur. Phys. J. C 82 659
- [768] Jusufi K and Stojkovic D 2022 Universe 8 194
- [769] Callan C G Jr, Myers R C and Perry M J 1989 Nucl. Phys. B 311 673
- [770] Garcia A, Galtsov D and Kechkin O 1995 Phys. Rev. Lett. 74 1276
- [771] Jacobson T and Mattingly D 2001 *Phys. Rev.* D 64 024028
- [772] Zlosnik T G, Ferreira P G and Starkman G D 2007 Phys. Rev. D 75 044017
- [773] Jacobson T 2007 *Proc. Sci.* **043** 020
- [774] Jacobson T and Mattingly D 2004 Phys. Rev. D 70 024003
- [775] Foster B Z and Jacobson T 2006 Phys. Rev. D 73 064015
- [776] Konoplya R A and Zhidenko A 2007 Phys. Lett. B 644 186
- [777] Jacobson T 2014 Phys. Rev. D 89 081501
- [778] Yagi K, Blas D, Barausse E and Yunes N 2014 *Phys. Rev.* D **89** 084067
  Yagi K, Blas D, Barausse E and Yunes N 2014 *Phys. Rev.* D **90** 069902 (erratum)
  Yagi K, Blas D, Barausse E and Yunes N 2014 *Phys. Rev.* D **90** 069901 (erratum)
- [779] Ding C, Wang A and Wang X 2015 Phys. Rev. D 92 084055
- [780] Gong Y, Hou S, Liang D and Papantonopoulos E 2018 Phys. Rev. D 97 084040
- [781] Leon G, Coley A and Paliathanasis A 2020 Ann. Phys., NY 412 168002
- [782] Paliathanasis A 2020 Phys. Rev. D 101 064008
- [783] Zhu Y and Wang T 2021 Phys. Rev. D 104 104052
- [784] Eling C and Jacobson T 2006 Class. Quantum Grav. 23 5643
   Eling C and Jacobson T 2010 Class. Quantum Grav. 27 049802 (erratum)
- [785] Barausse E, Jacobson T and Sotiriou T P 2011 Phys. Rev. D 83 124043
- [786] Oost J, Mukohyama S and Wang A 2018 Phys. Rev. D 97 124023
- [787] Atamurotov F, Abdujabbarov A and Ahmedov B 2013 Astrophys. Space Sci. 348 179
- [788] Horava P 2009 Phys. Rev. D 79 084008
- [789] Mukohyama S 2010 Class. Quantum Grav. 27 223101
- [790] Charmousis C, Niz G, Padilla A and Saffin P M 2009 J. High Energy Phys. JHEP08(2009)070
- [791] Nojiri S and Odintsov S D 2010 Phys. Rev. D 81 043001
- [792] Blas D, Pujolas O and Sibiryakov S 2009 J. High Energy Phys. JHEP10(2009)029
- [793] Blas D, Pujolas O and Sibiryakov S 2010 Phys. Rev. Lett. 104 181302
- [794] Papazoglou A and Sotiriou T P 2010 Phys. Lett. B 685 197
- [795] Henneaux M, Kleinschmidt A and Lucena Gómez G 2010 Phys. Rev. D 81 064002
- [796] Blas D, Pujolas O and Sibiryakov S 2010 Phys. Lett. B 688 350
- [797] Nojiri S and Odintsov S D 2010 Phys. Lett. B 691 60
- [798] Blas D, Pujolas O and Sibiryakov S 2011 J. High Energy Phys. JHEP04(2011)018
- [799] Cognola G, Myrzakulov R, Sebastiani L, Vagnozzi S and Zerbini S 2016 Class. Quantum Grav. 33 225014
- [800] Jacobson T 2010 Phys. Rev. D 81 101502
   Jacobson T 2010 Phys. Rev. D 82 129901 (erratum)
- [801] Cai R-G, Cao L-M and Ohta N 2009 Phys. Rev. D 80 024003
- [802] Lim E A, Sawicki I and Vikman A 2010 J. Cosmol. Astropart. Phys. JCAP05(2010)012
- [803] Chamseddine A H and Mukhanov V 2013 J. High Energy Phys. JHEP11(2013)135
- [804] Chamseddine A H, Mukhanov V and Vikman A 2014 J. Cosmol. Astropart. Phys. JCAP06(2014)017
- [805] Ramazanov S, Arroja F, Celoria M, Matarrese S and Pilo L 2016 J. High Energy Phys. JHEP06(2016)020
- [806] Myrzakulov R, Sebastiani L, Vagnozzi S and Zerbini S 2015 Fundam. J. Mod. Phys. 8 119

- [807] Chamseddine A H, Mukhanov V and Russ T B 2019 Phys. Lett. B 798 134939
- [808] Momeni D, Altaibayeva A and Myrzakulov R 2014 Int. J. Geom. Methods Mod. Phys. 11 1450091
- [809] Nojiri S and Odintsov S D 2014 Mod. Phys. Lett. A 29 1450211 (erratum)
- [810] Leon G and Saridakis E N 2015 J. Cosmol. Astropart. Phys. JCAP04(2015)031
- [811] Arroja F, Bartolo N, Karmakar P and Matarrese S 2015 J. Cosmol. Astropart. Phys. JCAP09(2015)051
- [812] Rabochaya Y and Zerbini S 2016 Eur. Phys. J. C 76 85
- [813] Odintsov S D and Oikonomou V K 2016 Astrophys. Space Sci. 361 174
- [814] Arroja F, Bartolo N, Karmakar P and Matarrese S 2016 J. Cosmol. Astropart. Phys. JCAP04(2016)042
- [815] Nojiri S, Odintsov S D and Oikonomou V K 2016 Class. Quantum Grav. 33 125017
- [816] Nojiri S, Odintsov S D and Oikonomou V K 2016 Phys. Rev. D 94 104050
- [817] Firouzjahi H, Gorji M A and Hosseini Mansoori S A 2017 J. Cosmol. Astropart. Phys. JCAP07(2017)031
- [818] Zheng Y, Shen L, Mou Y and Li M 2017 J. Cosmol. Astropart. Phys. JCAP08(2017)040
- [819] Takahashi K and Kobayashi T 2017 J. Cosmol. Astropart. Phys. JCAP11(2017)038
- [820] Paston S A 2017 Phys. Rev. D 96 084059
- [821] Dutta J, Khyllep W, Saridakis E N, Tamanini N and Vagnozzi S 2018 J. Cosmol. Astropart. Phys. JCAP02(2018)041
- [822] Gorji M A, Mukohyama S, Firouzjahi H and Hosseini Mansoori S A 2018 J. Cosmol. Astropart. Phys. JCAP08(2018)047
- [823] Ganz A, Karmakar P, Matarrese S and Sorokin D 2019 Phys. Rev. D 99 064009
- [824] Gorji M A, Mukohyama S and Firouzjahi H 2019 J. Cosmol. Astropart. Phys. JCAP05(2019)019
- [825] Khalifeh A R, Bellomo N, Bernal J L and Jimenez R 2020 Phys. Dark Universe 30 100646
- [826] Ganz A, Bartolo N and Matarrese S 2019 J. Cosmol. Astropart. Phys. JCAP12(2019)037
- [827] Sur S, Dutta A and Ramo Chothe H 2021 Eur. Phys. J. C 81 306
- [828] Izaurieta F, Medina P, Merino N, Salgado P and Valdivia O 2020 J. High Energy Phys. JHEP10(2020)150
- [829] de Cesare M and Husain V 2020 Phys. Rev. D 102 104052
- [830] Benisty D, Chaichian M and Oksanen M 2021 arXiv:2107.12161 [gr-qc]
- [831] Mansoori S A H, Talebian A, Molaee Z and Firouzjahi H 2022 Phys. Rev. D 105 023529
- [832] Nashed G G L and Saridakis E N 2022 arXiv:2206.12256 [gr-qc]
- [833] Jiroušek P, Shimada K, Vikman A and Yamaguchi M 2022 J. Cosmol. Astropart. Phys. JCAP11(2022)019
- [834] Sebastiani L, Vagnozzi S and Myrzakulov R 2017 Adv. High Energy Phys. 2017 3156915
- [835] Deruelle N and Rua J 2014 J. Cosmol. Astropart. Phys. JCAP09(2014)002
- [836] Myrzakulov R and Sebastiani L 2015 Gen. Relativ. Gravit. 47 89
- [837] Myrzakulov R, Sebastiani L, Vagnozzi S and Zerbini S 2016 Class. Quantum Grav. 33 125005
- [838] Chen C-Y, Bouhmadi-López M and Chen P 2018 Eur. Phys. J. C 78 59
- [839] Nashed G G L, El Hanafy W and Bamba K 2019 J. Cosmol. Astropart. Phys. JCAP01(2019)058
- [840] Sheykhi A and Grunau S 2021 Int. J. Mod. Phys. A 36 2150186
- [841] Gorji M A, Allahyari A, Khodadi M and Firouzjahi H 2020 Phys. Rev. D 101 124060
- [842] Sheykhi A 2020 J. High Energy Phys. JHEP07(2020)031
- [843] Casalino A, Sebastiani L and Zerbini S 2020 Phys. Rev. D 101 104059
- [844] Nashed G G L and Nojiri S 2021 Phys. Rev. D 104 044043
- [845] Nashed G G L and Nojiri S 2022 J. Cosmol. Astropart. Phys. JCAP05(2022)011
- [846] Nojiri S and Nashed G G L 2022 Phys. Lett. B 830 137140
- [847] Lovelock D 1971 J. Math. Phys. 12 498
- [848] Glavan D and Lin C 2020 Phys. Rev. Lett. 124 081301
- [849] Casalino A, Colleaux A, Rinaldi M and Vicentini S 2021 Phys. Dark Universe 31 100770
- [850] Fernandes P G S, Carrilho P, Clifton T and Mulryne D J 2022 Class. Quantum Grav. 39 063001
- [851] Lu H and Pang Y 2020 Phys. Lett. B 809 135717
- [852] Kobayashi T 2020 J. Cosmol. Astropart. Phys. JCAP07(2020)013
- [853] Ai W-Y 2020 Commun. Theor. Phys. 72 095402
- [854] Arrechea J, Delhom A and Jiménez-Cano A 2021 Chin. Phys. C 45 013107
- [855] Gürses M, Şişman T C and Tekin B 2020 Eur. Phys. J. C 80 647
- [856] Fernandes P G S, Carrilho P, Clifton T and Mulryne D J 2020 Phys. Rev. D 102 024025
- [857] Hennigar R A, Kubizňák D, Mann R B and Pollack C 2020 J. High Energy Phys. JHEP07(2020)027
- [858] Mahapatra S 2020 *Eur. Phys. J.* C 80 992
- [859] Aoki K, Gorji M A and Mukohyama S 2020 Phys. Lett. B 810 135843
- [860] Gurses M, Şişman T c and Tekin B 2020 Phys. Rev. Lett. 125 149001
- [861] Doneva D D and Yazadjiev S S 2021 J. Cosmol. Astropart. Phys. JCAP05(2021)024
- [862] Eslam Panah B, Jafarzade K and Hendi S H 2020 Nucl. Phys. B 961 115269
- [863] Aragón A, Bécar R, González P A and Vásquez Y 2020 Eur. Phys. J. C 80 773
- [864] Malafarina D, Toshmatov B and Dadhich N 2020 Phys. Dark Universe 30 100598
- [865] Yang S-J, Wan J-J, Chen J, Yang J and Wang Y-Q 2020 Eur. Phys. J. C 80 937
- [866] Shu F-W 2020 Phys. Lett. B 811 135907
- [867] Jusufi K, Banerjee A and Ghosh S G 2020 Eur. Phys. J. C 80 698
- [868] Yang K, Gu B-M, Wei S-W and Liu Y-X 2020 Eur. Phys. J. C 80 662
- [869] Chakraborty S and Dadhich N 2020 Phys. Dark Universe 30 100658
- [870] Banerjee A, Tangphati T, Samart D and Channuie P 2021 Astrophys. J. 906 114
- [871] Wei S-W, Liu Y-X and Wang Y-Q 2022 *Nucl. Phys.* B **976** 115692
- [872] Hohmann M, Pfeifer C and Voicu N 2021 Eur. Phys. J. Plus 136 180
- [873] Jafarzade K, Kord Zangeneh M and Lobo F S N 2022 *Universe* 8 182
- [874] Aoki K, Gorji M A, Mizuno S and Mukohyama S 2021 J. Cosmol. Astropart. Phys. JCAP01(2021)054
- [875] Wang D and Mota D 2021 Phys. Dark Universe 32 100813
- [876] Atamurotov F, Shaymatov S, Sheoran P and Siwach S 2021 J. Cosmol. Astropart. Phys. JCAP08(2021)045
- [877] Övgün A 2021 Phys. Lett. B 820 136517
- [878] Gyulchev G, Nedkova P, Vetsov T and Yazadjiev S 2021 Eur. Phys. J. C 81 885
- [879] Nashed G G L, Odintsov S D and Oikonomou V K 2022 Symmetry 14 545
- [880] Donmez O, Dogan F and Sahin T 2022 Universe 8 458
- [881] Konoplya R A and Zhidenko A 2020 Phys. Rev. D 101 084038
- [882] Kumar A, Walia R K and Ghosh S G 2022 Universe 8 232
- [883] Hosseini Mansoori S A 2021 Phys. Dark Universe 31 100776
- [884] Kumar A, Baboolal D and Ghosh S G 2022 Universe 8 244
- [885] Kumar R, Islam S U and Ghosh S G 2020 Eur. Phys. J. C 80 1128
- [886] El Moumni H, Masmar K and Övgün A 2022 Int. J. Geom. Methods Mod. Phys. 19 2250094
- [887] Zahid M, Khan S U and Ren J 2021 Chin. J. Phys. 72 575
- [888] Donmez O 2022 Phys. Lett. B 827 136997
- [889] Narzilloev B, Shaymatov S, Hussain I, Abdujabbarov A, Ahmedov B and Bambi C 2021 Eur. Phys. J. C 81 849
- [890] Papnoi U and Atamurotov F 2022 Phys. Dark Universe 35 100916
- [891] Clifton T, Carrilho P, Fernandes P G S and Mulryne D J 2020 *Phys. Rev. D* 102 084005
  [892] Weinberg S 1980 Ultraviolet divergences in quantum theories of gravitation *General Relativity:*
- An Einstein Centenary Survey ed S W Hawking and W Israel (Cambridge: Cambridge University Press) pp 790–831
- [893] Reuter M 1998 Phys. Rev. D 57 971
- [894] Koch B and Saueressig F 2014 Int. J. Mod. Phys. A 29 1430011
- [895] Bonanno A and Saueressig F 2017 C. R. Physique 18 254
- [896] Platania A 2020 Front. Phys. 8 188
- [897] Bonanno A, Eichhorn A, Gies H, Pawlowski J M, Percacci R, Reuter M, Saueressig F and Vacca G P 2020 Front. Phys. 8 269
- [898] Shaposhnikov M and Wetterich C 2010 Phys. Lett. B 683 196
- [899] Eichhorn A and Gies H 2010 Phys. Rev. D 81 104010
- [900] Bonanno A, Contillo A and Percacci R 2011 Class. Quantum Grav. 28 145026
- [901] Bonanno A and Reuter M 2011 Entropy 13 274
- [902] Bonanno A 2012 Phys. Rev. D 85 081503
- [903] Eichhorn A 2012 Phys. Rev. D 86 105021
- [904] Litim D F and Sannino F 2014 J. High Energy Phys. JHEP12(2014)178
- [905] Litim D F, Mojaza M and Sannino F 2016 J. High Energy Phys. JHEP01(2016)081
- [906] Bonanno A and Platania A 2015 Phys. Lett. B 750 638
- [907] Meibohm J, Pawlowski J M and Reichert M 2016 Phys. Rev. D 93 084035

- [908] Donà P, Eichhorn A, Labus P and Percacci R 2016 *Phys. Rev.* D **93** 044049
- Donà P, Eichhorn A, Labus P and Percacci R 2016 *Phys. Rev.* D 93 129904 (erratum)
- [909] Biemans J, Platania A and Saueressig F 2017 Phys. Rev. D 95 086013
- [910] Mann R, Meffe J, Sannino F, Steele T, Wang Z-W and Zhang C 2017 Phys. Rev. Lett. 119 261802
- [911] Eichhorn A, Lippoldt S and Skrinjar V 2018 Phys. Rev. D 97 026002
- [912] Eichhorn A, Held A and Vander Griend P 2018 J. High Energy Phys. JHEP08(2018)147
- [913] Bonanno A, Platania A and Saueressig F 2018 Phys. Lett. B 784 229
- [914] Eichhorn A and Held A 2018 Phys. Rev. Lett. 121 151302
- [915] Adeifeoba A, Eichhorn A and Platania A 2018 Class. Quantum Grav. 35 225007
- [916] de Alwis S, Eichhorn A, Held A, Pawlowski J M, Schiffer M and Versteegen F 2019 Phys. Lett. B 798 134991
- [917] Draper T, Knorr B, Ripken C and Saueressig F 2020 J. High Energy Phys. JHEP11(2020)136
- [918] Platania A and Wetterich C 2020 Phys. Lett. B 811 135911
- [919] Basile I and Platania A 2021 Universe 7 389
- [920] Eichhorn A, Pauly M and Ray S 2021 J. High Energy Phys. JHEP10(2021)100
- [921] Fehre J, Litim D F, Pawlowski J M and Reichert M 2023 Phys. Rev. Lett. 130 081501
- [922] de Brito G P, Eichhorn A and Lino dos Santos R R 2020 J. High Energy Phys. JHEP22(2020)013
- [923] Trivedi O and Khlopov M 2022 J. Cosmol. Astropart. Phys. JCAP11(2022)007
- [924] Lambiase G and Scardigli F 2022 Phys. Rev. D 105 124054
- [925] Scardigli F and Lambiase G 2022 arXiv:2205.07088 [gr-qc]
- [926] Knorr B, Pirlo S, Ripken C and Saueressig F 2022 arXiv:2205.01738 [hep-th]
- [927] Platania A 2022 J. High Energy Phys. JHEP09(2022)167
- [928] Fraaije M, Platania A and Saueressig F 2022 Phys. Lett. B 834 137399
- [929] Platania A 2023 arXiv:2302.04272 [gr-qc]
- [930] Bonanno A and Reuter M 2000 Phys. Rev. D 62 043008
- [931] Donoghue J F 1994 Phys. Rev. Lett. 72 2996
- [932] Reuter M and Tuiran E 2011 Phys. Rev. D 83 044041
- [933] Becker D and Reuter M 2012 J. High Energy Phys. JHEP07(2012)172
- [934] Falls K and Litim D F 2014 *Phys. Rev.* D 89 084002
- [935] Koch B, Contreras C, Rioseco P and Saueressig F 2016 1st Karl Schwarzschild Meeting on Gravitational Physics (Springer Proc. Physics vol 170) p 263
- [936] Saueressig F, Alkofer N, D'Odorico G and Vidotto F 2016 Proc. Sci. 224 174
- [937] González C and Koch B 2016 Int. J. Mod. Phys. A 31 1650141
- [938] Bonanno A, Koch B and Platania A 2017 Class. Quantum Grav. 34 095012
- [939] Bonanno A, Koch B and Platania A 2018 Found. Phys. 48 1393
- [940] Platania A 2019 Eur. Phys. J. C 79 470
- [941] Rincón A and Panotopoulos G 2020 Phys. Dark Universe 30 100639
- [942] Eichhorn A and Held A 2021 Eur. Phys. J. C 81 933
- [943] Eichhorn A and Held A 2021 J. Cosmol. Astropart. Phys. JCAP05(2021)073
- [944] Knorr B and Platania A 2022 Phys. Rev. D 106 L021901
- [945] Delaporte H, Eichhorn A and Held A 2022 Class. Quantum Grav. 39 134002
- [946] Eichhorn A, Held A and Johannsen P-V 2023 J. Cosmol. Astropart. Phys. JCAP01(2023)043
- [947] Eichhorn A and Held A 2023 J. Cosmol. Astropart. Phys. JCAP01(2023)032
- [948] Kumar R, Singh B P and Ghosh S G 2020 Ann. Phys., NY **420** 168252
- [949] Zhou B, Abdikamalov A B, Ayzenberg D, Bambi C, Nampalliwar S and Tripathi A 2021 J. Cosmol. Astropart. Phys. JCAP01(2021)047
- [950] Rastall P 1972 Phys. Rev. D 6 3357
- [951] Fabris J C, Daouda M H and Piattella O F 2012 Phys. Lett. B 711 232
- [952] Batista C E M, Daouda M H, Fabris J C, Piattella O F and Rodrigues D C 2012 Phys. Rev. D 85 084008
- [953] Moradpour H, Sadeghnezhad N and Hendi S H 2017 Can. J. Phys. 95 1257
- [954] Moradpour H and Salako I G 2016 Adv. High Energy Phys. 2016 3492796
- [955] Moradpour H, Heydarzade Y, Darabi F and Salako I G 2017 Eur. Phys. J. C 77 259
- [956] Xu Z, Hou X, Gong X and Wang J 2018 Eur. Phys. J. C 78 513
- [957] Darabi F, Moradpour H, Licata I, Heydarzade Y and Corda C 2018 Eur. Phys. J. C 78 25
- [958] Hansraj S, Banerjee A and Channuie P 2019 Ann. Phys., NY 400 320
- [959] Cruz M, Lepe S and Morales-Navarrete G 2019 Class. Quantum Grav. 36 225007
- [960] Khyllep W and Dutta J 2019 Phys. Lett. B 797 134796

- [961] Pourhassan B and Upadhyay S 2021 Eur. Phys. J. Plus 136 311
- [962] Akarsu O, Katırcı N, Kumar S, Nunes R C, Öztürk B and Sharma S 2020 Eur. Phys. J. C 80 1050
- [963] Lobo I P, Richarte M G, Morais Graca J P and Moradpour H 2020 Eur. Phys. J. Plus 135 550
- [964] Ghosh S, Dey S, Das A, Chanda A and Paul B C 2021 J. Cosmol. Astropart. Phys. JCAP07(2021)004
- [965] Gogoi D J and Goswami U D 2021 Phys. Dark Universe 33 100860
- [966] Shahidi S 2021 Phys. Rev. D 104 084033
- [967] Narzilloev B, Hussain I, Abdujabbarov A, Ahmedov B and Bambi C 2021 Eur. Phys. J. Plus 136 1032
- [968] Guo S, He K-J, Li G-R and Li G-P 2021 Class. Quantum Grav. 38 165013
- [969] Zhong Y, Yang K and Liu Y-X 2022 J. High Energy Phys. JHEP09(2022)128
- [970] Visser M 2018 Phys. Lett. B 782 83
- [971] Heydarzade Y, Moradpour H and Darabi F 2017 Can. J. Phys. 95 1253
- [972] Heydarzade Y and Darabi F 2017 Phys. Lett. B 771 365
- [973] Kumar R and Ghosh S G 2018 Eur. Phys. J. C 78 750
- [974] Ashtekar A and Lewandowski J 2004 Class. Quantum Grav. 21 R53
- [975] Rovelli C 2004 Quantum Gravity (Cambridge Monographs on Mathematical Physics) (Cambridge: Cambridge University Press)
- [976] Thiemann T 2007 Modern Canonical Quantum General Relativity (Cambridge Monographs on Mathematical Physics (Cambridge: Cambridge University Press)
- [977] Rovelli C and Vidotto F 2014 Covariant Loop Quantum Gravity: An Elementary Introduction to Quantum Gravity and Spinfoam Theory (Cambridge Monographs on Mathematical Physics (Cambridge: Cambridge University Press)
- [978] Ashtekar A and Bianchi E 2021 Rep. Prog. Phys. 84 042001
- [979] Bojowald M 2001 Phys. Rev. Lett. 86 5227
- [980] Bojowald M 2005 Living Rev. Relativ. 8 11
- [981] Bojowald M 2007 Phys. Rev. D 75 081301
- [982] Modesto L 2008 Int. J. Theor. Phys. 47 357
- [983] Engle J, Livine E, Pereira R and Rovelli C 2008 Nucl. Phys. B 799 136
- [984] Rovelli C and Vidotto F 2008 Class. Quantum Grav. 25 225024
- [985] Modesto L 2009 Class. Quantum Grav. 26 242002
- [986] Bianchi E, Dona P and Speziale S 2011 Phys. Rev. D 83 044035
- [987] Henderson A, Rovelli C, Vidotto F and Wilson-Ewing E 2011 Class. Quantum Grav. 28 025003
- [988] Singh P and Vidotto F 2011 Phys. Rev. D 83 064027
- [989] Bojowald M and Paily G M 2012 Phys. Rev. D 86 104018
- [990] Cailleteau T, Barrau A, Vidotto F and Grain J 2012 Phys. Rev. D 86 087301
- [991] Agullo I, Ashtekar A and Nelson W 2013 Class. Quantum Grav. 30 085014
- [992] Rovelli C and Vidotto F 2013 Phys. Rev. Lett. 111 091303
- [993] Haggard H M and Rovelli C 2015 Phys. Rev. D 92 104020
- [994] Odintsov S D and Oikonomou V K 2014 Phys. Rev. D 90 124083
- [995] Odintsov S D, Oikonomou V K and Saridakis E N 2015 Ann. Phys., NY 363 141
- [996] Haro J, Makarenko A N, Myagky A N, Odintsov S D and Oikonomou V K 2015 Phys. Rev. D 92 124026
- [997] Odintsov S D and Oikonomou V K 2016 Europhys. Lett. 116 49001
- [998] Bianchi E, Christodoulou M, D'Ambrosio F, Haggard H M and Rovelli C 2018 Class. Quantum Grav. 35 225003
- [999] De Haro J, Odintsov S D and Oikonomou V K 2018 Phys. Rev. D 97 084052
- [1000] de Haro J, Aresté Saló L and Pan S 2019 Gen. Relativ. Gravit. 51 49
- [1001] Odintsov S D and Oikonomou V K 2018 Phys. Rev. D 97 124042
- [1002] Benetti M, Graef L and Ramos R O 2019 J. Cosmol. Astropart. Phys. JCAP10(2019)066
- [1003] Casalino A, Sebastiani L, Vanzo L and Zerbini S 2020 Phys. Dark Universe 29 100594
- [1004] Bouhmadi-López M, Brahma S, Chen C-Y, Chen P and Yeom D-H 2020 J. Cosmol. Astropart. Phys. JCAP07(2020)066
- [1005] Graef L L, Ramos R O and Vicente G S 2020 Phys. Rev. D 102 043518
- [1006] Brahma S, Chen C-Y and Yeom D-H 2021 Phys. Rev. Lett. 126 181301
- [1007] Barboza L N, Levy G L L W, Graef L L and Ramos R O 2022 Phys. Rev. D 106 103535
- [1008] Vicente G S, Ramos R O and Graef L L 2022 Phys. Rev. D 106 043518
- [1009] Rovelli C 1998 Living Rev. Relativ. 1 1

- [1010] Modesto L 2006 Class. Quantum Grav. 23 5587
- [1011] Modesto L 2008 Adv. High Energy Phys. 2008 459290
- [1012] Caravelli F and Modesto L 2010 Class. Quantum Grav. 27 245022
- [1013] Brown E G, Mann R B and Modesto L 2011 Phys. Rev. D 84 104041
- [1014] Modesto L 2010 Int. J. Theor. Phys. 49 1649
- [1015] Immirzi G 1997 Class. Quantum Grav. 14 L177
- [1016] Immirzi G 1997 Nucl. Phys. B 57 65
- [1017] Perez A and Rovelli C 2006 Phys. Rev. D 73 044013
- [1018] Ashtekar A, Lewandowski J and Sahlmann H 2003 Class. Quantum Grav. 20 L11
- [1019] McVittie G C 1933 Mon. Not. R. Astron. Soc. 93 325
- [1020] Einstein A and Straus E G 1945 Rev. Mod. Phys. 17 120
- [1021] Einstein A and Straus E G 1946 Rev. Mod. Phys. 18 148
- [1022] Lake K and Roeder R C 1977 Phys. Rev. D 15 3513
- [1023] Stuchlik Z 1983 Bull. Astron. Inst. Czech. 34 129
- [1024] Nolan B C 1998 Phys. Rev. D 58 064006
- [1025] Nolan B C 1999 Class. Quantum Grav. 16 1227
- [1026] Nolan B C 1999 Class. Quantum Grav. 16 3183
- [1027] Stuchlik Z and Hledik S 1999 *Phys. Rev.* D 60 044006
- [1028] Gibbons G W and Maeda K-I 2010 Phys. Rev. Lett. 104 131101
- [1029] Nandra R, Lasenby A N and Hobson M P 2012 Mon. Not. R. Astron. Soc. 422 2931
- [1030] Nandra R, Lasenby A N and Hobson M P 2012 Mon. Not. R. Astron. Soc. 422 2945
- [1031] da Silva A M, Fontanini M and Guariento D C 2013 Phys. Rev. D 87 064030
- [1032] Piattella O F 2016 Phys. Rev. D 93 024020
- Piattella O F 2016 Phys. Rev. D 93 129901 (erratum)
- [1033] Piattella O F 2016 Universe 2 25
- [1034] Perlick V, Tsupko O Y and Bisnovatyi-Kogan G S 2018 Phys. Rev. D 97 104062
- [1035] Bisnovatyi-Kogan G S and Tsupko O Y 2018 Phys. Rev. D 98 084020
- [1036] Firouzjaee J T and Allahyari A 2019 Eur. Phys. J. C 79 930
- [1037] Tsupko O Y, Fan Z and Bisnovatyi-Kogan G S 2020 Class. Quantum Grav. 37 065016
- [1038] Tsupko O Y and Bisnovatyi-Kogan G S 2020 Int. J. Mod. Phys. D 29 2050062
- [1039] Roy R and Chakrabarti S 2020 Phys. Rev. D 102 024059
- [1040] Belhaj A, Benali M, Balali A E, Hadri W E and El Moumni H 2021 Int. J. Geom. Methods Mod. Phys. 18 2150188
- [1041] Chang Z and Zhu Q-H 2021 J. Cosmol. Astropart. Phys. JCAP09(2021)003
- [1042] Renzi F and Martinelli M 2022 Phys. Dark Universe 37 101104
- [1043] Odintsov S D and Oikonomou V K 2022 Europhys. Lett. 139 59003
- [1044] Adler S L 2023 Gen. Relativ. Gravit. 55 1
- [1045] Adler S L and Virbhadra K S 2022 Gen. Relativ. Gravit. 54 93
- [1046] Weyl H 1918 Math. Z. 2 384
- [1047] Drummond I T and Hathrell S J 1980 Phys. Rev. D 22 343
- [1048] Ritz A and Ward J 2009 Phys. Rev. D 79 066003
- [1049] Cho H T 1997 Phys. Rev. D 56 6416
- [1050] Cai R-G 1998 Nucl. Phys. B 524 639
- [1051] Zhang Z, Chen S, Qin X and Jing J 2021 Eur. Phys. J. C 81 991
- [1052] Chen S and Jing J 2015 J. Cosmol. Astropart. Phys. JCAP10(2015)002
- [1053] Lu X, Yang F-W and Xie Y 2016 Eur. Phys. J. C 76 357
- [1054] Huang Y, Chen S and Jing J 2016 Eur. Phys. J. C 76 594
- [1055] Chen S, Wang S, Huang Y, Jing J and Wang S 2017 Phys. Rev. D 95 104017
- [1056] Zhang R and Jing J 2017 arXiv:1703.08758 [gr-qc]
- [1057] Jing J, Chen S, Pan Q and Wang J 2017 arXiv:1704.08794 [gr-qc]
- [1058] Ovgün A, Jusufi K and Sakalli I 2018 Ann. Phys., NY 399 193
- [1059] Zhang R and Jing J 2018 Eur. Phys. J. C 78 796
- [1060] Zhang H-X, Li C, He P-Z, Fan Q-Q and Deng J-B 2020 Eur. Phys. J. C 80 461
- [1061] Bergliaffa S E P, Filho E E d S and Maier R 2020 Phys. Rev. D 101 124038
- [1062] Lu X and Xie Y 2020 Eur. Phys. J. C 80 625
- [1063] Li G and Deng X-M 2017 Ann. Phys., NY 382 136
- [1064] Li G and Deng X-M 2018 Commun. Theor. Phys. 70 721
- [1065] Deser S, Sarioglu O and Tekin B 2008 Gen. Relativ. Gravit. 40 1

- [1066] Bellini E, Di Criscienzo R, Sebastiani L and Zerbini S 2010 Entropy 12 2186
- [1067] Cognola G, Gorbunova O, Sebastiani L and Zerbini S 2011 Phys. Rev. D 84 023515
- [1068] Myrzakulov R, Sebastiani L and Zerbini S 2013 Int. J. Mod. Phys. D 22 1330017
- [1069] Colléaux A and Zerbini s 2015 Entropy 17 6643
- [1070] Chinaglia S, Colléaux A and Zerbini S 2017 Galaxies 5 51
- [1071] Sebastiani L, Vanzo L and Zerbini S 2018 Phys. Rev. D 97 044009
- [1072] Colléaux A, Chinaglia S and Zerbini S 2018 Int. J. Mod. Phys. D 27 1830002
- [1073] Chinaglia S 2018 arXiv:1805.03899 [gr-qc]
- [1074] Vetsov T 2019 Eur. Phys. J. C 79 71
- [1075] González P A, Olivares M, Vásquez Y, Saavedra J and Övgün A 2019 *Eur. Phys. J.* C **79** 528 [1076] Grumiller D 2010 *Phys. Rev. Lett.* **105** 211303
- Grumiller D 2011 Phys. Rev. Lett. 106 039901 (erratum)
- [1077] Grumiller D and Preis F 2011 Int. J. Mod. Phys. D 20 2761
- [1078] Anderson J D, Campbell J K, Ekelund J E, Ellis J and Jordan J F 2008 Phys. Rev. Lett. 100 091102
- [1079] Anderson J D, Laing P A, Lau E L, Liu A S, Nieto M M and Turyshev S G 1998 Phys. Rev. Lett. 81 2858
- [1080] Pardo K and Spergel D N 2020 Phys. Rev. Lett. 125 211101
- [1081] Turyshev S G, Toth V T, Kinsella G, Lee S-C, Lok S M and Ellis J 2012 Phys. Rev. Lett. 108 241101
- [1082] Grumiller D, Kummer W and Vassilevich D V 2002 Phys. Rep. 369 327
- [1083] Sakalli I and Ovgun A 2017 Europhys. Lett. 118 60006
- [1084] Perivolaropoulos L and Skara F 2019 Phys. Rev. D 99 124006
- [1085] Riegert R J 1984 Phys. Rev. Lett. 53 315
- [1086] Mannheim P D and Kazanas D 1989 Astrophys. J. 342 635
- [1087] Klemm D 1998 Class. Quantum Grav. 15 3195
- [1088] Saffari R and Rahvar S 2008 Phys. Rev. D 77 104028
- [1089] de Rham C, Gabadadze G and Tolley A J 2011 Phys. Rev. Lett. 106 231101
- [1090] Soroushfar S, Saffari R, Kunz J and Lämmerzahl C 2015 Phys. Rev. D 92 044010
- [1091] Gregoris D, Ong Y C and Wang B 2021 Eur. Phys. J. C 81 684
- [1092] Kiselev V V 2003 Class. Quantum Grav. 20 1187
- [1093] Milgrom M 1983 Astrophys. J. 270 365
- [1094] Milgrom M 1983 Astrophys. J. 270 371
- [1095] Bertolami O, Boehmer C G, Harko T and Lobo F S N 2007 Phys. Rev. D 75 104016
- [1096] Mannheim P D and O'Brien J G 2011 Phys. Rev. Lett. 106 121101
- [1097] Vagnozzi S 2017 Class. Quantum Grav. 34 185006
- [1098] Chen S-b and Jing J-l 2005 Class. Quantum Grav. 22 4651
- [1099] Chen S, Wang B and Su R 2008 Phys. Rev. D 77 124011
- [1100] Zeng X-X, Chen D-Y and Li L-F 2015 Phys. Rev. D 91 046005
- [1101] Azreg-Aïnou M 2015 Phys. Rev. D 91 064049
- [1102] Abdujabbarov A, Toshmatov B, Stuchlík Z and Ahmedov B 2016 Int. J. Mod. Phys. D 26 1750051
- [1103] Ghosh S G 2016 Eur. Phys. J. C 76 222
- [1104] Toshmatov B, Stuchlík Z and Ahmedov B 2017 Eur. Phys. J. Plus 132 98
- [1105] de Oliveira J and Fontana R D B 2018 Phys. Rev. D 98 044005
- [1106] Benavides-Gallego C A, Abdujabbarov A A and Bambi C 2020 Phys. Rev. D 101 044038
- [1107] Saadati R and Shojai F 2019 Phys. Rev. D 100 104041
- [1108] Ali M S, Ahmed F and Ghosh S G 2020 Ann. Phys., NY 412 168024
- [1109] Belhaj A, Benali M, El Balali A, El Moumni H and Ennadifi S E 2020 Class. Quantum Grav. 37 215004
- [1110] Sheoran P, Nandan H, Hackmann E, Nucamendi U and Abebe A 2020 Phys. Rev. D 102 064046
- [1111] Zeng X-X and Zhang H-Q 2020 Eur. Phys. J. C 80 1058
- [1112] Lambiase G and Mastrototaro L 2021 Eur. Phys. J. C 81 932
- [1113] Hendi S H and Jafarzade K 2021 Phys. Rev. D 103 104011
- [1114] Mastrototaro L and Lambiase G 2021 Phys. Rev. D 104 024021
- [1115] Cárdenas V H, Fathi M, Olivares M and Villanueva J R 2021 Eur. Phys. J. C 81 866
- [1116] Ndongmo R B, Mahamat S, Bouetou T B, Tabi C B and Kofane T C 2021 arXiv:2111.05045 [gr-qc]
- [1117] Sun C, Liu Y, Qian W-L and Yue R 2022 Chin. Phys. C 46 065103

- [1118] Kiselev V V 2003 arXiv:gr-qc/0303031
- [1119] Rahaman F, Nandi K K, Bhadra A, Kalam M and Chakraborty K 2011 Phys. Lett. B 694 10
- [1120] Li M-H and Yang K-C 2012 Phys. Rev. D 86 123015
- [1121] Visser M 2020 Class. Quantum Grav. 37 045001
- [1122] Xu Z, Hou X, Wang J and Liao Y 2019 Adv. High Energy Phys. 2019 2434390
- [1123] Xu Z, Hou X and Wang J 2018 Class. Quantum Grav. 35 115003
- [1124] Bahamonde S, Bamba K and Camci U 2019 J. Cosmol. Astropart. Phys. JCAP02(2019)016
- [1125] Haroon S, Jamil M, Jusufi K, Lin K and Mann R B 2019 Phys. Rev. D 99 044015
- [1126] Hou X, Xu Z and Wang J 2018 J. Cosmol. Astropart. Phys. JCAP12(2018)040
- [1127] Rizwan M, Jamil M and Jusufi K 2019 Phys. Rev. D 99 024050
- [1128] Hendi S H, Nemati A, Lin K and Jamil M 2020 Eur. Phys. J. C 80 296
- [1129] Shaymatov S, Malafarina D and Ahmedov B 2021 Phys. Dark Universe 34 100891
- [1130] Shaymatov S, Ahmedov B and Jamil M 2021 *Eur. Phys. J.* C 81 588 Shaymatov S, Ahmedov B and Jamil M 2021 *Eur. Phys. J.* C 81 724 (erratum)
- [1131] Zhang H-X, Chen Y, Ma T-C, He P-Z and Deng J-B 2021 Chin. Phys. C 45 055103
- [1132] Das A, Saha A and Gangopadhyay S 2021 Class. Quantum Grav. 38 065015
- [1133] Saurabh K and Jusufi K 2021 Eur. Phys. J. C 81 490
- [1134] Ma T-C, Zhang H-X, He P-Z, Zhang H-R, Chen Y and Deng J-B 2021 Mod. Phys. Lett. A 36 2150112
- [1135] Narzilloev B, Rayimbaev J, Shaymatov S, Abdujabbarov A, Ahmedov B and Bambi C 2020 Phys. Rev. D 102 104062
- [1136] Cao Y, Feng H, Hong W and Tao J 2021 Commun. Theor. Phys. 73 095403
- [1137] Atamurotov F, Papnoi U and Jusufi K 2022 Class. Quantum Grav. 39 025014
- [1138] Rayimbaev J, Shaymatov S and Jamil M 2021 Eur. Phys. J. C 81 699
- [1139] Atamurotov F, Abdujabbarov A and Han W-B 2021 Phys. Rev. D 104 084015
- [1140] Shaymatov S, Sheoran P and Siwach S 2022 Phys. Rev. D 105 104059
- [1141] Bhandari L S and Thalapillil A M 2022 J. Cosmol. Astropart. Phys. JCAP03(2022)043
- [1142] Hindmarsh M, Davis A-C and Brandenberger R H 1994 Phys. Rev. D 49 1944
- [1143] Vilenkin A 1981 Phys. Rev. D 23 852
- [1144] Vilenkin A 1981 Phys. Rev. D 24 2082
- [1145] Vilenkin A 1982 Nucl. Phys. B 196 240
- [1146] Vilenkin A 1985 Phys. Rep. 121 263
- [1147] Goldhaber A S 1989 Phys. Rev. Lett. 63 2158
- [1148] Rhie S H and Bennett D P 1991 Phys. Rev. Lett. 67 1173
- [1149] Perivolaropoulos L 1992 Nucl. Phys. B 375 665
- [1150] Barriola M and Vilenkin A 1989 Phys. Rev. Lett. 63 341
- [1151] Dadhich N, Narayan K and Yajnik U A 1998 Pramana 50 307
- [1152] Abdujabbarov A, Atamurotov F, Kucukakca Y, Ahmedov B and Camci U 2013 Astrophys. Space Sci. 344 429
- [1153] Letelier P S 1979 Phys. Rev. D 20 1294
- [1154] Richarte M G and Simeone C 2008 Int. J. Mod. Phys. D 17 1179
- [1155] Bahamonde S and Jamil M 2015 Eur. Phys. J. C 75 508
- [1156] Morais Graça J P, Salako G I and Bezerra V B 2017 Int. J. Mod. Phys. D 26 1750113
- [1157] Jusufi K, Werner M C, Banerjee A and Övgün A 2017 Phys. Rev. D 95 104012
- [1158] Jusufi K 2018 Phys. Rev. D 98 044016
- [1159] Panotopoulos G and Rincón A 2018 Eur. Phys. J. C 78 851
- [1160] Li Z and Zhou T 2021 Phys. Rev. D 104 104044
- [1161] Singh D V, Ghosh S G and Maharaj S D 2020 Phys. Dark Universe 30 100730
- [1162] Güllü I and Övgün A 2022 Ann. Phys., NY 436 168721
- [1163] Mustafa G and Hussain I 2021 Eur. Phys. J. C 81 419
- [1164] He A, Tao J, Xue Y and Zhang L 2022 Chin. Phys. C 46 065102
- [1165] Khodadi M and Pourkhodabakhshi R 2021 Phys. Lett. B 823 136775
- [1166] Belhaj A and Sekhmani Y 2022 Gen. Relativ. Gravit. 54 17
- [1167] Fathi M, Olivares M and Villanueva J R 2022 Eur. Phys. J. C 82 629
- [1168] Fathi M, Olivares M and Villanueva J R 2023 Eur. Phys. J. Plus 138 7
- [1169] Ovalle J 2008 Mod. Phys. Lett. A 23 3247
- [1170] Ovalle J 2009 Int. J. Mod. Phys. D 18 837
- [1171] Ovalle J 2010 Mod. Phys. Lett. A 25 3323

- [1172] Casadio R and Ovalle J 2012 Phys. Lett. B 715 251
- [1173] Casadio R and Ovalle J 2014 Gen. Relativ. Gravit. 46 1669
- [1174] Ovalle J, Linares F, Pasqua A and Sotomayor A 2013 Class. Quantum Grav. 30 175019
- [1175] Casadio R, Ovalle J and da Rocha R a 2014 Class. Quantum Grav. 31 045016
- [1176] Ovalle J and Linares F 2013 Phys. Rev. D 88 104026
- [1177] Ovalle J, Gergely L A and Casadio R 2015 Class. Quantum Grav. 32 045015
- [1178] Casadio R, Ovalle J and da Rocha R 2015 Europhys. Lett. 110 40003
- [1179] Casadio R, Ovalle J and da Rocha R a 2015 Class. Quantum Grav. 32 215020
- [1180] Cavalcanti R T, da Silva A G and da Rocha R 2016 Class. Quantum Grav. 33 215007
- [1181] da Rocha R 2017 Phys. Rev. D 95 124017
- [1182] da Rocha R a 2017 Eur. Phys. J. C 77 355
- [1183] Ovalle J 2017 Phys. Rev. D 95 104019
- [1184] Ovalle J, Casadio R, da Rocha R and Sotomayor A 2018 Eur. Phys. J. C 78 122
- [1185] Fernandes-Silva A and da Rocha R 2018 Eur. Phys. J. C 78 271
- [1186] Gabbanelli L, Rincón A and Rubio C 2018 Eur. Phys. J. C 78 370
- [1187] Fernandes-Silva A, Ferreira-Martins A J and Da Rocha R 2018 Eur. Phys. J. C 78 631
- [1188] Contreras E and Bargueño P 2018 Eur. Phys. J. C 78 558
- [1189] Contreras E 2018 Eur. Phys. J. C 78 678
- [1190] Contreras E and Bargueño P 2018 Eur. Phys. J. C 78 985
- [1191] Ovalle J, Casadio R, da Rocha R, Sotomayor A and Stuchlik Z 2018 Europhys. Lett. 124 20004
- [1192] Ovalle J 2019 Phys. Lett. B 788 213
- [1193] Contreras E 2019 Class. Quantum Grav. 36 095004
- [1194] Contreras E and Bargueño P 2019 Class. Quantum Grav. 36 215009
- [1195] Gabbanelli L, Ovalle J, Sotomayor A, Stuchlik Z and Casadio R 2019 Eur. Phys. J. C 79 486
- [1196] Ovalle J, Posada C and Stuchlík Z 2019 Class. Quantum Grav. 36 205010
- [1197] Hensh S and Stuchlík Z 2019 Eur. Phys. J. C 79 834
- [1198] Linares Cedeño F X and Contreras E 2020 Phys. Dark Universe 28 100543
- [1199] León P and Sotomayor A 2019 Fortschr. Phys. 67 1900077
- [1200] Torres-Sánchez V A and Contreras E 2019 Eur. Phys. J. C 79 829
- [1201] Rincón A, Gabbanelli L, Contreras E and Tello-Ortiz F 2019 Eur. Phys. J. C 79 873
- [1202] Casadio R, Contreras E, Ovalle J, Sotomayor A and Stuchlick Z 2019 Eur. Phys. J. C 79 826
- [1203] Singh K N, Maurya S K, Jasim M K and Rahaman F 2019 Eur. Phys. J. C 79 851
- [1204] Maurya S K 2019 Eur. Phys. J. C 79 958
- [1205] Abellán G, Torres-Sánchez V A, Fuenmayor E and Contreras E 2020 Eur. Phys. J. C 80 177
- [1206] Contreras E, Tello-Ortiz F and Maurya S K 2020 Class. Quantum Grav. 37 155002
- [1207] Arias C, Tello-Ortiz F and Contreras E 2020 Eur. Phys. J. C 80 463
- [1208] Tello-Ortiz F 2020 Eur. Phys. J. C 80 413
- [1209] Maurya S K 2020 Eur. Phys. J. C 80 429
- [1210] Rincón A, Contreras E, Tello-Ortiz F, Bargueño P and Abellán G 2020 Eur. Phys. J. C 80 490
- [1211] Maurya S K, Singh K N and Dayanandan B 2020 Eur. Phys. J. C 80 448
- [1212] Meert P and da Rocha R 2022 Eur. Phys. J. C 82 175
- [1213] Maurya S K, Mishra B, Ray S and Nag R 2022 Chin. Phys. C 46 105105
- [1214] Cavalcanti R T, de Paiva R C and da Rocha R 2022 Eur. Phys. J. Plus 137 1185
- [1215] Yang Y, Liu D, Övgün A, Long Z-W and Xu Z 2022 arXiv:2203.11551 [gr-qc]
- [1216] Cavalcanti R T, Alves K d S and Hoff da Silva J M 2022 Universe 8 363
- [1217] Maurya S K, Newton Singh K, Lohakare S V and Mishra B 2022 Fortschr. Phys. 70 2200061
- [1218] Mahapatra S and Banerjee I 2023 Phys. Dark Universe **39** 101172
- [1219] Zhang C-M, Zhang M and Zou D-C 2023 Chin. Phys. C 47 015106
- [1220] Ovalle J, Casadio R, Contreras E and Sotomayor A 2021 Phys. Dark Universe 31 100744
- [1221] Stelle K S 1977 Phys. Rev. D 16 953
- [1222] Stelle K S 1978 Gen. Relativ. Gravit. 9 353
- [1223] Salvio A 2018 Front. Phys. 6 77
- [1224] Visser M 2009 Phys. Rev. D 80 025011
- [1225] Maggiore M 1993 Phys. Lett. B 304 65
- [1226] Maggiore M 1994 Phys. Rev. D 49 5182
- [1227] Scardigli F 1999 Phys. Lett. B 452 39
- [1228] Adler R J and Santiago D I 1999 Mod. Phys. Lett. A 14 1371
- [1229] Bruneton J-P and Larena J 2017 Gen. Relativ. Gravit. 49 56

- [1230] Veneziano G 1986 Europhys. Lett. 2 199
- [1231] Gross D J and Mende P F 1988 Nucl. Phys. B 303 407
- [1232] Amati D, Ciafaloni M and Veneziano G 1989 Phys. Lett. B 216 41
- [1233] Konishi K, Paffuti G and Provero P 1990 Phys. Lett. B 234 276
- [1234] Witten E 1996 Phys. Today 49 24
- [1235] Yoneya T 1989 Mod. Phys. Lett. A 4 1587
- [1236] Maggiore M 1993 Phys. Lett. B 319 83
- [1237] Ashtekar A, Fairhurst S and Willis J L 2003 Class. Quantum Grav. 20 1031
- [1238] Hossain G M, Husain V and Seahra S S 2010 Class. Quantum Grav. 27 165013
- [1239] Isi M, Mureika J and Nicolini P 2013 J. High Energy Phys. JHEP11(2013)139
- [1240] Bosso P and Luciano G G 2021 *Eur. Phys. J.* C **81** 982
- [1241] Luciano G G 2021 Eur. Phys. J. C 81 1086
- [1242] Jizba P, Lambiase G, Luciano G G and Petruzziello L 2022 Phys. Rev. D 105 L121501
- [1243] Bosso P, Fridman M and Luciano G G 2022 Front. Astron. Space Sci. 9 932276
- [1244] Bosso P, Petruzziello L, Wagner F and Illuminati F 2022 arXiv:2207.10418 [quant-ph]
- [1245] Snyder H S 1947 *Phys. Rev.* **71** 38
- [1246] Yang C N 1947 Phys. Rev. 72 874
- [1247] Mead C A 1964 *Phys. Rev.* **135** B849
- [1248] Hossenfelder S 2013 Living Rev. Relativ. 16 2
- [1249] Tawfik A N and Diab A M 2014 Int. J. Mod. Phys. D 23 1430025
- [1250] Bambi C and Urban F R 2008 Class. Quantum Grav. 25 095006
- [1251] Kempf A, Mangano G and Mann R B 1995 Phys. Rev. D 52 1108
- [1252] Ali A F, Das S and Vagenas E C 2009 Phys. Lett. B 678 497
- [1253] Das S, Vagenas E C and Ali A F 2010 Phys. Lett. B 690 407 Das S, Vagenas E C and Ali A F 2010 Phys. Lett. B 692 342 (erratum)
- [1254] Frassino A M and Panella O 2012 Phys. Rev. D 85 045030
- [1255] Nozari K and Etemadi A 2012 Phys. Rev. D 85 104029
- [1256] Jusufi K, Channuie P and Jamil M 2020 Eur. Phys. J. C 80 127
- [1257] Amati D, Ciafaloni M and Veneziano G 1987 Phys. Lett. B 197 81
- [1258] Capozziello S, Lambiase G and Scarpetta G 2000 Int. J. Theor. Phys. 39 15
- [1259] Scardigli F, Lambiase G and Vagenas E 2017 Phys. Lett. B 767 242
- [1260] Carr B J, Mureika J and Nicolini P 2015 J. High Energy Phys. JHEP07(2015)052
- [1261] Vagenas E C, Alsaleh S M and Farag A 2017 Europhys. Lett. 120 40001
- [1262] Carr B, Mentzer H, Mureika J and Nicolini P 2020 Eur. Phys. J. C 80 1166
- [1263] Feng Z-W, Yang S-Z, Li H-L and Zu X-T 2017 Phys. Lett. B 768 81
- [1264] Jusufi K, Azreg-Aïnou M, Jamil M and Zhu T 2022 Int. J. Geom. Methods Mod. Phys. 19 2250068
- [1265] Tamburini F, Feleppa F and Thidé B 2022 Phys. Lett. B 826 136894
- [1266] Anacleto M A, Campos J A V, Brito F A and Passos E 2021 Ann. Phys., NY 434 168662
- [1267] Scardigli F and Casadio R 2015 Eur. Phys. J. C 75 425
- [1268] Ökcü O and Aydiner E 2021 Nucl. Phys. B 964 115324
- [1269] Mathur S D 2002 Int. J. Mod. Phys. D 11 1537
- [1270] Mathur S D 2005 Fortschr. Phys. 53 793
- [1271] Giddings S B 2017 Nat. Astron. 1 0067
- [1272] Dvali G and Gomez C 2013 Fortschr. Phys. 61 742
- [1273] Dvali G and Gomez C 2012 Phys. Lett. B 716 240
- [1274] Dvali G and Gomez C 2013 Phys. Lett. B 719 419
- [1275] Casadio R, Lenzi M and Micu O 2018 Phys. Rev. D 98 104016
- [1276] Buoninfante L, Luciano G G and Petruzziello L 2019 Eur. Phys. J. C 79 663
- [1277] Zhu T, Ren J-R and Li M-F 2009 Phys. Lett. B 674 204
- [1278] Moradpour H, Corda C, Ziaie A H and Ghaffari S 2019 Europhys. Lett. 127 60006
- [1279] Moradpour H, Ziaie A H, Ghaffari S and Feleppa F 2019 Mon. Not. R. Astron. Soc. 488 L69
- [1280] Aghababaei S, Moradpour H and Vagenas E C 2021 Eur. Phys. J. Plus 136 997
- [1281] Mureika J R 2019 Phys. Lett. B 789 88
- [1282] Kumaran Y and Övgün A 2020 Chin. Phys. C 44 025101
- [1283] Pantig R C, Yu P K, Rodulfo E T and Övgün A 2022 Ann. Phys., NY 436 168722
- [1284] Pedram P 2012 Phys. Lett. B 714 317
- [1285] Pedram P 2012 Phys. Lett. B 718 638

- [1286] Doplicher S, Fredenhagen K and Roberts J E 1995 Commun. Math. Phys. 172 187
- [1287] Connes A 1994 Noncommutative Geometry (Cambridge, MA: Academic)
- [1288] Witten E 1996 Nucl. Phys. B 460 335
- [1289] Weyl H 1927 Z. Phys. 46 1
- [1290] Wigner E P 1932 Phys. Rev. 40 749
- [1291] Moyal J E 1949 Proc. Camb. Phil. Soc. 45 99
- [1292] Alvarez-Gaume L and Wadia S R 2001 Phys. Lett. B 501 319
- [1293] Alvarez-Gaume L and Barbon J L F 2001 Int. J. Mod. Phys. A 16 1123
- [1294] Smailagic A and Spallucci E 2003 J. Phys. A: Math. Gen. 36 L467
- [1295] Smailagic A and Spallucci E 2003 J. Phys. A: Math. Gen. 36 L517
- [1296] Gruppuso A 2005 J. Phys. A: Math. Gen. 38 2039
- [1297] Bimonte G, Lizzi F and Sparano G 1994 Phys. Lett. B 341 139
- [1298] Lizzi F, Mangano G, Miele G and Sparano G 1996 Int. J. Mod. Phys. A 11 2907
- [1299] Bimonte G, Ercolessi E, Landi G, Lizzi F, Sparano G and Teotonio-Sobrinho P 1996 J. Geom. Phys. 20 329
- [1300] Balachandran A P, Bimonte G, Ercolessi E, Landi G, Lizzi F, Sparano G and Teotonio-Sobrinho P 1996 J. Geom. Phys. 18 163
- [1301] Connes A 1996 Commun. Math. Phys. 182 155
- [1302] Lizzi F, Mangano G, Miele G and Sparano G 1996 Mod. Phys. Lett. A 11 2561
- [1303] Balachandran A P, Bimonte G, Landi G, Lizzi F and Teotonio-Sobrinho P 1998 J. Geom. Phys. 24 353
- [1304] Lizzi F, Mangano G, Miele G and Sparano G 1997 Phys. Rev. D 55 6357
- [1305] Figueroa H, Gracia-Bondia J M, Lizzi F and Varilly J C 1998 J. Geom. Phys. 26 329
- [1306] Lizzi F, Mangano G, Miele G and Sparano G 1998 Mod. Phys. Lett. A 13 231
- [1307] Landi G, Lizzi F and Szabo R J 1999 Commun. Math. Phys. 206 603
- [1308] Lizzi F 1999 Int. J. Mod. Phys. A 14 4501
- [1309] Hashimoto A and Itzhaki N 1999 Phys. Lett. B 465 142
- [1310] Maldacena J M and Russo J G 1999 J. High Energy Phys. JHEP09(1999)025
- [1311] Matusis A, Susskind L and Toumbas N 2000 J. High Energy Phys. JHEP12(2000)002
- [1312] Seiberg N, Susskind L and Toumbas N 2000 J. High Energy Phys. JHEP06(2000)044
- [1313] Jurco B, Schraml S, Schupp P and Wess J 2000 Eur. Phys. J. C 17 521
- [1314] Lizzi F, Mangano G and Miele G 2001 Mod. Phys. Lett. A 16 1
- [1315] Lizzi F, Zampini A and Szabo R J 2001 J. High Energy Phys. JHEP08(2001)032
- [1316] Calmet X, Jurco B, Schupp P, Wess J and Wohlgenannt M 2002 Eur. Phys. J. C 23 363
- [1317] Lizzi F, Mangano G, Miele G and Peloso M 2002 J. High Energy Phys. JHEP06(2002)049
- [1318] Chaichian M, Sheikh-Jabbari M M and Tureanu A 2004 Eur. Phys. J. C 36 251
- [1319] Schupp P, Trampetic J, Wess J and Raffelt G 2004 Eur. Phys. J. C 36 405
- [1320] Tsujikawa S, Maartens R and Brandenberger R 2003 Phys. Lett. B 574 141
- [1321] Lizzi F, Stern A and Vitale P 2005 J. Math. Phys. 46 103516
- [1322] Lizzi F 2007 Int. J. Mod. Phys. A 22 1317
- [1323] Galluccio S, Lizzi F and Vitale P 2008 Phys. Rev. D 78 085007
- [1324] Galluccio S, Lizzi F and Vitale P 2009 J. High Energy Phys. JHEP09(2009)054
- [1325] Rinaldi M 2011 Class. Quantum Grav. 28 105022
- [1326] Rinaldi M 2010 Mod. Phys. Lett. A 25 2805
- [1327] Machado U D and Opher R 2012 Class. Quantum Grav. 29 065003
- [1328] Perrier H, Durrer R and Rinaldi M 2013 J. High Energy Phys. JHEP01(2013)067
- [1329] Machado U D and Opher R 2013 Phys. Rev. D 87 123517
- [1330] Li N and Zhang X 2013 Phys. Rev. D 88 023508
- [1331] Calcagni G, Kuroyanagi S, Ohashi J and Tsujikawa S 2014 J. Cosmol. Astropart. Phys. JCAP03(2014)052
- [1332] Shiraishi M, Mota D F, Ricciardone A and Arroja F 2014 J. Cosmol. Astropart. Phys. JCAP07(2014)047
- [1333] Rahaman F, Karmakar S, Karar I and Ray S 2015 Phys. Lett. B 746 73
- [1334] Joby P K, Chingangbam P and Das S 2015 Phys. Rev. D 91 083503
- [1335] D'Andrea F, Kurkov M A and Lizzi F 2016 Phys. Rev. D 94 025030
- [1336] Kobakhidze A, Lagger C and Manning A 2016 Phys. Rev. D 94 064033
- [1337] Araujo T, Bakhmatov I, Colgáin E O, Sakamoto J-I, Sheikh-Jabbari M M and Yoshida K 2018 J. Phys. A: Math. Theor. 51 235401

- [1338] Lizzi F and Pinzul A 2017 Phys. Rev. D 96 126013
- [1339] Kurkov M A and Lizzi F 2018 Phys. Rev. D 97 085024
- [1340] Bouhmadi-López M, Brahma S, Chen C-Y, Chen P and Yeom D-H 2020 Phys. Dark Universe 30 100701
- [1341] Devastato A, Kurkov M and Lizzi F 2019 Int. J. Mod. Phys. A 34 1930010
- [1342] Lizzi F, Manfredonia M and Mercati F 2020 Nucl. Phys. B 958 115117
- [1343] Lizzi F and Vitale P 2021 *Phys. Lett.* B **818** 136372
- [1344] Cuzinatto R R, de Montigny M and Pompeia P J 2022 Class. Quantum Grav. 39 075006
- [1345] Cuzinatto R R, de Montigny M and Pompeia P J 2022 Class. Quantum Grav. 39 075007
- [1346] Guendelman E and Singleton D 2023 Symmetry 15 126
- [1347] Nicolini P, Smailagic A and Spallucci E 2006 Phys. Lett. B 632 547
- [1348] Banerjee R, Gangopadhyay S and Modak S K 2010 Phys. Lett. B 686 181
- [1349] Övgün A and Jusufi K 2016 Eur. Phys. J. Plus 131 177
- [1350] Nunes R C, Moradpour H, Barboza E M, Abreu E M C and Neto J A 2017 Int. J. Geom. Methods Mod. Phys. 15 1850004
- [1351] Gangopadhyay S, Paik B and Mandal R 2018 Int. J. Mod. Phys. A 33 1850084
- [1352] Haldar S, Corda C and Chakraborty S 2018 Adv. High Energy Phys. 2018 9851598
- [1353] Kumar R and Ghosh S G 2017 Eur. Phys. J. C 77 577
- [1354] Ghosh S G 2018 Class. Quantum Grav. 35 085008
- [1355] Övgün A, Sakallı I, Saavedra J and Leiva C 2020 Mod. Phys. Lett. A 35 2050163
- [1356] Panotopoulos G and Rincón A 2020 Eur. Phys. J. Plus 135 33
- [1357] Ghosh S G and Maharaj S D 2021 Phys. Dark Universe 31 100793
- [1358] Maceda M, Macias A and Martinez-Carbajal D 2021 Int. J. Mod. Phys. A 36 2150191
- [1359] Zeng X-X, Li G-P and He K-J 2022 Nucl. Phys. B 974 115639
- [1360] Jha S K and Rahaman A 2021 arXiv:2111.02817 [gr-qc]
- [1361] Graça J P M, Capossoli E F and Boschi-Filho H 2021 Europhys. Lett. 135 41002
- [1362] Rayimbaev J, Bokhari A H and Ahmedov B 2022 Class. Quantum Grav. 39 075021
- [1363] Wei S-W, Cheng P, Zhong Y and Zhou X-N 2015 J. Cosmol. Astropart. Phys. JCAP08(2015)004
- [1364] Nozari K and Mehdipour S H 2008 Class. Quantum Grav. 25 175015
- [1365] Anacleto M A, Brito F A, Campos J A V and Passos E 2020 Phys. Lett. B 803 135334
- [1366] Campos J A V, Anacleto M A, Brito F A and Passos E 2022 Sci. Rep. 12 8516
- [1367] Lambiase G, Sakellariadou M and Stabile A 2013 J. Cosmol. Astropart. Phys. JCAP12(2013)020
- [1368] Hawking S W 1971 Phys. Rev. Lett. 26 1344
- [1369] Bekenstein J D 1973 Phys. Rev. D 7 2333
- [1370] Bardeen J M, Carter B and Hawking S W 1973 Commun. Math. Phys. 31 161
- [1371] Wald R M 2001 Living Rev. Relativ. 46
- [1372] Bekenstein J D 1972 Lett. Nuovo Cimento 4 737
- [1373] Tsallis C 1988 J. Stat. Phys. 52 479
- [1374] Tsallis C and Cirto L J L 2013 Eur. Phys. J. C 73 2487
- [1375] Rényi A 1959 Acta Math. Acad. Sci. Hung. 10 193
- [1376] Kaniadakis G 2005 Phys. Rev. E 72 036108
- [1377] Majhi A 2017 Phys. Lett. B 775 32
- [1378] Kaul R K and Majumdar P 2000 Phys. Rev. Lett. 84 5255
- [1379] Banerjee S, Gupta R K, Mandal I and Sen A 2011 J. High Energy Phys. JHEP11(2011)143
- [1380] Nojiri S, Odintsov S D and Saridakis E N 2019 Eur. Phys. J. C 79 242
- [1381] Nojiri S, Odintsov S D and Faraoni V 2022 Phys. Rev. D 105 044042
- [1382] Nojiri S, Odintsov S D and Paul T 2022 Phys. Lett. B 831 137189
- [1383] Nojiri S, Odintsov S D and Faraoni V 2021 Phys. Rev. D 104 084030
- [1384] Nojiri S, Odintsov S D and Faraoni V 2022 Int. J. Geom. Methods Mod. Phys. 19 2250210
- [1385] Di Gennaro S, Xu H and Ong Y C 2022 Eur. Phys. J. C 82 1066
- [1386] Çimdiker I, Dabrowski M P and Gohar H 2022 arXiv:2208.04473 [gr-qc]
- [1387] Abreu E M C and Neto J A 2020 Phys. Lett. B 810 135805
- [1388] Barrow J D 2020 Phys. Lett. B 808 135643
- [1389] Saridakis E N 2020 Phys. Rev. D 102 123525
- [1390] Saridakis E N 2020 J. Cosmol. Astropart. Phys. JCAP07(2020)031
- [1391] Mamon A A, Paliathanasis A and Saha S 2021 Eur. Phys. J. Plus 136 134
- [1392] Srivastava S and Sharma U K 2021 Int. J. Geom. Methods Mod. Phys. 18 2150014
- [1393] Sharma U K, Varshney G and Dubey V C 2021 Int. J. Mod. Phys. D 30 2150021

- [1394] Pradhan A, Dixit A and Bhardwaj V K 2021 Int. J. Mod. Phys. A 36 2150030
- [1395] Sheykhi A 2021 Phys. Rev. D 103 123503
- [1396] Bhardwaj V K, Dixit A and Pradhan A 2021 New Astron. 88 101623
- [1397] Dixit A, Bharadwaj V K and Pradhan A 2022 Chin. J. Phys. 77 646
- [1398] Chakraborty G, Chattopadhyay S, Güdekli E and Radinschi I 2021 Symmetry 13 562
- [1399] Adhikary P, Das S, Basilakos S and Saridakis E N 2021 Phys. Rev. D 104 123519
- [1400] Asghari M and Sheykhi A 2022 Eur. Phys. J. C 82 388
- [1401] Nojiri S, Odintsov S D and Paul T 2022 Phys. Lett. B 825 136844
- [1402] Maity S and Rudra P 2022 J. Hologr. Appl. Phys. 2 1
- [1403] Paul B C, Roy B C and Saha A 2022 Eur. Phys. J. C 82 76
- [1404] Sharma U K and Kumar M 2022 Int. J. Geom. Methods Mod. Phys. 19 2250066
- [1405] Remya A, Ram Babu Kanaujia P and Sharma U K 2022 Int. J. Geom. Methods Mod. Phys. 19 2250082
- [1406] Srivastava M, Kumar M and Srivastava S 2022 Grav. Cosmol. 28 70
- [1407] Luciano G G and Saridakis E N 2022 Eur. Phys. J. C 82 558
- [1408] Oliveros A, Sabogal M A and Acero M A 2022 Eur. Phys. J. Plus 137 783
- [1409] Sheykhi A and Farsi B 2022 Eur. Phys. J. C 82 1111
- [1410] Di Gennaro S and Ong Y C 2022 Universe 8 541
- [1411] Luciano G G and Liu Y 2022 arXiv:2205.13458 [hep-th]
- [1412] Luciano G G 2022 *Phys. Rev.* D **106** 083530
- [1413] Luciano G G and Giné J 2022 arXiv:2210.09755 [gr-qc]
- [1414] Luciano G G 2023 arXiv:2301.12509 [gr-qc]
- [1415] Luciano G G 2023 arXiv:2301.12488 [gr-qc]
- [1416] Anagnostopoulos F K, Basilakos S and Saridakis E N 2020 Eur. Phys. J. C 80 826
- [1417] Leon G, Magaña J, Hernández-Almada A, García-Aspeitia M A, Verdugo T and Motta V 2021 J. Cosmol. Astropart. Phys. JCAP12(2021)032
- [1418] Barrow J D, Basilakos S and Saridakis E N 2021 Phys. Lett. B 815 136134
- [1419] Knox L and Millea M 2020 Phys. Rev. D 101 043533
- [1420] Di Valentino E, Mena O, Pan S, Visinelli L, Yang W, Melchiorri A, Mota D F, Riess A G and Silk J 2021 Class. Quantum Grav. 38 153001
- [1421] Perivolaropoulos L and Skara F 2022 New Astron. Rev. 95 101659
- [1422] Schöneberg N, Franco Abellán G, Pérez Sánchez A, Witte S J, Poulin V and Lesgourgues J 2022 Phys. Rep. 984 1
- [1423] Abdalla E et al 2022 J. High Energy Astrophys. 34 49
- [1424] Vagnozzi S, Dhawan S, Gerbino M, Freese K, Goobar A and Mena O 2018 Phys. Rev. D 98 083501
- [1425] Yang W, Pan S, Di Valentino E, Nunes R C, Vagnozzi S and Mota D F 2018 J. Cosmol. Astropart. Phys. JCAP09(2018)019
- [1426] El-Zant A, El Hanafy W and Elgammal S 2019 Astrophys. J. 871 210
- [1427] Li X and Shafieloo A 2019 Astrophys. J. Lett. 883 L3
- [1428] Vagnozzi S 2020 Phys. Rev. D 102 023518
- [1429] Visinelli L, Vagnozzi S and Danielsson U 2019 Symmetry 11 1035
- [1430] Di Valentino E, Melchiorri A, Mena O and Vagnozzi S 2020 Phys. Rev. D 101 063502
- [1431] Zumalacarregui M 2020 Phys. Rev. D 102 023523
- [1432] Alestas G, Kazantzidis L and Perivolaropoulos L 2020 Phys. Rev. D 101 123516
- [1433] Di Valentino E, Mukherjee A and Sen A A 2021 Entropy 23 404
- [1434] Yang W, Pan S, Di Valentino E, Mena O and Melchiorri A 2021 J. Cosmol. Astropart. Phys. JCAP10(2021)008
- [1435] Kumar S 2021 Phys. Dark Universe 33 100862
- [1436] Bag S, Sahni V, Shafieloo A and Shtanov Y 2021 Astrophys. J. 923 212
- [1437] Roy N, Goswami S and Das S 2022 Phys. Dark Universe 36 101037
- [1438] Sharma R K, Pandey K L and Das S 2022 Astrophys. J. 934 113
- [1439] Horndeski G W 1976 J. Math. Phys. 17 1980
- [1440] Barrow J D, Thorsrud M and Yamamoto K 2013 J. High Energy Phys. JHEP02(2013)146
- [1441] Beltran Jimenez J, Durrer R, Heisenberg L and Thorsrud M 2013 J. Cosmol. Astropart. Phys. JCAP10(2013)064
- [1442] Heisenberg L 2014 J. Cosmol. Astropart. Phys. JCAP05(2014)015
- [1443] Davydov E and Gal'tsov D 2016 Phys. Lett. B 753 622

- [1444] De Felice A, Heisenberg L, Kase R, Tsujikawa S, Zhang Y-L and Zhao G-B 2016 Phys. Rev. D 93 104016
- [1445] De Felice A, Heisenberg L, Kase R, Mukohyama S, Tsujikawa S and Zhang Y-L 2016 Phys. Rev. D 94 044024
- [1446] Heisenberg L, Kase R and Tsujikawa S 2016 Phys. Lett. B 760 617
- [1447] Beltran Jimenez J, Heisenberg L, Kase R, Namba R and Tsujikawa S 2017 Phys. Rev. D 95 063533
- [1448] Heisenberg L, Kase R, Minamitsuji M and Tsujikawa S 2017 J. Cosmol. Astropart. Phys. JCAP08(2017)024
- [1449] Nakamura S, Kase R and Tsujikawa S 2017 Phys. Rev. D 96 084005
- [1450] Davydov E 2017 Int. J. Mod. Phys. D 27 1850038
- [1451] Brihaye Y and Verbin Y 2020 Phys. Rev. D 102 124021
- [1452] Gallego Cadavid A, Rodriguez Y and Gómez L G 2020 Phys. Rev. D 102 104066
- [1453] Verbin Y 2022 Phys. Rev. D 106 024057
- [1454] Brihaye Y and Verbin Y 2021 Phys. Rev. D 104 024047
- [1455] Gurses M, Heydarzade Y and Senturk C 2021 Eur. Phys. J. C 81 1147
- [1456] Garnica J C, Gomez L G, Navarro A A and Rodriguez Y 2022 Ann. Phys., Lpz. 534 2100453
- [1457] Brihaye Y, Hartmann B, Kleihaus B and Kunz J 2022 Phys. Rev. D 105 044050
- [1458] Gallego Cadavid A, Nieto C M and Rodriguez Y 2022 Phys. Rev. D 105 124060
- [1459] Gallego Cadavid A, Nieto C M and Rodriguez Y 2022 Phys. Rev. D 105 104051
- [1460] Allahyari A, Gorji M A and Mukohyama S 2020 J. Cosmol. Astropart. Phys. JCAP05(2020)013 Allahyari A, Gorji M A and Mukohyama S 2021 J. Cosmol. Astropart. Phys. JCAP05(2021)E02 (erratum)
- [1461] Bluhm R and Kostelecky V A 2005 Phys. Rev. D 71 065008
- [1462] Bluhm R, Gagne N L, Potting R and Vrublevskis A 2008 Phys. Rev. D 77 125007
- Bluhm R, Gagne N L, Potting R and Vrublevskis A 2009 *Phys. Rev.* D **79** 029902 (erratum) [1463] Bertolami O and Paramos J 2005 *Phys. Rev.* D **72** 044001
- [1464] Seifert M D 2010 Phys. Rev. D 81 065010
- [1465] Maluf R V, Almeida C A S, Casana R and Ferreira M M Jr 2014 Phys. Rev. D 90 025007
- [1466] Santos A F, Jesus W D R, Nascimento J R and Petrov A Y 2015 Mod. Phys. Lett. A 30 1550011
- [1467] Hernaski C 2014 Phys. Rev. D 90 124036
- [1468] Capelo D and Páramos J 2015 Phys. Rev. D 91 104007
- [1469] Maluf R V, Silva J E G and Almeida C A S 2015 Phys. Lett. B 749 304
- [1470] Gomes D A, Maluf R V and Almeida C A S 2020 Ann. Phys., NY 418 168198
- [1471] Oliveira R, Dantas D M, Santos V and Almeida C A S 2019 Class. Quantum Grav. 36 105013
- [1472] Assunção J F, Mariz T, Nascimento J R and Petrov A Y 2019 Phys. Rev. D 100 085009
- [1473] Kanzi S and Sakallı I 2019 Nucl. Phys. B 946 114703
- [1474] Li Z and Övgün A 2020 *Phys. Rev.* D **101** 024040
- [1475] Maluf R V and Neves J C S 2021 *Phys. Rev.* D 103 044002
- [1476] Jha S K and Rahaman A 2021 Eur. Phys. J. C 81 345
- [1477] Kumar Jha S, Barman H and Rahaman A 2021 J. Cosmol. Astropart. Phys. JCAP04(2021)036
- [1478] Ding C, Chen X and Fu X 2022 Nucl. Phys. B 975 115688
- [1479] Khodadi M 2021 Phys. Rev. D 103 064051
- [1480] Jha S K, Aziz S and Rahaman A 2022 Eur. Phys. J. C 82 106
- [1481] Maluf R V and Neves J C S 2021 J. Cosmol. Astropart. Phys. JCAP10(2021)038
- [1482] Jiang R, Lin R-H and Zhai X-H 2021 Phys. Rev. D 104 124004
- [1483] Wang Z, Chen S and Jing J 2022 Eur. Phys. J. C 82 528
- [1484] Khodadi M 2022 Phys. Rev. D 105 023025
- [1485] Gogoi D J and Goswami U D 2022 J. Cosmol. Astropart. Phys. JCAP06(2022)029
- [1486] Gu J, Riaz S, Abdikamalov A B, Ayzenberg D and Bambi C 2022 Eur. Phys. J. C 82 708
- [1487] Uniyal A, Kanzi S and Sakallı I 2022 arXiv:2207.10122 [hep-th]
- [1488] Casana R, Cavalcante A, Poulis F P and Santos E B 2018 Phys. Rev. D 97 104001
- [1489] Ding C, Liu C, Casana R and Cavalcante A 2020 Eur. Phys. J. C 80 178
- [1490] Ding C and Chen X 2021 Chin. Phys. C 45 025106
- [1491] Maluf R V and Muniz C R 2022 Eur. Phys. J. C 82 94
- [1492] Kanzi S and Sakallı I 2022 Eur. Phys. J. C 82 93
- [1493] Kuang X-M and Övgün A 2022 Ann. Phys., NY 447 169147
- [1494] Jackiw R and Pi S Y 2003 Phys. Rev. D 68 104012

- [1495] Alexander S and Yunes N 2009 Phys. Rep. 480 1
- [1496] Achucarro A and Townsend P K 1986 Phys. Lett. B 180 89
- [1497] Alexander S H S and Gates S J Jr 2006 J. Cosmol. Astropart. Phys. JCAP06(2006)018
- [1498] Weinberg S 2008 Phys. Rev. D 77 123541
- [1499] Taveras V and Yunes N 2008 Phys. Rev. D 78 064070
- [1500] Calcagni G and Mercuri S 2009 Phys. Rev. D 79 084004
- [1501] Bartolo N and Orlando G 2017 J. Cosmol. Astropart. Phys. JCAP07(2017)034
- [1502] Visinelli L and Vagnozzi S 2019 Phys. Rev. D 99 063517
- [1503] Bartolo N, Orlando G and Shiraishi M 2019 J. Cosmol. Astropart. Phys. JCAP01(2019)050
- [1504] Nojiri S, Odintsov S D, Oikonomou V K and Popov A A 2019 Phys. Rev. D 100 084009
- [1505] Nojiri S, Odintsov S D, Oikonomou V K and Popov A A 2020 Phys. Dark Universe 28 100514
- [1506] Mirzagholi L, Komatsu E, Lozanov K D and Watanabe Y 2020 J. Cosmol. Astropart. Phys. JCAP06(2020)024
- [1507] Odintsov S D and Oikonomou V K 2022 Phys. Rev. D 105 104054
- [1508] Hou J, Slepian Z and Cahn R N 2022 arXiv:2206.03625 [astro-ph.CO]
- [1509] Philcox O H E 2022 *Phys. Rev.* D **106** 063501
- [1510] Lambiase G, Mastrototaro L and Visinelli L 2023 J. Cosmol. Astropart. Phys. JCAP01(2023)011
- [1511] Grumiller D and Yunes N 2008 Phys. Rev. D 77 044015
- [1512] Yunes N and Pretorius F 2009 Phys. Rev. D 79 084043
- [1513] Ali-Haimoud Y and Chen Y 2011 Phys. Rev. D 84 124033
- [1514] Yagi K, Yunes N and Tanaka T 2012 Phys. Rev. D 86 044037
- Yagi K, Yunes N and Tanaka T 2014 Phys. Rev. D 89 049902 (erratum)
- [1515] Maselli A, Pani P, Cotesta R, Gualtieri L, Ferrari V and Stella L 2017 Astrophys. J. 843 25
  [1516] Casimir H B G 1948 Indag. Math. 10 261
- [1517] Visser M 1995 Lorentzian Wormholes: From Einstein to Hawking (College Park, MD: American Institute of Physics)
- [1518] Khabibullin A R, Khusnutdinov N R and Sushkov S V 2006 Class. Quantum Grav. 23 627
- [1519] Butcher L M 2014 Phys. Rev. D 90 024019
- [1520] Tripathy S K 2021 Phys. Dark Universe 31 100757
- [1521] Santos A C L, Muniz C R and Oliveira L T 2021 Int. J. Mod. Phys. D 30 2150032
- [1522] Garattini R 2020 Eur. Phys. J. C 80 1172
- [1523] Javed W, Hamza A and Övgün A 2020 Mod. Phys. Lett. A 35 2050322
- [1524] Garattini R 2021 Eur. Phys. J. C 81 824
- [1525] Samart D, Tangphati T and Channuie P 2022 Nucl. Phys. B 980 115848
- [1526] Sokoliuk O, Baransky A and Sahoo P K 2022 Nucl. Phys. B 980 115845
- [1527] Hassan Z, Ghosh S, Sahoo P K and Bamba K 2022 Eur. Phys. J. C 82 1116
- [1528] Garattini R 2019 Eur. Phys. J. C 79 951
- [1529] Green M B, Schwarz J H and Brink L 1982 Nucl. Phys. B 198 474
- [1530] Padmanabhan T 1997 Phys. Rev. Lett. 78 1854
- [1531] Gaete P and Nicolini P 2022 Phys. Lett. B 829 137100
- [1532] Mondal V 2022 Eur. Phys. J. C 82 358
- [1533] Nicolini P, Spallucci E and Wondrak M F 2019 Phys. Lett. B 797 134888
- [1534] Gaete P, Jusufi K and Nicolini P 2022 Phys. Lett. B 835 137546
- [1535] Will C M 2001 Living Rev. Relativ. 4 4
- [1536] Wex N and Kramer M 2020 Universe 6 156
- [1537] Yunes N and Pretorius F 2009 Phys. Rev. D 80 122003
- [1538] Buonanno A and Damour T 1999 Phys. Rev. D 59 084006
- [1539] Khan S, Husa S, Hannam M, Ohme F, Pürrer M, Jiménez Forteza X and Bohé A 2016 Phys. Rev. D 93 044007
- [1540] Khoury J and Weltman A 2004 Phys. Rev. Lett. 93 171104
- [1541] Sakstein J 2018 Int. J. Mod. Phys. D 27 1848008
- [1542] Barausse E, Cardoso V and Pani P 2014 Phys. Rev. D 89 104059
- [1543] Honardoost M, Khosravi N and Rahmanpour N 2019 arXiv:1905.12946 [gr-qc]
- [1544] ter Haar L, Bezares M, Crisostomi M, Barausse E and Palenzuela C 2021 *Phys. Rev. Lett.* 126 091102
- [1545] Bezares M, ter Haar L, Crisostomi M, Barausse E and Palenzuela C 2021 Phys. Rev. D 104 044022

- [1546] Bezares M, Aguilera-Miret R, ter Haar L, Crisostomi M, Palenzuela C and Barausse E 2022 Phys. Rev. Lett. 128 091103
- [1547] Psaltis D, Perrodin D, Dienes K R and Mocioiu I 2008 Phys. Rev. Lett. 100 091101
- [1548] Barausse E and Sotiriou T P 2008 Phys. Rev. Lett. 101 099001
- [1549] Gair J and Yunes N 2011 Phys. Rev. D 84 064016
- [1550] Abuter R et al 2022 Astron. Astrophys. 665 A75
- [1551] Lopez B et al 2022 Astron. Astrophys. 659 A192
- [1552] Blackburn L et al 2019 Bull. Am. Astron. Soc. 51 256
- [1553] Fish V L, Shea M and Akiyama K 2020 Adv. Space Res. 65 821
- [1554] Uttley P et al 2021 Exp. Astron. 51 1081
- [1555] Tamburini F, Feleppa F, Licata I and Thidé B 2021 Phys. Rev. A 104 013718
- [1556] Tamburini F, Feleppa F and Thidé B 2021 Int. J. Mod. Phys. D 30 2142017
- [1557] Davoudiasl H and Denton P B 2019 Phys. Rev. Lett. 123 021102
- [1558] Yuan G-W, Xia Z-Q, Tang C, Zhao Y, Cai Y-F, Chen Y, Shu J and Yuan Q 2021 J. Cosmol. Astropart. Phys. JCAP03(2021)018
- [1559] Chen Y, Liu Y, Lu R-S, Mizuno Y, Shu J, Xue X, Yuan Q and Zhao Y 2022 Nat. Astron. 6 592
- [1560] Himwich E, Johnson M D, Lupsasca A and Strominger A 2020 Phys. Rev. D 101 084020
- [1561] Hadar S, Johnson M D, Lupsasca A and Wong G N 2021 Phys. Rev. D 103 104038
- [1562] Chen C-Y, Chen Y-J, Ho M-Y and Tseng Y-H 2022 arXiv:2212.10028 [gr-qc]
- [1563] Zhu Q-H 2023 arXiv:2301.00913 [gr-qc]
- [1564] Wang X, Hou Y and Guo M 2023 arXiv:2301.04851 [gr-qc]
- [1565] Shahzadi M, Kološ M, Stuchlík Z and Habib Y 2022 Eur. Phys. J. C 82 407
- [1566] Beltrán Jiménez J, Heisenberg L and Koivisto T S 2019 Universe 5 173
- [1567] Nashed G G L 2002 Phys. Rev. D 66 064015
- [1568] Capozziello S, Gonzalez P A, Saridakis E N and Vasquez Y 2013 J. High Energy Phys. JHEP02(2013)039
- [1569] Nashed G G L and Saridakis E N 2019 Class. Quantum Grav. 36 135005
- [1570] Pfeifer C and Schuster S 2021 Universe 7 153
- [1571] Bahamonde S, Dialektopoulos K F, Escamilla-Rivera C, Farrugia G, Gakis V, Hendry M, Hohmann M, Levi Said J, Mifsud J and Di Valentino E 2023 *Rep. Prog. Phys.* 86 026901
- [1572] D'Ambrosio F, Fell S D B, Heisenberg L and Kuhn S 2022 *Phys. Rev.* D 105 024042
- [1573] Bahamonde S, Ducobu L and Pfeifer C 2022 J. Cosmol. Astropart. Phys. JCAP04(2022)018
- [1574] Bahamonde S, Gigante Valcarcel J, Järv L and Lember J 2022 J. Cosmol. Astropart. Phys. JCAP08(2022)082
- [1575] Calzá M and Sebastiani L 2023 Eur. Phys. J. C 83 247
- [1576] Bahamonde S, Chevrier J and Gigante Valcarcel J 2023 J. Cosmol. Astropart. Phys. JCAP02(2023)018
- [1577] Dymnikova I 1992 Gen. Relativ. Gravit. 24 235
- [1578] Anabalon A and Oliva J 2012 Phys. Rev. D 86 107501
- [1579] Nashed G G L and Capozziello S 2019 Phys. Rev. D 99 104018
- [1580] Dimakis N, Leon G and Paliathanasis A 2021 Phys. Rev. D 103 044001
- [1581] Islam S U, Ghosh S G and Maharaj S D 2022 arXiv:2203.00957 [gr-qc]
- [1582] Chen C-Y 2022 Phys. Rev. D 106 044009
- [1583] Jusufi K, Capozziello S, Bahamonde S and Jamil M 2022 Eur. Phys. J. C 82 1018
- [1584] Wang D 2022 arXiv:2205.08026 [gr-qc]
- [1585] Uniyal A, Pantig R C and Övgün A 2023 Phys. Dark Universe 40 101178
- [1586] Carballo-Rubio R, Di Filippo F, Liberati S and Visser M 2022 J. Cosmol. Astropart. Phys. JCAP08(2022)055
- [1587] Pantig R C and Övgün A 2023 Ann. Phys., NY 448 169197
- [1588] Ghosh S G and Afrin M 2023 Astrophys. J. 944 174
- [1589] Kuang X-M, Tang Z-Y, Wang B and Wang A 2022 Phys. Rev. D 106 064012
- [1590] Wang D 2022 arXiv:2206.07577 [gr-qc]
- [1591] Khodadi M and Lambiase G 2022 Phys. Rev. D 106 104050
- [1592] Banerjee I, Sau S and SenGupta S 2022 J. Cosmol. Astropart. Phys. JCAP09(2022)066
- [1593] Kumar Walia R, Ghosh S G and Maharaj S D 2022 Astrophys. J. 939 77
- [1594] Kumar Walia R 2023 J. Cosmol. Astropart. Phys. JCAP03(2023)029
- [1595] Banerjee I, Sau S and SenGupta S 2022 arXiv:2207.06034 [gr-qc]
- [1596] Mustafa G, Atamurotov F, Hussain I, Shaymatov S and Övgün A 2022 Chin. Phys. C 46 125107

- [1597] Banerjee I, Chakraborty S and SenGupta S 2022 Phys. Rev. D 106 084051
- [1598] Carballo-Rubio R, Cardoso V and Younsi Z 2022 Phys. Rev. D 106 084038
- [1599] Shaikh R 2022 arXiv:2208.01995 [gr-qc]
- [1600] Pantig R C, Övgün A and Demir D 2023 Eur. Phys. J. C 83 250
- [1601] Saha A K, Parashari P, Maity T N, Dubey A, Bouri S and Laha R 2022 arXiv:2208.03530 [astro-ph.HE]
- [1602] Pantig R C, Mastrototaro L, Lambiase G and Övgün A 2022 Eur. Phys. J. C 82 1155
- [1603] Zakharov A F 2022 arXiv:2208.06805 [astro-ph.GA]
- [1604] Atamurotov F, Hussain I, Mustafa G and Jusufi K 2022 Eur. Phys. J. C 82 831
- [1605] Oikonomou V K, Tsyba P and Razina O 2022 Universe 8 484
- [1606] Kumar J, Islam S U and Ghosh S G 2022 Astrophys. J. 938 104
- [1607] Sengo I, Cunha P V P, Herdeiro C A R and Radu E 2023 J. Cosmol. Astropart. Phys. JCAP01(2023)047
- [1608] Ghosh R, Rahman M and Mishra A K 2023 Eur. Phys. J. C 83 91
- [1609] Afrin M, Vagnozzi S and Ghosh S G 2023 Astrophys. J. 944 149
- [1610] Kumar J, Islam S U and Ghosh S G 2022 J. Cosmol. Astropart. Phys. JCAP11(2022)032
- [1611] Kumaran Y and Övgün A 2022 Symmetry 14 2054
- [1612] Pantig R C and Övgün A 2023 Fortschr. Phys. 71 2200164
- [1613] Badía J and Eiroa E F 2022 arXiv:2210.03081 [gr-qc]
- [1614] Hu S, Deng C, Li D, Wu X and Liang E 2022 Eur. Phys. J. C 82 885
- [1615] Frizo D A, de Melo C A M, Medeiros L G and Neves J C S 2022 arXiv:2210.09938 [gr-qc]
- [1616] Wu M-H, Guo H and Kuang X-M 2022 Symmetry 14 2237
- [1617] Antoniou G, Papageorgiou A and Kanti P 2023 Universe 9 147
- [1618] Islam S U, Kumar J, Kumar Walia R and Ghosh S G 2023 Astrophys. J. 943 22
- [1619] Atamurotov F, Hussain I, Mustafa G and Övgün A 2023 Chin. Phys. C 47 025102
- [1620] Sau S and Moffat J W 2022 arXiv:2211.15040 [gr-qc]
- [1621] Joshi A B, Tahelyani D, Dey D and Joshi P S 2022 arXiv:2212.03042 [gr-qc]
- [1622] Singh B P 2023 arXiv:2301.00956 [gr-qc]
- [1623] Tan H S 2023 arXiv:2301.04967 [gr-qc]
- [1624] Gómez G and Rodríguez J F 2023 arXiv:2301.05222 [gr-qc]
- [1625] Anjum A, Afrin M and Ghosh S G 2023 Phys. Dark Universe 40 101195
- [1626] Röder J, Cruz-Osorio A, Fromm C M, Mizuno Y, Younsi Z and Rezzolla L 2023 arXiv:2301.09549 [astro-ph.HE]
- [1627] Singh B P, Kumar R and Ghosh S G 2023 New Astron. 99 101945
- [1628] Xu R, Liang D and Shao L 2023 Astrophys. J. 945 148
- [1629] Nguyen B, Christian P and Chan C-K 2023 arXiv:2302.08094 [astro-ph.HE]
- [1630] Pantig R C 2023 arXiv:2303.01698 [gr-qc]